UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-Q

(Mark One)

☑ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended June 30, 2011

OR

o TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Commission files number 0-50626

CYCLACEL PHARMACEUTICALS, INC.

(Exact name of registrant as specified in its charter)

Delaware

(State or Other Jurisdiction of Incorporation or Organization)

91-1707622 (I.R.S. Employer Identification No.)

200 Connell Drive, Suite 1500 Berkeley Heights, New Jersey (Address of principal executive offices)

07922

(Zip Code)

Registrant's telephone number, including area code: (908) 517-7330

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \square No o

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes \square No o

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer o

Accelerated filer o

Non-accelerated filer o

Smaller reporting filer ☑

(Do not check if a smaller reporting company)

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes o No 🗵

As of August 12, 2011 there were 54,224,093 shares of the registrant's common stock outstanding.

CYCLACEL PHARMACEUTICALS, INC.

INDEX

	Page
Part I. Financial Information	3
Item 1. Financial Statements (Unaudited)	3
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	26
Item 3. Quantitative and Qualitative Disclosures About Market Risk	41
Item 4. Controls and Procedures	42
Part II. Other Information	
<u>Item 1. Legal Proceedings</u>	44
Item 1A. Risk Factors	44
Item 2. Unregistered Sales of Equity Securities and Use of Proceeds	69
Item 3. Defaults upon Senior Securities	69
Item 4. (Removed and Reserved)	69
Item 5. Other Information	69
<u>Item 6. Exhibits</u>	69
SIGNATURE PAGE	
Exhibit 10.1	
Exhibit 10.2	
Exhibit 31.1	
Exhibit 31.2	
Exhibit 32.1	
Exhibit 32.2	
EX-101 INSTANCE DOCUMENT	
EX-101 SCHEMA DOCUMENT	
EX-101 CALCULATION LINKBASE DOCUMENT	
EX-101 LABELS LINKBASE DOCUMENT	
EX-101 PRESENTATION LINKBASE DOCUMENT	

PART I. FINANCIAL INFORMATION

Item 1. Financial Statements.

CYCLACEL PHARMACEUTICALS, INC. (A Development Stage Company) CONDENSED CONSOLIDATED BALANCE SHEETS (In \$000s, except share amounts)

	Dec	cember 31, 2010		(une 30, 2011 naudited)
ASSETS				
Current assets:				
Cash and cash equivalents	\$	29,495	\$	20,614
Inventory		174		147
Prepaid expenses and other current assets		1,382		1,688
Total current assets		31,051		22,449
Property, plant and equipment (net)		408		246
Total assets	\$	31,459	\$	22,695
LIABILITIES AND STOCKHOLDERS' EQUITY				
Current liabilities:				
Accounts payable	\$	1,723	\$	1,104
Accrued liabilities and other current liabilities		4,132		4,459
Warrant liability		680		477
Total current liabilities		6,535		6,040
Total liabilities		6,535		6,040
Stockholders' equity:	<u> </u>			
Preferred stock, \$0.001 par value; 5,000,000 shares authorized at December 31, 2010 and				
June 30, 2011; 1,213,142 shares issued and outstanding at December 31, 2010 and June 30,				
2011. Aggregate preference in liquidation (including undeclared cumulative dividends) of				
\$13,344,562 at December 31, 2010 and June 30, 2011		1		1
Common stock, \$0.001 par value; 100,000,000 shares authorized at December 31, 2010 and				
June 30, 2011; 46,564,914 and 46,587,182 shares issued and outstanding at December 31,				
2010 and June 30, 2011, respectively		47		47
Additional paid-in capital		266,666		266,529
Accumulated other comprehensive loss		31		39
Deficit accumulated during the development stage		(241,821)		(249,961)
Total stockholders' equity		24,924	_	16,655
Total liabilities and stockholders' equity	\$	31,459	\$	22,695

The accompanying notes are an integral part of these consolidated financial statements.

CYCLACEL PHARMACEUTICALS, INC.

(A Development Stage Company)

CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS

(In \$000s, except share and per share amounts) (Unaudited)

	Thre	ee Mor Jund	nths Er e 30,	nded		Six Mont Jun	ths Enc	led	Aug (inc	riod from ust 13, 1996 ception) to June 30,
	2010			2011		2010		2011		2011
Revenues:										
Collaboration and research and										
development revenue	\$	100	\$	_	\$	100	\$	_	\$	3,100
Product revenue		19		168		273		360		2,682
Grant revenue						16				3,648
		119		168		389		360		9,430
Operating expenses:		0.0		=0		20.4		450		4 550
Cost of goods sold	1	92		72		234		178		1,570
Research and development		,322		1,859		3,497		4,939		181,532
Selling, general and administrative	3	,091		2,034		5,491		3,840		85,806
Goodwill and intangible										7,934
impairment Restructuring costs		_		_		_		_		2,634
G		,505		3,965		9,222		8,957		
Total operating expenses					_					279,476
Operating loss Other income (expense):	(4	,386)		(3,797)		(8,833)		(8,597)		(270,046)
Other income (expense).										
Costs associated with aborted 2004										
IPO		_		_		_		_		(3,550)
Payment under guarantee		_		_		_		_		(1,652)
Change in valuation of derivative		_		_		_		_		(308)
Change in valuation of warrants		273		125		(516)		203		6,273
Warrant re-pricing		_		_		_		_		(44)
Foreign exchange (losses)/gains		(49)		(19)		(38)		(87)		(4,342)
Interest income		8		13		17		24		13,704
Interest expense		(9)				(33)				(4,677)
Total other income (expense)		223		119		(570)		140		5,404
Loss before taxes	(4	,163)		(3,678)		(9,403)		(8,457)		(264,642)
Income tax benefit		230		126		363		317		18,196
Net loss	(3	,933)		(3,552)		(9,040)		(8,140)		(246,446)
Dividends on preferred ordinary shares				_		_		_		(38,123)
Deemed dividend on convertible										(50,125)
exchangeable preferred shares	(2	,496)				(2,915)		_		(3,515)
Dividend on convertible exchangeable	(=	, 150)				(2,515)				(5,515)
preferred shares		(114)		(182)		(403)		(364)		(3,293)
Net loss applicable to common										
shareholders	\$ (6	,543)	\$	(3,734)	\$	(12,358)	\$	(8,504)	\$	(291,377)
Net loss per share — Basic and diluted	\$ (0.18)	\$	(80.0)	\$	(0.36)	\$	(0.18)		
Weighted average common shares		=	-							
outstanding	36,565	<u>,972</u>	46	,582,915		4,157,279	46	5,577,577		

The accompanying notes are an integral part of these consolidated financial statements.

CYCLACEL PHARMACEUTICALS, INC.

(A Development Stage Company) CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (In \$000s) (Unaudited)

		Six Mont	-		A (iı	riod from ugust 13, 1996 nception) to
		2010		2011		2011
Cash flows from operating activities:	ф	(0.040)	ď	(0.1.40)	ď	(0.46, 4.46)
Net loss	\$	(9,040)	\$	(8,140)	\$	(246,446)
Adjustments to reconcile net loss to net cash used in operating activities:		2				100
Accretion of interest on notes payable, net of amortization of debt premium		2		_		100
Amortization of investment premiums, net		_		_		(2,297)
Change in valuation of derivative				(202)		308
Change in valuation of warrants		516		(203)		(6,273) 44
Warrant re-pricing Depreciation and amortization		240		— 172		12,486
		240		1/2		886
Amortization of intangible assets Fixed asset impairment		_		<u> </u>		221
Goodwill and intangibles impairment		_		_		7,934
Unrealized foreign exchange loss		<u> </u>		21		7,768
Deferred revenue		_		21		(98)
Compensation for warrants issued to non-employees						1,215
Shares issued for IP rights						446
(Gain) loss on disposal of property, plant and equipment		(9)				99
Stock based compensation		786		455		18,596
Provision for restructuring		700		4 33		1,779
Amortization of issuance costs of Preferred Ordinary "C" shares						2,517
Changes in operating assets and liabilities:						2,517
Prepaid expenses, inventory and other current assets		(88)		(279)		(511)
Accounts payable, accrued liabilities and other current liabilities		(2,183)		(482)		(6,372)
Net cash used in operating activities	_	(9,776)		(8,456)	_	(207,598)
		(9,770)		(0,430)	_	(207,396)
Investing activities: Purchase of ALIGN						(2.762)
		(0)				(3,763)
Purchase of property, plant and equipment Proceeds from sale of property, plant and equipment		(8) 35		_		(8,831) 158
Purchase of short-term investments		33		<u> </u>		(156,657)
Redemptions of short-term investments, net of maturities		_		_		162,729
Net cash provided by (used in) investing activities		27			_	(6,364)
Financing activities:						(2.510)
Payment of capital lease obligations		_		_		(3,719)
Proceeds from issuance of ordinary and preferred ordinary shares, net of						101 670
issuance costs		15 155		(00)		121,678
Proceeds from issuance of common stock and warrants, net of issuance costs		15,155		(80)		82,324
Net proceeds from stock options and warrants exercised		2,700		3		173

CYCLACEL PHARMACEUTICALS, INC. (A Development Stage Company) CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (In \$000s) (Unaudited)

	Six Month June	30,	Period from August 13, 1996 (inception) to June 30,
	2010	2011	2011
Payment of preferred stock dividend	_	(364)	(1,898)
Repayment of government loan	_		(455)
Government loan received	_	_	414
Loan received from Cyclacel Group Plc	_	_	9,103
Proceeds of committable loan notes issued from shareholders	_	_	8,883
Loans received from shareholders	_	_	1,645
Cash and cash equivalents assumed on stock purchase	_	_	17,915
Costs associated with stock purchase			(1,951)
Net cash provided by (used in) financing activities	17,855	(441)	234,112
Effect of exchange rate changes on cash and cash equivalents	(56)	16	464
Net increase (decrease) in cash and cash equivalents	8,050	(8,881)	20,614
Cash and cash equivalents at beginning of period	11,493	29,495	
Cash and cash equivalents at end of period	\$ 19,543	\$ 20,614	\$ 20,614
Supplemental disclosure of cash flows information:			
Cash received during the period for:			
Interest	11	11	11,726
Taxes	110	_	17,522
Cash paid during the period for:			
Interest	(187)	_	(1,914)
Schedule of non-cash transactions:			
Acquisitions of equipment purchased through capital leases	_	_	3,470
Issuance of common shares in connection with license agreements	_	_	592
Issuance of ordinary shares on conversion of bridging loan	_	_	1,638
Issuance of preferred ordinary "C" shares on conversion of secured			
convertible loan notes and accrued interest	_	_	8,893
Issuance of ordinary shares in lieu of cash bonus	_	_	164
Issuance of other long term payable on ALIGN acquisition	_	_	1,122

The accompanying notes are an integral part of these consolidated financial statements.

CYCLACEL PHARMACEUTICALS, INC. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

1. NATURE OF OPERATIONS AND BASIS OF PRESENTATION

Nature of Operations

Cyclacel Pharmaceuticals, Inc. ("Cyclacel" or the "Company") is a development-stage biopharmaceutical company dedicated to the development and commercialization of novel, mechanism-targeted drugs to treat human cancers and other serious diseases. Cyclacel's strategy is to build a diversified biopharmaceutical business focused in hematology and oncology based on a portfolio of commercial products and a development pipeline of novel drug candidates.

Cyclacel's clinical development priorities are focused on sapacitabine in the following indications:

- Acute myeloid leukemia, or AML, in the elderly;
- · Myelodysplastic syndromes, or MDS; and
- Non-small cell lung cancer, or NSCLC.

On January 11, 2011, the Company opened enrollment of the SEAMLESS pivotal Phase 3 trial for the Company's sapacitabine oral capsules as a front-line treatment of elderly patients aged 70 years or older with newly diagnosed AML who are not candidates for intensive induction chemotherapy under a Special Protocol Assessment, or SPA, reached with the U.S. Food & Drug Administration, or FDA.

The Company has advanced two additional product candidates, seliciclib in Phase 2 for NSCLC and nasopharyngeal cancer or NPC, and CYC116 in Phase 1 clinical development. The combination of sapacitabine with seliciclib is also being evaluated in a Phase 1 clinical trial. The Company will determine the feasibility of pursuing further development and/or partnering these assets depending on the availability of funding and further clinical data. In addition, the Company markets directly in the United States Xclair® Cream for radiation dermatitis and Numoisyn® Liquid and Numoisyn® Lozenges for xerostomia.

As a development stage enterprise, substantially all efforts of the Company to date have been devoted to performing research and development, conducting clinical trials, developing and acquiring intellectual property and raising capital.

Basis of Presentation

The condensed consolidated balance sheet as of June 30, 2011, the condensed consolidated statements of operations for the three and six months ended June 30, 2011 and 2010 and the condensed consolidated statements of cash flows for the six months ended June 30, 2011 and 2010, and related disclosures contained in the accompanying notes are unaudited. The condensed consolidated balance sheet as of December 31, 2010 is derived from the audited consolidated financial statements included in the Annual Report on Form 10-K for the year ended December 31, 2010 filed with the Securities and Exchange Commission ("SEC"). The condensed consolidated financial statements are presented on the basis of accounting principles that are generally accepted in the United States ("GAAP") for interim financial information and in accordance with the rules and regulations of the SEC; accordingly, they do not include all the information and footnotes required by accounting principles generally accepted in the United States for a complete set of financial statements. In the opinion of management, all adjustments, which include only normal recurring adjustments, necessary to present fairly the condensed consolidated balance sheet as of June 30, 2011, the results of operations for the three and six months ended June 30, 2011 and 2010 and the consolidated statements of cash flows for the six months ended June 30, 2011 and 2010 have been made. The interim results for the three and six months ended June 30, 2011 are not necessarily indicative of the results to be expected for the year ending December 31, 2011 or for any other year. The condensed consolidated financial statements should be read in conjunction with the audited consolidated financial statements and the accompanying notes included in the Company's Annual Report on Form 10-K for the year ended December 31, 2010 filed with the SEC.

Subsequent Developments

Preferred Stock Dividend

On July 8, 2011, the Board of Directors (the "Board") decided not to declare the quarterly cash dividend on the Company's 6% Convertible Exchangeable Preferred Stock ("Preferred Stock") with respect to the second quarter of 2011 that would have otherwise been payable on August 1, 2011.

The Board also did not declare the quarterly cash dividend with respect to each of the four quarters of fiscal year 2009 and the first, second and third quarters of 2010. To the extent that any dividends payable on the Preferred Stock are not paid, such unpaid dividends are added to the liquidity preference of the Preferred Stock. As the Company failed to pay in an aggregate amount equal to at least six quarterly dividends (whether or not consecutive) on the Preferred Stock, the size of the Company's Board was increased by two members and the holders of the Preferred Stock, voting separately as a class, voted on May 24, 2011 and elected two directors to fill the vacancies created thereby, which directorships shall terminate when the Company pays all accrued but unpaid dividends.

License Agreement

On July 11, 2011, the Company entered into an amendment to its license agreement with Daiichi Sankyo Company, Limited ("Daiichi Sankyo"), dated September 10, 2003, relating to certain rights which the Company has licensed from Daiichi Sankyo with regard to the Company's sapacitabine drug. Effective July 11, 2011, the license agreement was amended to irrevocably waive a termination right Daiichi Sankyo possessed under a provision of the agreement that required the Company to obtain regulatory approval to sell sapacitabine in at least one country by September 2011, and releases the Company from all claims and liability of any kind arising under such provision. The amendment further provides that the royalty due from the Company to Daiichi Sankyo on future net sales of sapacitabine be increased by a percentage between 1.25% and 1.50% depending on the level of net sales of sapacitabine realized.

Financing

On July 7, 2011, the Company closed an underwritten offering for an aggregate of 7,617,646 units, at an offering price of \$1.36 per unit, for gross proceeds of approximately \$10.4 million. Each unit consists of (i) one share of common stock and (ii) a five-year warrant to purchase 0.5 of a share of common stock at an exercise price of \$1.36 per share, exercisable beginning six months after the date of issuance. The shares of common stock and warrants were immediately separable and were issued separately such that no units were issued. The net proceeds, after underwriting discounts and commissions and other fees and expenses payable by the Company, were approximately \$9.3 million.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Consolidation

The accompanying condensed consolidated financial statements include the accounts of the Company and its wholly-owned subsidiaries for the indicated periods. All significant intercompany transactions and balances have been eliminated.

Use of Estimates

The preparation of financial statements in accordance with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets, liabilities and related disclosures of contingent assets and liabilities at the date of the financial statements and the reported amounts of income and expenses during the reporting period. The Company reviews its estimates on an ongoing basis. The estimates are based on historical experience and on various other assumptions that the Company believes to be reasonable under the circumstances. Actual results may differ from these estimates.

Cash and Cash Equivalents

Cash and cash equivalents are stated at cost, which is substantially the same as fair value. The Company considers all highly liquid investments with an original maturity of three months or less at the time of purchase to be cash equivalents. The Company's cash equivalents generally include investments in money market funds and corporate commercial paper.

Trade Accounts Receivable and Allowance for Doubtful Accounts

An allowance for doubtful accounts is provided, as necessary, on trade receivables based on their respective aging categories and historical collection experience, taking into consideration the type of payer, historical and projected collection outcomes, and current economic and business conditions that could affect the collectability of the Company's receivables. The allowance for doubtful accounts is reviewed, at a minimum, on a quarterly basis. Changes in the allowance for doubtful accounts are recorded as an adjustment to bad debt expense within general and administrative expenses. Material revisions to reserve estimates may result from adverse changes in collection experience. The Company writes off accounts against the allowance for doubtful accounts when reasonable collection efforts have been unsuccessful and it is likely the receivable will not be recovered. At December 31, 2010 and June 30, 2011, all receivables were deemed collectible.

For the three months ended June 30, 2010 and 2011, approximately 87% and 92%, respectively, of the Company's product sales in the United States were to three wholesalers. For the six months ended June 30, 2010 and 2011, approximately 83% and 91%, respectively, of the Company's product sales in the United States were to three wholesalers.

Inventory

Cyclacel values inventories at the lower of cost or market value. The Company determines cost using the first-in, first-out method. As of December 31, 2010 and June 30, 2011, all inventories were classified as finished goods. The Company analyzes its inventory levels at least quarterly to identify any items that may expire prior to sale, inventory that has a cost basis in excess of net realizable value, or inventory in excess of expected sales requirements. The determination of whether or not inventory costs will be realizable requires estimates by the Company's management. A critical input is future sales forecasts. The Company writes down the value of inventory to the extent that inventory is expected to expire before being sold or if the cost basis is in excess of net realizable value. If actual market conditions are less favorable than those projected by management, additional inventory write-downs may be required in future periods.

Revenue Recognition

Product sales

The Company recognizes revenue from product sales when persuasive evidence of an arrangement exists; delivery has occurred or services have been rendered; the selling price is fixed or determinable; and collectability is reasonably assured.

The Company offers a general right of return on these product sales, and has considered the guidance in ASC 605-15, "Revenue Recognition -Products" ("ASC 605-15") and ASC 605 -10 "Revenue Recognition — Overall" ("ASC 605-10"). Under these guidelines, the Company accounts for all product sales using the "sell-through" method. Under the sell-through method, revenue is not recognized upon shipment of product to distributors. Instead, the Company records deferred revenue at gross invoice sales price and deferred cost of sales at the cost at which those goods were held in inventory. The Company recognizes revenue when such inventory is sold through to pharmacies. To estimate product sold through to pharmacies, the Company relies on third-party information, including information obtained from significant distributors with respect to their inventory levels and sell-through to pharmacies. The Company estimates product returns based on historical returns experience and other factors that affect demand for the Company's products as well as the amount of product subject to return.

Deferred revenue was \$0.1 million at June 30, 2010 and 2011. Deferred cost of goods sold was approximately \$58,000 and \$11,000 million at June 30, 2010 and 2011, respectively.

Collaboration, research and development, and grant revenue

Certain of the Company's revenues are earned from collaborative agreements. The Company recognizes revenue when persuasive evidence of an arrangement exists; delivery has occurred or services have been rendered; the fee is fixed or determinable; and collectability is reasonably assured. Determination of whether these criteria have been met is based on management's judgments regarding the nature of the research performed, the substance of the milestones met relative to those the Company must still perform, and the collectability of any related fees. Should changes in conditions cause management to determine these criteria are not met for certain future transactions, revenue recognized for any reporting period could be adversely affected.

Research and development revenues, which are earned under agreements with third parties for contract research and development activities, are recorded as the related services are performed. Milestone payments are non-refundable and recognized as revenue when earned, as evidenced by achievement of the specified milestones and the absence of ongoing performance obligations. Any amounts received in advance of performance are recorded as deferred revenue. None of the revenues recognized to date are refundable if the relevant research effort is not successful.

Grant revenues from government agencies and private research foundations are recognized as the related qualified research and development costs are incurred, up to the limit of the prior approval funding amounts. All grants received are not refundable.

Clinical Trial Accounting

Data management and monitoring of all of the Company's clinical trials are performed by contract research organizations ("CROs") or clinical research associates ("CRAs") in accordance with the Company's standard operating procedures. Typically, CROs and some CRAs bill monthly for services performed, and others bill based upon milestones achieved. The Company accrues unbilled clinical trial expenses based on estimates of the level of services performed each period. Costs of setting up investigational sites for participation in the Company's clinical trials are expensed immediately as research and development expenses. Clinical trial site costs related to patient enrollment are accrued as patients are entered into the trial. Any initial payment made to the clinical trial site is recognized upon execution of the clinical trial agreements and expensed as research and development expenses.

Research and Development Expenditures

Research and development expenses consist primarily of costs associated with the Company's product candidates, upfront fees, milestones, compensation and other expenses for research and development personnel, supplies and development materials, costs for consultants and related contract research, facility costs, amortization of purchased technology and depreciation. Expenditures relating to research and development are expensed as incurred.

Foreign currency and currency translation

Transactions that are denominated in a foreign currency are re-measured into the functional currency at the current exchange rate on the date of the transaction. Any foreign currency-denominated monetary assets and liabilities are subsequently re-measured at current exchange rates, with gains or losses recognized as foreign exchange (losses)/gains in the statement of operations.

The assets and liabilities of the Company's international subsidiary are translated from its functional currency into United States dollars at exchange rates prevailing at the balance sheet date. Average rates of exchange during the period are used to translate the statement of operations, while historical rates of exchange are used to translate any equity transactions.

Translation adjustments arising on consolidation due to differences between average rates and balance sheet rates, as well as unrealized foreign exchange gains or losses arising from translation of intercompany loans that are of a long-term-investment nature, are recorded in other comprehensive income.

Derivative Instruments

Derivative financial instruments are measured at fair value. Significant judgments and estimates are necessary in determining the fair value in the absence of quoted market values. These estimates are based on valuation methodologies and assumptions deemed appropriate in the circumstances. The use of different assumptions may have a material effect on the estimated fair value amount and the Company's results of operations.

Fair Value Measurements

Inputs used to determine fair value of financial and non-financial assets and liabilities are categorized using a fair value hierarchy that prioritizes observable and unobservable inputs into three broad levels, from Level 1, which is the most reliable, to Level 3, which is the least reliable (see "Note 3 — Fair Value Measurements"). Management reviews the categorization of fair value inputs on a periodic basis and may determine that it is necessary to transfer an input from one level of the fair value hierarchy to another based on changes in events or circumstances, such as a change in the observability of an input. Any such transfer will be recognized at the end of the reporting period.

Income Taxes

The Company accounts for income taxes under the liability method. Under this method, deferred tax assets and liabilities are determined based on the difference between the financial statement and tax bases of assets and liabilities using enacted tax rates in effect for the year in which the differences are expected to affect taxable income. Valuation allowances are established when necessary to reduce deferred tax assets to the amounts expected to be realized. The Company's management has established a full valuation allowance against its deferred tax assets based on the determination that it is not more likely than not that the Company will recognize the benefits of those assets.

The Company adopted the guidance related to accounting for uncertainty in income taxes, primarily codified in ASC 740, "*Income taxes*" ("ASC 740"). ASC 740 specifies the accounting for uncertainty in income taxes recognized in a company's financial statements by prescribing a minimum probability threshold a tax position is required to meet before being recognized in the financial statements.

The Company records income tax benefits related to research and development tax credits, which will be claimed from H. M. Revenue & Customs, the United Kingdom's taxation and customs authority, with respect to qualifying research and development costs incurred in the same accounting period.

Stock-based Compensation

The Company grants stock options, restricted stock units and restricted stock to officers, employees and directors under the 2006 Amended and Restated Equity Incentive Plan ("2006 Plan"), which was approved on March 16, 2006 and subsequently amended and restated on April 14, 2008. The Company also has outstanding options under various stock-based compensation plans for employees and directors. These plans are described more fully in "Note 6 — Stock-Based Compensation". The Company accounts for these plans under ASC 718, "Compensation — Stock Compensation" ("ASC 718").

ASC 718 requires measurement of compensation cost for all stock-based awards at fair value on date of grant and recognition of compensation over the requisite service period for awards expected to vest. The fair value of restricted stock and restricted stock units is determined based on the number of shares granted and the quoted price of the Company's common stock on the date of grant. The determination of grant-date fair value for stock option awards is estimated using the Black-Scholes model, which includes variables such as the expected volatility of our share price, the anticipated exercise behavior of our employees, interest rates, and dividend yields. These variables are projected based on our historical data, experience, and other factors. Changes in any of these variables could result in material adjustments to the expense recognized for share-based payments. Such value is recognized as expense over the requisite service period, net of estimated forfeitures, using the straight-line attribution method. The estimation of stock awards that will ultimately vest requires judgment, and to the extent actual results or updated estimates differ from current estimates, such amounts are recorded as a cumulative adjustment in the period estimates are revised. The Company considers many factors when estimating expected forfeitures, including type of awards granted employee class, and historical experience. Actual results and future estimates may differ substantially from current estimates.

Segments

The Company has determined its reportable segments in accordance with ASC 280, "Segment Reporting" ("ASC 280") and related disclosures about products, services, geographic areas and major customers. After considering its business activities and geographic reach, the Company has concluded that it operates in just one operating segment being the discovery, development and commercialization of novel, mechanism-targeted drugs to treat cancer and other serious disorders, with development operations in two geographic areas, namely the United States and the United Kingdom.

Net Loss per Common Share

The Company calculates net loss per common share in accordance with ASC 260, "Earnings Per Share" ("ASC 260"). Basic and diluted net loss per common share were determined by dividing net loss applicable to common stockholders by the weighted average number of common shares outstanding during the period. The Company's potentially dilutive shares, which include outstanding common stock options, restricted stock, restricted stock units, convertible preferred stock, and common stock warrants, have not been included in the computation of diluted net loss per share for all periods as the result would be anti-dilutive.

	June 30,	June 30,
	2010	2011
Stock options	3,187,291	3,543,529
Restricted stock and restricted stock units	75,515	44,255
Convertible preferred stock	516,228	516,228
Options to purchase common stock and common stock warrants issued in connection with the		
October 2010 financing	_	6,242,398
Common stock warrants	5,843,597	10,005,192
Total shares excluded from calculation	9,622,631	20,351,602

Comprehensive Income (Loss)

In accordance with ASC 220, "Comprehensive Income" ("ASC 220") all components of comprehensive income (loss), including net income (loss), are reported in the financial statements in the period in which they are recognized. ASC 220 defines comprehensive income (loss) as the change in equity during a period from transactions and other events and circumstances from nonowner sources. Net income (loss) and other comprehensive income (loss), including foreign currency translation adjustments, are reported, net of any related tax effect, to arrive at comprehensive income (loss). No taxes were recorded on items of other comprehensive income.

	Three Montl June 3		Six Months June 3		Period from August 13, 1996 (inception) to June 30,
	2010	2011	2010	2011	2011
			\$000		
Net loss	(3,933)	(3,552)	(9,040)	(8,140)	(246,446)
Translation adjustment	(19)	139	4,001	(2,826)	5,952
Unrealized foreign exchange					
(loss) gain on intercompany loans	9	(108)	(3,979)	2,834	(5,913)
Comprehensive loss	(3,943)	(3,521)	(9,018)	(8,132)	(246,407)

3. FAIR VALUE MEASUREMENTS

As defined in ASC 820, fair value is based on the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. In order to increase consistency and comparability in fair value measurements, ASC 820 establishes a fair value hierarchy that prioritizes observable and unobservable inputs used to measure fair value into three broad levels, which are described below:

- Level 1: Quoted prices (unadjusted) in active markets that are accessible at the measurement date for assets or liabilities. The fair value hierarchy gives the highest priority to Level 1 inputs.
- Level 2: Inputs other than quoted prices within Level 1 that are observable for the asset or liability, either directly or indirectly.
- Level 3: Unobservable inputs that are used when little or no market data is available. The fair value hierarchy gives the lowest priority to Level 3 inputs.

In determining fair value, the Company utilizes valuation techniques that maximize the use of observable inputs and minimize the use of unobservable inputs to the extent possible as well as considers counterparty credit risk in its assessment of fair value. Financial assets and liabilities carried at fair value on a recurring basis as of December 31, 2010 are classified in the table below in one of the three categories described above:

	Level 1 	Level 2 \$000	Level 3 \$000	Total \$000
Cash equivalents	29,066	_	_	29,066
Warrants			680	680
Total	29,066		680	29,746

Financial assets and liabilities carried at fair value on a recurring basis as of June 30, 2011 are classified in the table below in one of the three categories described above:

	Level 1 	Level 2 \$000	Level 3 \$000	Total \$000
Cash equivalents	17,333	_	_	17,333
Warrants			477	477
Total	17,333		477	17,810

Warrants Liability

The Company issued warrants to purchase shares of common stock under the registered direct financing completed in February 2007. These warrants are being accounted for as a liability in accordance with ASC 815, "*Derivatives and Hedging*" ("ASC 815"). At the date of the transaction, the fair value of the warrants of \$6.8 million was determined utilizing the Black-Scholes option pricing model utilizing the following assumptions: risk free interest rate — 4.68%, expected volatility — 85%, expected dividend yield — 0%, and a contractual life of 7 years. The fair value of the warrant is being re-measured each reporting period, with a derivative gain or loss recognized in the consolidated statement of operations. Such gains or losses will continue to be reported until the warrants are exercised or expired. The Company used the Black-Scholes option-pricing model with the following assumptions to value the warrants:

	De	December 31,		June 30,
		2010		2011
Exercise price	\$	8.44	\$	8.44
Expected term		3.13 Yrs.		2.63 Yrs.
Risk free interest rate		1.02%		0.68%
Expected volatility		121%		118%
Expected dividend yield over expected term		_		

During the three months ended June 30, 2010 and 2011, the Company recognized the change in the value of warrants of approximately \$0.3 million and \$0.1 million, respectively, as a gain on the consolidated statement of operations. During the six months ended June 30, 2010, the Company recognized the \$0.5 million increase in the value of warrants as a loss on the consolidated statement of operations. During the six months ended June 30, 2011, the Company recognized the \$0.2 million decrease in the value of warrants as a gain on the consolidated statement of operations. The following table reconciles the beginning and ending balance of Level 3 inputs for the six months ended June 30, 2011:

	Level 3
	\$000
Balance as of December 31, 2010	680
Gain from change in valuation of warrants liability reported in earnings	(203)
Balance as of June 30, 2011	477

4. PREPAID EXPENSES AND OTHER CURRENT ASSETS

Prepaid expenses and other current assets consist of the following:

	December 31,	June 30,	
	2010	2011	
	(\$000s)		
Research and development tax credit receivable	660	998	
Prepayments	317	476	
Other current assets	405	214	
Total prepaid expenses and other current assets	1,382	1,688	

5. ACCRUED LIABILITIES AND OTHER CURRENT LIABILITIES

Accrued and other current liabilities consisted of the following:

	December 31,	June 30,
	2010	2011
	(\$00	0 s)
Accrued research and development	2,793	3,343
Other current liabilities	1,339	1,116
	4,132	4,459

6. STOCK BASED COMPENSATION

Stock based compensation has been reported within expense line items on the consolidated statement of operations for the three months ended June 30, 2010 and 2011 as shown in the following table:

		ree months June 30,		ix months June 30,
	2010	2010 2011		2011
		(\$00	00s)	
Research and development	93	41	165	90
General and administrative	374	164	621	365
Stock-based compensation costs before income taxes	467	205	786	455

Under the 2006 Plan, 5,200,000 shares of the Company's common stock have been reserved. The awards granted under the 2006 Plan have a maximum maturity of 10 years and generally vest over a four-year period from the date of grant.

A summary of activity for the options under the Company's 2006 Plan for the six months ended June 30, 2011 is as follows:

	<u>Options</u>	A	eighted verage cise Price	Weighted Average Remaining Contractual Term (years)	Aggregate Intrinsic Value (in \$000s)
Options outstanding at December 31, 2010	3,489,932	\$	3.96	7.22	938
Granted	199,500	\$	1.52		
Exercised	(6,638)	\$	0.41		
Expired	_				
Cancelled / forfeited	(139,265)	\$	6.32		
Options outstanding at June 30, 2011	3,543,529	\$	3.74	6.95	828
Unvested at June 30, 2011	968,024	\$	1.83	8.75	153
Vested and exercisable at June 30, 2011	2,575,505	\$	4.45	6.27	675

ASC 718 requires compensation expense associated with share-based awards to be recognized over the requisite service period, which for the Company is the period between the grant date and the date the award vests or becomes exercisable. Most of the awards granted by the Company (and still outstanding), vest ratably over four years, with 1/4 of the award vesting one year from the date of grant and 1/48 of the award vesting each month thereafter. However, certain awards made to executive officers vest over three to five years, depending on the terms of their employment with the Company. In addition, recent awards made to other employees vest ratably over four years, with 1/48 of the award vesting each month.

Effective January 1, 2006, the Company elected to recognize all share-based awards issued after the adoption of ASC 718 under the straight-line attribution method. ASC 718 requires forfeitures to be estimated at the time of grant and revised, if necessary, in subsequent periods if actual forfeitures differ from those estimates. This analysis is evaluated quarterly and the forfeiture rate adjusted as necessary. Ultimately, the actual expense recognized over the vesting period is based on only those shares that vest.

The Company used the Black-Scholes option-pricing model with the following assumptions for stock option grants to employees and directors for the six months ended June 30, 2010 and 2011:

	For the si ended J	
	2010	2011
Expected term	5 – 6Yrs	5 – 6Yrs
Risk free interest rate	2.37 - 2.96%	1.47 - 2.29%
Expected volatility	90 - 100%	93 - 99%
Expected dividend yield over expected term	_	_
Resulting weighted average grant fair value	\$1.80	\$1.15

There were 199,500 options granted during the six months ended June 30, 2011. For grants made during the six months ended June 30, 2010 and June 30, 2011, the expected term assumption was estimated using past history of early exercise behavior and estimated expectations of future exercise behavior. Starting with the December 2010 annual grants to the Company's employees, the Company relied exclusively on its historical volatility as an input to the option pricing model as the Company's management believes that this rate will be representative of future volatility over the expected term of the options. Prior to December 2010, because the Company had been publicly traded for a limited period, the expected volatility assumption was based on the historical volatility of peer companies over the expected term of the option awards.

Estimates of pre-vesting option forfeitures are based on the Company's experience. For outstanding options, the Company uses a forfeiture rate of 0 — 50% depending on when and to whom the options are granted. The Company adjusts its estimate of forfeitures over the requisite service period based on the extent to which actual forfeitures differ, or are expected to differ, from such estimates. Changes in estimated forfeitures are recognized through a cumulative adjustment in the period of change and may impact the amount of compensation expense to be recognized in future periods.

The weighted average risk-free interest rate represents interest rate for treasury constant maturities published by the Federal Reserve Board. If the term of available treasury constant maturity instruments is not equal to the expected term of an employee option, Cyclacel uses the weighted average of the two Federal Reserve securities closest to the expected term of the employee option.

During the six months ended June 30, 2011, 6,638 stock options were exercised resulting in approximately \$3,000 of cash proceeds to the Company. During the six months ended June 30, 2010, there were 124,277 stock options exercised for proceeds of approximately \$0.1 million. As the Company presently has tax loss carry forwards from prior periods and expects to incur tax losses in 2011, the Company is not able to benefit from the deduction for exercised stock options in the current reporting period.

Restricted Stock

In November 2008, the Company issued restricted common stock to an employee subject to certain forfeiture provisions. Specifically, one quarter of the award vested one year from the date of grant and 1/48 of the award effectively vests each month thereafter. This restricted stock grant is accounted for at fair value at the date of grant and an expense is recognized during the vesting term. Summarized information for the restricted stock grant for the six months ended June 30, 2011 is as follows:

	Restricted Stock	Weighted Average Grant Date Value Per Share
Non-vested at December 31, 2010	23,954	\$ 0.44
Vested	(6,252)	\$ 0.44
Non-vested at June 30, 2011	17,702	\$ 0.44

Restricted Stock Units

Restricted stock units were issued to senior executives of the Company in November 2008, which entitle the holders to receive a specified number of shares of the Company's common stock over the four year vesting term. A restricted stock unit grant is accounted for at fair value at the date of grant which is equivalent to the market price of a share of the Company's common stock, and an expense is recognized during the vesting term. There were no restricted stock unit grants prior to November 2008. Summarized information for restricted stock grants for the six months ended June 30, 2011 is as follows:

	Restricted Stock Units	U	ted Average Grant Value Per Share
Non-vested at December 31, 2010	35,931	\$	0.44
Vested	(9,378)	\$	0.44
Non-vested at June 30, 2011	26,553	\$	0.44

7. COMMITMENTS AND CONTINGENCIES

Licensing Agreements

The Company has entered into licensing agreements with academic and research organizations. Under the terms of these agreements, the Company has received licenses to technology and patent applications. The Company is required to pay royalties on future sales of product employing the technology or falling under claims of patent applications.

Pursuant to the Daiichi Sankyo license under which the Company licenses certain patent rights for sapacitabine, its lead drug candidate, the Company is under an obligation to use reasonable endeavors to develop a product and obtain regulatory approval to sell a product and has agreed to pay Daiichi Sankyo an up-front fee, reimbursement for Daiichi Sankyo's enumerated expenses, milestone payments and royalties on a country-by-country basis. Under this agreement, aggregate milestone payments totaling \$11.7 million could be payable subject to achievement of all the specific contractual milestones and the Company's decision to continue with these projects. The up-front fee and certain past reimbursements have been paid and, as a result of the SEAMLESS trial entering Phase 3 during the first quarter of 2011, \$1.6 million was paid in July 2011. Following these milestone payments, a further \$9.3 million in aggregate milestone payments could be payable. Royalties are payable in each country for the term of patent protection in the country or for ten years following the first commercial sale of licensed products in the country, whichever is later. Royalties are payable on net sales. Net sales are defined as the gross amount invoiced by the Company or its affiliates or licensees, less discounts, credits, taxes, shipping and bad debt losses. The agreement extends from its commencement date to the date on which no further amounts are owed under it. If the Company wishes to appoint a third party to develop or commercialize a sapacitabine-based product in Japan, within certain limitations, Daiichi Sankyo must be notified and given a right of first refusal to develop and/or commercialize in Japan. In general, the license may be terminated by the Company for technical, scientific, efficacy, safety, or commercial reasons on six months notice, or twelve months, if after a launch of a sapacitabine-based product, or by either party for material default. Effective July 11, 2011, the license agreement was amended to irrevocably waive a termination right Daiichi Sankyo possessed under a provision of the agreement that required the Company to obtain regulatory approval to sell sapacitabine in at least one country by September 2011, and releases the Company from all claims and liability of any kind arising under such provision. The amendment further provides that the royalty due from the Company to Daiichi Sankyo on future net sales of sapacitabine be increased by a percentage between 1.25% and 1.50% depending on the level of net sales of sapacitabine realized.

Guarantee

On July 28, 2005 and as amended on March 27, 2006, Cyclacel Group plc ("Group") signed a convertible Loan Note Instrument constituting convertible unsecured loan notes (the "Loan") and entered into a Facility Agreement ("Agreement") with Scottish Enterprise ("SE"), as lender, whereby SE subscribed for £5 million, or approximately \$9 million at the time, of the convertible loan notes. The loan was subsequently converted into 1,231,527 preferred D shares of the Group in satisfaction of all amounts owed by Group under the convertible loan notes. The number of preferred D shares that SE received was calculated by dividing the principal amount outstanding under the loan note by £4.06. The preferred D shares were exchanged for shares in Xcyte Therapies, Inc. on March 27, 2006 as part of the transaction between Xcyte and Cyclacel Limited. However, Scottish Enterprise retained the ability it had under the Agreement to receive a cash payment should the research operations in Scotland be significantly reduced. The Company had guaranteed approximately £5 million, the amount potentially due to SE, which will be calculated as a maximum of £5 million less the market value of the shares held by SE at the time of any significant reduction in research facilities.

On June 22, 2009, the Company amended the Agreement with SE (the "Amendment"), in order to allow the Company to implement a reduction of the Company's research operations located in Scotland in exchange for the parties' agreement to modify the payment terms of the Agreement in the principal amount of £5 million (approximately \$8.0 million at December 31, 2009), which SE had previously entered into with the Company. The Agreement provided for repayment of £5 million in the event the Company significantly reduced its Scottish research operations. Pursuant to the terms of the Amendment, in association with Cyclacel's material reduction in staff at its Scottish research facility, the parties agreed to a modified payment of £1 million (approximately \$1.7 million at June 22, 2009) payable in two equal tranches. On July 1, 2009, the first installment of £0.5 million (approximately \$0.8 million) was paid and the remaining amount of £0.5 million (approximately \$0.8 million) was paid on January 6, 2010. In addition, should a further reduction below current minimum staff levels be effectuated before July 2014 without SE's prior consent, the Company will guarantee approximately £4 million, the amount potentially due to SE, which will be calculated as a maximum of £4 million less the market value of the shares held by SE at the time of any further reduction in research facilities.

This arrangement is accounted for as a liability and is measured at fair value. Changes in fair value are recognized in earnings. Due to the nature of the associated contingency and the likelihood of occurrence, the Company has concluded the fair value of this liability is immaterial as of December 31, 2010 and June 30, 2011.

Legal proceedings

On April 27, 2010, the Company was served with a complaint filed by Celgene Corporation in the United States District Court for the District of Delaware seeking a declaratory judgment that four of its own patents, claiming the use of romidepsin injection in T-cell lymphomas, are invalid and not infringed by Celgene's products, but directly involve the use and administration of Celgene's ISTODAX® (romidepsin for injection) product. On June 17, 2010, the Company filed its answer and counterclaims to the declaratory judgment complaint. The Company filed counterclaims charging Celgene with infringement of each of the Company's four patents and seeking damages for Celgene's infringement as well as injunctive relief. The four patents directly involve the use and administration of Celgene's ISTODAX® (romidepsin for injection) product.

8. STOCKHOLDERS' EQUITY

Preferred Stock

As of June 30, 2011, there were 1,213,142 shares of Preferred Stock issued and outstanding at an issue price of \$10.00 per share. Dividends on the Preferred Stock are cumulative from the date of original issuance at the annual rate of 6% of the liquidation preference of the Preferred Stock, payable quarterly on the first day of February, May, August and November, commencing February 1, 2005. Any dividends must be declared by the Company's Board of Directors and must come from funds that are legally available for dividend payments. The Preferred Stock has a liquidation preference of \$10 per share, plus accrued and unpaid dividends.

The Preferred Stock is convertible at the option of the holder at any time into the Company's shares of common stock at a conversion rate of approximately 0.42553 shares of common stock for each share of Preferred Stock based on a price of \$23.50. During 2010, 833,671 shares of Preferred Stock were converted into 1,655,599 shares of the Company's common stock, which is described in more detail below. Since inception through June 30, 2011, holders have voluntarily converted 1,776,858 shares of Preferred Stock into common stock. The Company has reserved 516,228 shares of common stock for issuance upon conversion of the remaining shares of Preferred Stock outstanding at June 30, 2011. The converted shares of Preferred Stock have been retired and canceled and shall upon cancellation be restored to the status of authorized but unissued shares of preferred stock, subject to reissuance by the Board of Directors as shares of Preferred Stock of one or more series.

The Company may automatically convert the Preferred Stock into common stock if the closing price of the Company's common stock has exceeded \$35.25, which is 150% of the conversion price of the Preferred Stock, for at least 20 trading days during any 30-day trading period, ending within five trading days prior to notice of automatic conversion.

The Certificate of Designations governing the Preferred Stock provides that if the Company fails to pay dividends on its Preferred Stock for six quarterly periods, holders of Preferred Stock are entitled to nominate and elect two directors to the Company's Board of Directors. This right accrued to the holders of Preferred Stock as of August 2, 2010 and two directors were nominated and elected at the annual meeting held on May 24, 2011.

The Preferred Stock has no maturity date and no voting rights prior to conversion into common stock, except under limited circumstances.

From November 6, 2007, the Company may, at its option, redeem the Preferred Stock in whole or in part, out of funds legally available at the redemption prices per share stated below, plus an amount equal to accrued and unpaid dividends up to the date of redemption:

Year from November 1, 2010 to October 31, 2011	\$ 10.24
Year from November 1, 2011 to October 31, 2012	\$ 10.18
Year from November 1, 2012 to October 31, 2013	\$ 10.12
Year from November 1, 2013 to October 31, 2014	\$ 10.06
November 1, 2014 and thereafter	\$ 10.00

The Preferred Stock is exchangeable, in whole but not in part, at the option of the Company on any dividend payment date beginning on November 1, 2005 (the "Exchange Date") for the Company's 6% Convertible Subordinated Debentures ("Debentures") at the rate of \$10 principal amount of Debentures for each share of Preferred Stock. The Debentures, if issued, will mature 25 years after the Exchange Date and have terms substantially similar to those of the Preferred Stock.

Conversion of Convertible Preferred Stock

During the first and second quarters of 2010, Cyclacel entered into agreements to exchange shares of the Company's Preferred Stock into shares of common stock. There were no such conversions for the six months ended June 30, 2011. The table below provides details of the aggregate activities in 2010:

	For the three months ended June 30, 2010	For the six months ended June 30, 2010
Preferred shares exchanged	710,271	833,671
Common shares issued:		
At stated convertible option	302,242	354,752
Incremental shares issued under an inducement offer	1,113,961	1,300,847
Total common shares issued	1,416,203	1,655,599

ASC 260, EITF Topic D-42, "The Effect of the Calculation of Earnings per Share for the Redemption or Induced Conversion of Preferred Stock" ("ASC 260") requires that if convertible preferred stock is converted to other securities pursuant to an inducement offer, the Company should record the excess of (1) the fair value of all securities and other consideration transferred to the holders of the convertible preferred stock and (2) the fair value of securities issuable to the original conversion terms as an increase to net loss to arrive at a net loss attributable to common shareholders. The stockholder received more shares of the Company's common stock than would have been delivered pursuant to the original conversion terms of the Preferred Stock, pursuant to a short term inducement offer. The excess of the fair value of the common stock transferred to the stockholder over the carrying amount of the preferred stock in the Company's balance sheet at the time of the transfer was considered to be an additional return to the holder of the Preferred Stock. Specifically, the Company recorded deemed dividends related to the additional shares issued under the exchange transactions of approximately \$2.5 million and \$2.9 million for the three and six months ended June 30, 2010, respectively.

Common Stock

October 2010 Private Placement

On October 7, 2010, the Company completed a private placement pursuant to which it sold approximately \$15.2 million of its units to several institutional investors, for net proceeds of approximately \$14.0 million. The units consist of one share of common stock and 0.5 of a warrant, with each whole warrant representing the right to purchase one share of common stock at an exercise price of \$1.92 per share for a period of five years. As of June 30, 2011, all options and warrants issued to the investors are outstanding and have been classified as equity. The investors purchased a total of 8,323,190 units at a price of \$1.82625 per unit. The investors also have the right to acquire up to 4,161,595 additional units at a price of \$1.67 per unit (for \$6.9 million in gross proceeds) at any time up to nine months after closing or by July 6, 2011. As of June 30, 2011, none of the additional units had been exercised and, as of July 6, 2011, the right to acquire the additional units lapsed. The transaction date fair value of the warrants and additional optional units was \$5.1 million and \$2.8 million, respectively. Net proceeds of approximately \$14.0 million were allocated based on relative transaction date fair values in the following manner: \$8.9 million (\$1.07 per share), \$3.3 million (\$0.79 per warrant) and \$1.8 million (\$0.43 per optional unit) to common shares, warrants and the additional optional units, respectively.

In connection with the October 2010 private placement, the Company granted to the investors certain registration rights pursuant to a Registration Rights Agreement, dated October 7, 2010, in which the Company agreed, among other things, to register all of the shares of common stock acquired from the Company (including upon exercise of the warrants and/or the options) within thirty calendar days after the Company becomes eligible to use a registration statement on Form S-3, and use commercially reasonable efforts to have the registration statement declared effective as promptly as practicable thereafter. Upon the Company's failure to comply with the terms of the Registration Rights Agreement and certain other conditions, the Company will be required to make pro rata payments to each investor, as liquidated damages, in an amount equal to 1.5% of the aggregate purchase price paid by such investor. The Company also agreed to other customary obligations regarding registration, including indemnification and maintenance of the effectiveness of the registration statement. The Company is currently in compliance with the applicable terms of the Registration Rights Agreement, and the securities that were registrable under the terms of the Registration Rights Agreement are currently subject to an effective registration statement.

January 2010 Registered Direct Financings

On January 25, 2010, the Company completed the sale of 2,350,000 units in a "registered direct" offering at a purchase price of \$2.50 per unit to certain institutional investors of the Company for gross proceeds of approximately \$5.9 million. Each unit consisted of one share of the Company's common stock and one warrant to purchase 0.30 of one share of its common stock. The warrants have a five-year term from the date of issuance, are exercisable beginning six months from the date of issuance at an exercise price of \$2.85 per share of common stock. As of June 30, 2011, warrants issued to the investors have been classified as equity. The transaction date fair value of the warrants of \$1.0 million was determined utilizing the Black-Scholes option pricing model utilizing the following assumptions: risk free interest rate — 2.39%, expected volatility — 90%, expected dividend yield — 0%, and a remaining contractual life of 5.00 years. As of June 30, 2011, all the warrants are outstanding. Net proceeds of approximately \$5.4 million were allocated based on relative transaction date fair values in the following manner: \$4.5 million (\$1.93 per share) to common shares and \$0.9 million (\$1.29 per warrant) to the warrants.

On January 13, 2010, the Company completed the sale of 2,850,000 units in a "registered direct" offering to certain institutional investors. Each unit was sold at a purchase price of \$2.51 per unit and consists of one share of the Company's common stock and one warrant to purchase 0.25 of one share of its common stock for gross proceeds of approximately \$7.2 million. The warrants have a five-year term from the date of issuance, are exercisable beginning six months from the date of issuance at an exercise price of \$3.26 per share of common stock. As of June 30, 2011, warrants issued to the investors have been classified as equity. The transaction date fair value of the warrants of \$1.3 million was determined utilizing the Black-Scholes option pricing model utilizing the following assumptions: risk free interest rate — 2.55%, expected volatility — 90%, expected dividend yield — 0%, and a remaining contractual life of 5.00 years. As of June 30, 2011, all the warrants are outstanding. Net proceeds of approximately \$6.5 million were allocated based on relative transaction date fair values in the following manner: \$5.6 million (\$1.95 per share) to common shares and \$0.9 million (\$1.32 per warrant) to the warrants.

July 2009 Registered Direct Financing

On July 29, 2009, the Company sold its securities to select institutional investors consisting of 4,000,000 units in a "registered direct" offering at a purchase price of \$0.85 per unit. Each unit consisted of (i) one share of the Company's common stock, (ii) one warrant to purchase 0.625 of one share of common stock (a "Series I Warrant") and (iii) one warrant to purchase 0.1838805 of one share of common stock (a "Series II Warrant"). The Series I Warrants had a seven-month term from the date of issuance, were exercisable beginning six months from the date of issuance at an exercise price of \$1.00 per share of common stock. During the first quarter of 2010, all of the Series I Warrants were exercised for \$2.5 million. The Series II Warrants have a five-year term from the date of issuance, are exercisable beginning six months from the date of issuance at an exercise price of \$1.00 per share of common stock. During the first quarter of 2010, 43,266 common shares were issued upon exercise of Series II Warrants with proceeds of \$43,266. There were no exercises during the six months ended June 30, 2011.

The net proceeds to the Company from the sale of the units, after deducting for the placement agent's fees and offering expenses, were approximately \$2.9 million. As of June 30, 2011, the remaining Series II Warrants outstanding and exercisable into 692,256 of the Company's shares of common stock have been classified as equity. The transaction date fair value of the Series II Warrants of \$0.6 million was determined utilizing the Black-Scholes option pricing model utilizing the following assumptions: risk free interest rate — 2.69%, expected volatility — 90%, expected dividend yield — 0%, and a remaining contractual life of 5.00 years.

December 2007 Committed Equity Financing Facility

On December 10, 2007 and as amended on November 24, 2009, Cyclacel entered into a Committed Equity Financing Facility, or CEFF, with Kingsbridge, in which Kingsbridge committed to purchase the lesser of 4,084,590 shares of common stock or \$60 million of common stock from Cyclacel over a three-year period. The CEFF lapsed on December 10, 2010.

During March 2010, the Company sold 1,563,208 shares of its common stock to Kingsbridge under the CEFF, in consideration of aggregate proceeds of \$3.1 million. During December 2009 and January 2010, the Company sold an aggregate of 1,583,626 shares of its common stock to Kingsbridge under the terms of the CEFF in consideration of an aggregate of \$1.3 million, of which approximately \$1.0 million was received in 2009, with the balance of \$0.3 million in respect of common shares subscribed but unissued at December 31, 2009, received by the Company in January 2010.

Common Stock Warrants

In connection with the Company's February 16, 2007 "registered direct" offering the Company issued to investors warrants to purchase 1,062,412 shares of common stock. The warrants issued to the investors are being accounted for as a liability. At the date of the transaction, the fair value of the warrants of \$6.8 million was determined utilizing the Black-Scholes option pricing model utilizing the following assumptions: risk free interest rate — 4.58%, expected volatility - 85%, expected dividend yield — 0%, and a remaining contractual life of 6.88 years. The value of the warrant is being marked to market each reporting period as a derivative gain or loss on the consolidated statement of operations until exercised or expiration. See "Note 3 — Fair Value" for further details.

The following table summarizes information about warrants outstanding at June 30, 2011:

Issued in Connection With	Expiration Date	Common Shares Issuable	Av Ex	eighted verage xercise Price
April 2006 stock issuance	2013	2,571,429	\$	7.00
February 2007 stock issuance	2014	1,062,412	\$	8.44
December 2007 CEFF	2013	100,000	\$	1.40
July 2009 Series II stock issuance	2014	692,256	\$	1.00
January 2010 stock issuance	2015	712,500	\$	3.26
January 2010 stock issuance	2015	705,000	\$	2.85
October 2010 stock issuance	2015	4,161,595	\$	1.92
Total		10,005,192	\$	4.01

Exercise of Stock Options

During the six months ended June 30, 2011, there were 5,131 stock option exercises totaling approximately \$3,000. During the three months ended June 30, 2010, there were 124,277 stock option exercises totaling approximately \$0.1 million.

9. SUBSEQUENT EVENTS

Preferred Stock Dividend

On July 8, 2011, the Board of Directors decided not to declare the quarterly cash dividend on the Preferred Stock with respect to the second quarter of 2011 that would have otherwise been payable on August 1, 2011.

The Board also did not declare the quarterly cash dividend with respect to each of the four quarters of fiscal year 2009 and the first, second and third quarters of 2010. To the extent that any dividends payable on the Preferred Stock are not paid, such unpaid dividends are added to the liquidity preference of the Preferred Stock. As the Company failed to pay in an aggregate amount equal to at least six quarterly dividends (whether or not consecutive) on the Preferred Stock, the size of the Company's Board was increased by two members and the holders of the Preferred Stock, voting separately as a class, voted on May 24, 2011 and elected two directors to fill the vacancies created thereby, which directorships shall terminate when the Company pays all accrued but unpaid dividends.

License Agreement

On July 11, 2011, the Company entered into an amendment to its license agreement with Daiichi Sankyo Company, Limited ("Daiichi Sankyo"), dated September 10, 2003, relating to certain rights which the Company has licensed from Daiichi Sankyo with regard to the Company's sapacitabine drug. Effective July 11, 2011, the license agreement was amended to irrevocably waive a termination right Daiichi Sankyo possessed under a provision of the agreement that required the Company to obtain regulatory approval to sell sapacitabine in at least one country by September 2011, and releases the Company from all claims and liability of any kind arising under such provision. The amendment further provides that the royalty due from the Company to Daiichi Sankyo on future net sales of sapacitabine be increased by a percentage between 1.25% and 1.50% depending on the level of net sales of sapacitabine realized.

Financing

On July 7, 2011 the Company closed an underwritten offering for an aggregate of 7,617,646 units, at an offering price of \$1.36 per unit, for gross proceeds of approximately \$10.4 million. Each unit consists of (i) one share of common stock and (ii) a five-year warrant to purchase 0.5 of a share of common stock at an exercise price of \$1.36 per share, exercisable beginning six months after the date of issuance. The shares of common stock and warrants were immediately separable and were issued separately such that no units were issued. The net proceeds, after underwriting discounts and commissions and other fees and expenses payable by the Company, were approximately \$9.3 million.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

CAUTIONARY STATEMENT REGARDING FORWARD-LOOKING STATEMENTS

This Quarterly Report on Form 10-Q, including, without limitation, Management's Discussion and Analysis of Financial Condition and Results of Operations, contains "forward-looking statements" within the meaning of Section 27A of the Securities Act and Section 21E of the Securities Exchange Act of 1934, as amended (the "Exchange Act"). We intend that the forward-looking statements be covered by the safe harbor for forward-looking statements in the Exchange Act. The forward-looking information is based on various factors and was derived using numerous assumptions. All statements, other than statements of historical fact, that address activities, events or developments that we intend, expect, project, believe or anticipate will or may occur in the future are forward-looking statements. Such statements are based upon certain assumptions and assessments made by our management in light of their experience and their perception of historical trends, current conditions, expected future developments and other factors they believe to be appropriate. These forward-looking statements are usually accompanied by words such as "believe," "anticipate," "plan," "seek," "expect," "intend" and similar expressions.

Forward-looking statements necessarily involve risks and uncertainties, and our actual results could differ materially from those anticipated in the forward looking statements due to a number of factors, including those set forth in Part I, Item 1A, entitled "Risk Factors," of our Annual Report on Form 10-K for the year ended December 31, 2010, as updated and supplemented by Part II, Item 1A, entitled "Risk Factors," of our Quarterly Reports on Form 10-Q and other reports we publicly file with the Securities and Exchange Commission, or the SEC. These factors as well as other cautionary statements made in this Quarterly Report on Form 10-Q, should be read and understood as being applicable to all related forward-looking statements wherever they appear herein. The forward-looking statements contained in this Quarterly Report on Form 10-Q represent our judgment as of the date hereof. We encourage you to read those descriptions carefully. We caution you not to place undue reliance on the forward-looking statements contained in this report. These statements, like all statements in this report, speak only as of the date of this report (unless an earlier date is indicated) and we undertake no obligation to update or revise the statements except as required by law. Such forward-looking statements are not guarantees of future performance and actual results will likely differ, perhaps materially, from those suggested by such forward-looking statements. In this report, "Cyclacel," the "Company," "we," "us," and "our" refer to Cyclacel Pharmaceuticals, Inc.

Overview

We are a biopharmaceutical company dedicated to the development and commercialization of novel, mechanism-targeted drugs to treat human cancers and other serious diseases. Our strategy is to build a diversified biopharmaceutical business focused in hematology and oncology based on a portfolio of commercial products and a development pipeline of novel drug candidates.

Our clinical development priorities are focused on sapacitabine in the following indications:

- Acute myeloid leukemia, or AML, in the elderly;
- · Myelodysplastic syndromes, or MDS; and
- Non-small cell lung cancer, or NSCLC.

On January 11, 2011, the Company opened enrollment of the SEAMLESS pivotal Phase 3 trial for the Company's sapacitabine oral capsules as a front-line treatment of elderly patients aged 70 years or older with newly diagnosed AML who are not candidates for intensive induction chemotherapy. SEAMLESS is a registration-directed clinical trial and is conducted under a Special Protocol Assessment, or SPA, agreement with the U.S. Food and Drug Administration, or FDA. SEAMLESS is a randomized study against an active control drug with the primary objective of demonstrating an improvement in overall survival. SEAMLESS has two parts. The first part is a pilot study of a single arm of the trial. The study's Data Safety Monitoring Board, or DSMB, will review the data from the pilot study before we are allowed to enroll patients in the second part in which patients will be randomized either to an active or control treatment.

The Company has advanced two additional product candidates, seliciclib in Phase 2 for NSCLC and nasopharyngeal cancer or NPC, and CYC116 in Phase 1 clinical development. The combination of sapacitabine with seliciclib is also being evaluated in a Phase 1 clinical trial. The Company will determine the feasibility of pursuing further development and/or partnering these assets depending on the availability of funding and further clinical data. In addition, we market directly in the United States Xclair® Cream for radiation dermatitis and Numoisyn® Liquid and Numoisyn® Lozenges for xerostomia.

Recent Developments

On June 6, 2011, we announced interim results from an ongoing, multicenter, Phase 1/2 clinical trial examining the safety and effectiveness of oral sapacitabine, our lead product candidate, administered sequentially with decitabine. Thirty-day mortality from all causes was 4.5%; 60-day mortality from all causes was 9.5%. The overall response rate was 34.8%. An additional 26.1% of patients stayed on study for more than 4 cycles with a decrease in bone marrow blast counts despite not meeting criteria of response. Approximately 60.9% of patients received 4 or more cycles of the regimen.

The data was reported during a poster session at the 2011 American Society of Clinical Oncology, or ASCO, Annual Meeting in Chicago, Illinois. As reported at ASCO, no dose-limiting toxicities were observed in 21 patients treated with the regimen who have had at least 60 days of follow-up. The median age in the group is 76 years (range 72-88). Common adverse events regardless of cause included anemia, anorexia, dehydration, diarrhea, dyspnea, edema, hypocalcemia, nausea, febrile neutropenia, neutropenia, pneumonia, thrombocytopenia, and weakness, which were mostly moderate in intensity.

We believe, based on the interim results reported at ASCO that the sequential administration of sapacitabine and decitabine appears safe and active in elderly patients with newly diagnosed acute myeloid leukemia, or AML.

The treatment regimen under evaluation in this pilot study is being used as one of the arms in SEAMLESS, the registration-directed, Phase 3 study of sapacitabine in elderly patients with newly diagnosed AML who are not candidates for or have refused induction chemotherapy. SEAMLESS is being conducted under a Special Protocol Assessment agreement that Cyclacel reached with the US Food and Drug Administration.

Our pipeline and expertise in cell cycle biology

Our core area of expertise is in cell cycle biology and we focus primarily on the development of orally-available anticancer agents that target the cell cycle with the aim of slowing the progression or shrinking the size of tumors, and enhancing the quality of life and improving survival rates of cancer patients. We are generating several families of anticancer drug candidates that act on the cell cycle including nucleoside analogues, cyclin dependent kinase, or CDK inhibitors and Aurora kinase/Vascular Endothelial Growth Factor Receptor 2, or AK/VEGFR2 inhibitors. Although a number of pharmaceutical and biotechnology companies are currently attempting to develop nucleoside analogues, CDK inhibitor and AK inhibitor drugs, we believe that our drug candidates are differentiated in that they are orally-available and interact with unique target profiles and mechanisms. For example we believe that our sapacitabine is the only orally-available nucleoside analogue to be tested in a Phase 3 trial for AML and in a Phase 2 trial in MDS and seliciclib is the most advanced orally-available CDK inhibitor in Phase 2 trials.

Although our resources are primarily directed towards advancing our anticancer drug candidate sapacitabine through in-house development activities we are also progressing, but with minimal investment, our other novel drug series which are at earlier stages. As a consequence of our progress in sapacitabine clinical development, research and development expenditures for the six months ended June 30, 2011 increased by \$1.5 million, or 42%, to \$5.0 million, compared to \$3.5 million for the six months ended June 30, 2010.

We have retained rights to commercialize our clinical development candidates and our business strategy is to enter into selective partnership arrangements with these programs.

Our corporate headquarters is located in Berkeley Heights, New Jersey, with a research facility located in Dundee, Scotland.

From our inception in 1996 through June 30, 2011, we have devoted substantially all our efforts and resources to our research and development activities. We have incurred significant net losses since inception. As of June 30, 2011, our accumulated deficit during the development stage was approximately \$250.0 million. We expect to continue incurring substantial losses for the next several years as we continue to develop our clinical and preclinical drug candidates. Our operating expenses are comprised of research and development expenses and selling and general and administrative expenses.

To date, we have not generated significant product revenue but have financed our operations and internal growth through, among other things, public and private equity offerings, licensing revenue, interest on investments, government grants and research and development tax credits. Prior to October 2007, our revenue consisted of collaboration and grant revenue. Beginning in 2008, we recognized revenue from sales of commercial products, for the first time, following the ALIGN acquisition in October 2007. We have recognized revenues from inception through June 30, 2011 totaling approximately \$9.4 million, of which approximately \$2.7 million is derived from product sales, approximately \$3.1 million from fees under collaborative agreements and approximately \$3.6 million of grant revenue from various United Kingdom government grant awards.

Subsequent Events

Preferred Stock Dividend

On July 8, 2011, our Board of Directors, or Board, decided not to declare the quarterly cash dividend on our 6% Convertible Exchangeable Preferred Stock, or Preferred Stock, with respect to the second quarter of 2011 that would have otherwise been payable on August 1, 2011.

The Board also did not declare the quarterly cash dividend with respect to each of the four quarters of fiscal year 2009 and the first, second and third quarters of 2010. To the extent that any dividends payable on the Preferred Stock are not paid, such unpaid dividends are added to the liquidity preference of the Preferred Stock. As we failed to pay in an aggregate amount equal to at least six quarterly dividends (whether or not consecutive) on the Preferred Stock, the size of the our Board was increased by two members and the holders of the Preferred Stock, voting separately as a class, voted on May 24, 2011 and elected two directors to fill the vacancies created thereby, which directorships shall terminate when we pay all accrued but unpaid dividends.

License Agreement

On July 11, 2011, we entered into an amendment to our license agreement with Daiichi Sankyo Company, Limited ("Daiichi Sankyo"), dated September 10, 2003, relating to certain rights which we have licensed from Daiichi Sankyo with regard to sapacitabine. Effective July 11, 2011, the license agreement was amended to irrevocably waive a termination right Daiichi Sankyo possessed under a provision of the agreement that required us to obtain regulatory approval to sell sapacitabine in at least one country by September 2011, and releases us from all claims and liability of any kind arising under such provision. The amendment further provides that the royalty due to Daiichi Sankyo on future net sales of sapacitabine be increased by a percentage between 1.25% and 1.50% depending on the level of net sales of sapacitabine realized.

Financing

On July 7, 2011, we closed an underwritten offering for an aggregate of 7,617,646 units, at an offering price of \$1.36 per unit, for gross proceeds of \$10.4 million. Each unit consists of (i) one share of common stock and (ii) a five-year warrant to purchase 0.5 of a share of common stock at an exercise price of \$1.36 per share, exercisable beginning six months after the date of issuance. The shares of common stock and warrants were immediately separable and were issued separately such that no units were issued. The net proceeds, after underwriting discounts and commissions and fees and expenses, were approximately \$9.3 million.

Results of Operations

Three Months Ended June 30, 2011 and 2010

Revenues

The following table summarizes the components of our revenues for the three months ended June 30, 2010 and 2011:

	Three Months Ended June 30, Increase (D						Decrease)	
	2	010	2	011		\$	%	
			(\$0	000s)			_	
Collaboration and research and development revenue	\$	100	\$	_	\$	(100)	(100)%	
Product revenue		19		168		149	784%	
Total revenue	\$	119	\$	168	\$	49	41%	

We recognized \$0.1 million of collaboration and research and development revenue for the three months ended June 30, 2010, derived from an agreement with a pharmaceutical company under which we provided one of our compounds for evaluation in the field of eye care. We had no collaboration and research and development revenue for the three months ended June 30, 2011.

Product revenue is derived from the sale of Xclair® Cream, Numoisyn® Liquid and Numoisyn® Lozenges following the ALIGN asset acquisition on October 5, 2007. During the three months ended June 30, 2010 and 2011, we recognized approximately \$19,000 and \$0.2 million, respectively in accordance with our revenue recognition policy. Product revenue was lower by \$0.2 million for the three months ended June 30, 2010 compared to the same period in 2011 as a result of recognizing \$0.2 million in product returns during the three months ended June 30, 2010.

Grant revenue is recognized as we incur and pay for qualifying costs and services under the applicable grant. Grant revenue is primarily derived from various United Kingdom government grant awards. We have earned grant revenue in the past and will continue to pursue opportunities in the future but did not recognize any grant revenue for the three months ended June 30, 2010 or 2011 as a result of finalization of a three-year European Union grant in the first quarter of 2010.

The future

We expect to continue to maintain the sales of ALIGN products in 2011 through the support of a small sales and marketing infrastructure. We do not expect to earn any grant revenue during the remainder of year, as no awards are expected during the year ended December 31, 2011.

Cost of goods sold

The following table summarizes cost of goods sold for the six months ended June 30, 2010 and 2011:

	T	hree Moi	nths End	led			
		June 30,				Increase (De	crease)
	20	10	2011		2011 \$		%
			(\$0	00s)			_
Cost of goods sold	\$	92	\$	72	\$	(20)	(22)%

Total cost of sales represented 43% of product revenue for the three months ended June 30, 2011. Total cost of sales was not a meaningful percentage of product revenue for the three months ended June 30, 2010 because of the low product revenue recognized during the period, as explained above.

Research and development expenses

We expense all research and development costs as they are incurred. Research and development expenses primarily include:

- clinical trial and regulatory-related costs;
- payroll and personnel-related expenses, including consultants and contract research;
- preclinical studies and laboratory supplies and materials;
- · technology license costs; and
- rent and facility expenses for our laboratories.

The following table provides information with respect to our research and development expenditure for the three months ended June 30, 2010 and 2011:

	Three Months Ended									
	June 30,					Increase (Decrease)				
Sapacitabine		2010	2011 (\$000s)		\$		<u>%</u>			
	\$	1,186	\$	1,760	\$	574	48%			
Seliciclib		72		16		(56)	(78)%			
Other research and development costs		64		83		19	30%			
Total research and development costs	\$	1,322	\$	1,859	\$	537	41%			

Total research and development expenses represented 29% and 47% of our operating expenses for the three months ended June 30, 2010 and 2011, respectively.

Research and development expenditure increased by \$0.5 million to \$1.9 million for the three month period ended June 30, 2011 from \$1.3 million for the three month period ended June 30, 2010. The increase in costs of \$0.5 million is mostly due to a \$0.6 million increase in sapacitabine-related expenses, primarily a \$0.2 million increase in capsule manufacture and testing and a \$0.2 million increase in clinical trial expenses, as a result of the SEAMLESS trial entering Phase 3. Seliciclib costs decreased \$56,000, from \$72,000 for the three months ended June 30, 2010, to \$16,000 for the three months ended June 30, 2011, primarily due to the completion of clinical trials. Other research and development costs remained relatively flat, increasing \$19,000 from approximately \$0.1 million for the three months ended June 30, 2010 to approximately \$0.1 million for three months ended June 30, 2011 as we have continued to concentrate financial resources on the development of sapacitabine and selectively invest in other compounds.

The future

We will continue to concentrate our resources on the development of sapacitabine. We anticipate that overall research and development expenditures in 2011 will increase as we enroll the SEAMLESS pivotal Phase 3 trial.

Selling, general and administrative expenses

Selling, general and administrative expenses include costs for sales and marketing and administrative personnel, legal and other professional expenses and general corporate expenses. The following table summarizes the selling, general and administrative expenses for the three months ended June 30, 2010 and 2011:

	Three Months Ended								
		June 30,				Increase (Decrease)			
		2010 2011			\$	%			
			(5	\$000s)					
Selling, general and administrative	\$	3,091	\$	2,034	\$	(1,057)	(34)%		

Total selling, general and administrative expenses represented 69% and 51% of our operating expenses for the three months ended June 30, 2010 and 2011, respectively. Our selling, general and administrative expenditure decreased by \$1.1 million from \$3.1 million for the three months ended June 30, 2010 to \$2.0 million for the three months ended June 30, 2011. The decrease of \$1.1 million in expenses was primarily attributable to a net decrease in professional and consultancy costs of \$0.8, million, a decrease in stock compensation of \$0.4 million, a decrease in rent expense of \$0.1 million, a decrease in salaries of \$0.1 million, and an increase in patent costs of \$0.3 million.

The future

We expect our selling, general and administrative expenditures in 2011 to remain at the same level or to be less than our expenditures in 2010.

Other income (expense)

The following table summarizes other income (expense) for the three months ended June 30, 2010 and 2011:

		Three Mon	ths En	ded				
	June 30,					Increase (Decrease)		
	2010		2	2011		\$	%	
				(\$00	00s)			
Other income (expense):								
Change in valuation of warrants	\$	273	\$	125	\$	(148)	(54)%	
Foreign exchange (losses)/gains		(49)		(19)		30	(61)%	
Interest income		8		13		5	63%	
Interest expense		(9)		_		9	(100)%	
Total other income (expense)	\$	223	\$	119	\$	(104)	(47)%	

Total other income and expense, net, decreased by approximately \$0.1 million, from income of \$0.2 million for the three months ended June 30, 2010, to income of \$0.1 million for the three months ended June 30, 2011. The most significant impact is the change in the valuation of the warrant liability described below.

The change in valuation of warrants relates to the issue of warrants to purchase shares of our common stock under the registered direct financing completed in February 2007. The warrants issued to the investors meet the requirements of and are being accounted for as a liability in accordance with ASC 480, "Accounting for Derivative Financial Instruments Indexed to, and Potentially Settled in, a Company's Own Stock." The fair value of the warrants is re-measured each reporting period with a derivative gain or loss recognized in the consolidated statement of operations. Such gains or losses will continue to be reported until the warrants are exercised or expired. For the three months ended June 30, 2010 and 2011, the change in the value of warrants was a \$0.3 million increase and a \$0.1 million decrease, respectively.

Foreign exchange gains (losses) decreased by \$30,000 to a loss of \$19,000 for the three months ended June 30, 2011 compared to a loss of \$49,000 for the three months ended June 30, 2010. Foreign exchange gains/(losses) are reported in the consolidated statement of operations as a separate line item within other income (expense).

Interest income increased by approximately \$5,000 to \$13,000 for the three months to June 30, 2010 from \$8,000 for the three months ended June 30, 2010. This is mostly attributed to a higher average daily balance of cash and cash equivalents during the three months ended June 30, 2011 compared to the same period in 2010.

Interest expense decreased by \$9,000 from \$9,000 for the three months ended June 30, 2010 to \$0 for the three months ended June 30, 2011. This reduction was primarily due to the elimination of the accretion expense associated with the restructured Bothell lease, which expired in December 2010.

The future

The valuation of the warrant liability will continue to be re-measured at the end of each reporting period. The valuation of the warrants is dependent upon many factors, including our stock price, interest rates and the remaining term of the instrument and may fluctuate significantly, which may have a significant impact on our statement of operations.

Income tax benefit

Credit is taken for research and development tax credits, which are claimed from the United Kingdom's revenue and customs authority, or HMRC, in respect of qualifying research and development costs incurred.

The following table summarizes research and development tax credits for the three months ended June 30, 2010 and 2011:

		Three Months Ended								
		Jun	e 30,		Increase (Decrease)					
	2	2010 2011		\$		%				
			(\$0	000s)						
Income tax benefit	\$	230	\$	126	\$	(104)	(45)%			

Research and development tax credits recoverable decreased by \$0.1 million to \$0.1 million for the three months ended June 30, 2011 from \$0.2 million for three months ended June 30, 2010. The level of tax credits recoverable is linked directly to qualifying research and development expenditure incurred in any one year, but restricted to payroll taxes paid by us in the United Kingdom in that same year. The \$0.1 million decrease between the three months ended June 30, 2010 and 2011 is due to lower payroll taxes paid as a result of lower headcount in 2011.

The future

We expect to continue to be eligible to receive United Kingdom research and development tax credits for the foreseeable future and will elect to do so.

Six Months Ended June 30, 2010 and 2011

Revenues

The following table summarizes the components of our revenues for the six months ended June 30, 2010 and 2011:

	Six Months Ended									
	June 30,					Increase (Decrease)				
	2	010	2011 (\$000s)				<u>%</u>			
Collaboration and research and development revenue	\$	100	\$	_	\$	(100)	(100)%			
Product revenue		273		360		87	32%			
Grant revenue		16		_		(16)	(100)%			
Total revenue	\$	389	\$	360	\$	(29)	(7)%			

We recognized \$0.1 million of collaboration and research and development revenue for the six months ended June 30, 2010 derived from an agreement with a pharmaceutical company under which we provided one of our compounds for evaluation in the field of eye care. We had no collaboration and research and development revenue for the six months ended June 30, 2011.

Product revenue is derived from the sale of Xclair® Cream, Numoisyn® Liquid and Numoisyn® Lozenges. During the six months ended June 30, 2010 and 2011, we recognized product revenue of approximately \$0.3 and \$0.4 million, respectively, in accordance with our revenue recognition policy. Product revenue was lower by \$0.1 million for the six months ended June 30, 2010 compared to the same period in 2011 due to \$0.2 million in product returns during six month ended June 30, 2011.

Grant revenue is recognized as we incur and pay for qualifying costs and services under the applicable grant. Grant revenue is primarily derived from various United Kingdom government grant awards. We recognized \$16,000 in grant revenue for the six months ended June 30, 2010. We did not recognize any grant revenue for the six months ended June 30, 2011 as a result of finalization of a three-year European Union grant in the first quarter of 2010.

Cost of goods sold

The following table summarizes cost of goods sold for the six months ended June 30, 2010 and 2011:

		Six Mont	hs Ende	ed				
		June 30,				Increase (Decrease)		
	20	2010 2011		\$		%		
			(\$0	000s)				
Cost of goods sold	\$	234	\$	178	\$	(56)	(24)%	

Total cost of sales represented 86% and 49% of product revenue for the six months ended June 30, 2010 and 2011, respectively. The high percentage for the six months ended June 30, 2010 is the result of lower product revenue driven by \$0.2 million of product returns during the period, as described above.

Research and development expenses

We expense all research and development costs as they are incurred. Research and development expenses primarily include:

- clinical trial and regulatory-related costs;
- payroll and personnel-related expenses, including consultants and contract research;
- preclinical studies and laboratory supplies and materials;
- technology license costs; and
- rent and facility expenses for our laboratories.

The following table provides information with respect to our research and development expenditure for the six months ended June 30, 2010 and 2011:

	Six Months Ended									
	June 30,					Increase (Decrease)				
Sapacitabine		2010		2011 (\$000s)		\$				
	\$	2,803	\$	4,731	\$	1,928	69%			
Seliciclib		133		25		(108)	(81)%			
Other research and development costs		561		183		(378)	(67)%			
Total research and development expenses	\$	3,497	\$	4,939	\$	1,442	41%			

Total research and development expenses represented 38% and 55% of our operating expenses for the six months ended June 30, 2010 and 2011, respectively.

Research and development expenditure increased by \$1.4 million to \$4.9 million for the six month period ended June 30, 2011 from \$3.5 million for the six month period ended June 30, 2010. The increase in costs of \$1.4 million is primarily due to a \$1.9 million increase in sapacitabine related costs and decreases of \$0.1 million and \$0.4 million in seliciclib and other research and development costs, respectively, as we continue to focus on the development of sapacitabine. The \$1.9 million increase in sapacitabine expenditures was primarily due to \$1.6 million of contractual expenses, resulting from an achievement of a milestone triggered by the opening of enrollment in our SEAMLESS trial, pursuant to the Daiichi Sankyo license under which we license certain patent rights for sapacitabine, and a \$0.4 million increase in capsule manufacturing and testing. Seliciclib costs decreased by \$0.1 million from \$133,000 for the six months ended June 30, 2010 to \$25,000 for six months ended June 30, 2011 primarily due to the completion of clinical trials. Other research and development costs decreased \$0.4 million to \$0.2 million for the six months ended June 30, 2011 from \$0.6 million for the six months ended June 30, 2010, as we have concentrated financial resources on the development of sapacitabine and reduced investment in other compounds.

Selling, general and administrative expenses

Selling, general and administrative expenses include costs for sales and marketing and administrative personnel, legal and other professional expenses and general corporate expenses. The following table summarizes the selling, general and administrative expenses for the six months ended June 30, 2010 and 2011:

	Six Months Ended								
	Jun	e 30,		Increase (Decrease)					
	 2010	2011		\$		%			
	 	(5	\$000s)						
Selling, general and administrative	\$ 5,491	\$	3,840	\$	(1,651)	(30)%			

Total selling, general and administrative expenses represented 60% and 43% of our operating expenses for the six months ended June 30, 2010 and 2011, respectively. Our selling, general and administrative expenditure decreased by \$1.7 million from \$5.5 million for the six months ended June 30, 2010 to \$3.8 million for the six months ended June 30, 2011. The decrease of \$1.7 million in expenses was primarily attributable to a net decrease in professional and consultancy costs of \$1.3 million, a decrease in salaries of \$0.4 million, a decrease in rent of \$0.2 million as a result of the expiration our lease on a facility in Bothell, Washington in December 2010, and a decrease in stock compensation of \$0.3 million. These amounts were partially offset by a \$0.4 million increase in patent-related costs.

Other income (expense)

The following table summarizes other income (expense) for the three months ended June 30, 2010 and 2011:

Six Months Ended

June 30,				Increase (Decrease)			
	2010	2011		\$		%	
(\$000s)							
\$	(516)	\$	203	\$	719	139%	
	(38)		(87)		(49)	(129)%	
	17		24		7	41%	
	(33)		_		33	(100)%	
\$	(570)	\$	140	\$	710	(125)%	
	\$	\$ (516) (38) 17 (33)	\$ (516) \$ (38) 17 (33)	2010 2011 (\$000s) \$ (516) \$ 203 (38) (87) 17 24 (33) —	2010 2011 (\$000s) \$ (516) \$ 203 \$ (38) (87) 17 24 (33) —	2010 2011 (\$000s) \$ \$ (516) \$ 203 \$ 719 (38) (87) (49) 17 24 7 (33) — 33	

Total other income and expense, net, increased by approximately \$0.7 million, from an expense of \$0.6 million for the six months ended June 30, 2010, to income of \$0.1 million for the six months ended June 30, 2011. The most significant impact is the \$0.7 million increase in the change in the valuation of the warrant liability, mostly due to the decrease in our common share price.

The change in valuation of warrants relates to the issue of warrants to purchase shares of our common stock under the registered direct financing completed in February 2007. The warrants issued to the investors meet the requirements of and are being accounted for as a liability in accordance with ASC 480, "Accounting for Derivative Financial Instruments Indexed to, and Potentially Settled in, a Company's Own Stock." The fair value of the warrants is re-measured each reporting period with a derivative gain or loss recognized in the consolidated statement of operations. Such gains or losses will continue to be reported until the warrants are exercised or expired. For the six months ended June 30, 2010 and 2011, the change in the value of warrants was a \$0.5 million increase and a \$0.2 million decrease, respectively.

Foreign exchange gains (losses) increased by \$49,000 to a loss of \$87,000 for the six months ended June 30, 2011 compared to a loss of \$38,000 for the six months ended June 30, 2010. Foreign exchange gains/(losses) are reported in the consolidated statement of operations as a separate line item within other income (expense).

Interest income increased by approximately \$7,000 to \$24,000 for the six months ended June 30, 2010 from \$17,000 for the six months ended June 30, 2010. This is mostly attributed to a higher average daily balance of cash and cash equivalents during the three months ended June 30, 2011 compared to the same period in 2010.

Interest expense decreased by \$33,000 from \$33,000 for the six months ended June 30, 2010 to \$0 for the six months ended June 30, 2011. This reduction was primarily due to the elimination of the accretion expense associated with the restructured Bothell lease, which expired in December 2010.

The future

The valuation of the warrant liability will continue to be re-measured at the end of each reporting period. The valuation of the warrants is dependent upon many factors, including our stock price, interest rates and the remaining term of the instrument and may fluctuate significantly, which may have a significant impact on our statement of operations.

Income tax benefit

Credit is taken for research and development tax credits, which are claimed from the United Kingdom's revenue and customs authority, or HMRC, in respect of qualifying research and development costs incurred.

The following table summarizes research and development tax credits for the six months ended June 30, 2010 and 2011:

		Six Mont	ths End	ed				
		June 30,				Increase (Decrease)		
	2	2010	2	2011	\$		%	
			(\$	000s)				
Income tax benefit	\$	363	\$	317	\$	(46)	(13)%	

Research and development tax credits recoverable decreased \$46,000, from \$0.4 million for the six months ended June 30, 2010, to \$0.3 million for the six months ended June 30, 2011. The level of tax credits recoverable is linked directly to qualifying research and development expenditure incurred in any one year, but restricted to payroll taxes paid by us in the United Kingdom in that same year. The \$46,000 decrease between the six months ended June 30, 2010 and 2011 is due to lower payroll taxes paid as a result of lower headcount in 2011.

Liquidity and Capital Resources

The following is a summary of our key liquidity measures at December 31, 2010 and June 30, 2011:

	ember 31, 2010	ine 30, 2011 (\$00	 ifference	% Difference
Cash and cash equivalents	\$ 29,495	\$ 20,614	\$ (8,881)	(30)%
Working capital:	 _	_	_	
Current assets	\$ 31,051	\$ 22,449	\$ (8,602)	(28)%
Current liabilities	 (6,535)	 (6,040)	 495	8%
Working capital	\$ 24,516	\$ 16,409	\$ (8,107)	(33)%

The objectives of our cash management policy are to safeguard and preserve funds, to maintain liquidity sufficient to meet our cash flow requirements and to attain a market rate of return. At June 30, 2011, we had cash and cash equivalents of \$20.6 million as compared to \$29.5 million at December 31, 2010. The decrease in balance at June 30, 2010 was primarily due to normal cash outflows required to operate our business. Since our inception, we have not generated any significant revenue and have relied primarily on the proceeds from sales of equity and preferred securities to finance our operations and internal growth. Additional funding has come through interest on investments, licensing revenue, government grants and research and development tax credits. We have incurred significant losses since our inception. As of June 30, 2011, we had an accumulated deficit during the development stage of \$250.0 million. We believe that existing funds together with cash generated from operations and recent financing activities are sufficient to satisfy our planned working capital, capital expenditures, debt service and other financial commitments for at least the next twelve months. Current business and capital market risks could have a detrimental effect on the availability of sources of funding and our ability to access them in the future which may delay or impede our progress of advancing our drugs currently in the clinic to approval by the FDA for commercialization.

Cash provided by (used in) operating, investing and financing activities

Cash provided by (used in) operating, investing and financing activities for the six months ended June 30, 2010 and 2011, is summarized as follows:

	Six months ended June 30,		
	2010	2011	
	(\$000s	<u>, </u>	
Net cash used in operating activities	(9,776)	(8,477)	
Net cash provided by investing activities	27	_	
Net cash provided by (used in) financing activities	17,855	(441)	

Operating activities

Net cash used in operating activities decreased \$1.3 million, from \$9.8 million for the six months ended June 30, 2010 to \$8.5 million for the six months ended June 30, 2011. The decrease in cash outflow is primarily the result of better management of selling, general and administrative expenses as the result of our continued focus on sapacitabine.

Investing activities

Net cash provided by investing activities for the six months ended June 30, 2010 was \$27,000 as a result of the sale of laboratory equipment. There were no cash flows from investing activities during the six months ended June 30, 2011.

Financing activities

Net cash used in financing activities for the six months ended June 30, 2011 was \$0.4 million. Net cash provided by financing activities was \$17.9 million for the six months ended June 30, 2010. During the six months ended June 30, 2010, we completed two registered direct offerings in January 2010 for gross proceeds of approximately \$13.0 million, drew down the Kingsbridge CEFF totaling approximately \$2.8 million and had investors exercising warrants totaling \$2.7 million. During the six months ended June 30, 2011, we paid a cash dividend to our holders of the Preferred Stock of approximately \$0.4 million.

Operating Capital and Capital Expenditure Requirements

We expect to continue to incur substantial operating losses in the future. While we have generated modest product revenues from ALIGN product sales from October 2007 through June 30, 2011, we cannot guarantee that we will generate any significant product revenues until a product candidate has been approved by the FDA or similar regulatory agencies in other countries and successfully commercialized.

We currently anticipate that our cash and cash equivalents, including net proceeds of \$9.3 million from the underwritten offering closed in July 2011, will be sufficient to fund our operations for at least the next 12 months. We cannot be certain that any of our programs will be successful or that we will be able to raise sufficient funds to complete the development and commercialize any of our product candidates currently in development, should they succeed. Additionally, we plan to continue to evaluate in-licensing and acquisition opportunities to gain access to new drugs or drug targets that would fit with our strategy. Any such transaction would likely increase our funding needs in the future.

Our future funding requirements will depend on many factors, including but not limited to:

- the rate of progress and cost of our clinical trials, preclinical studies and other discovery and research and development activities;
- the costs associated with establishing manufacturing and commercialization capabilities;
- the costs of acquiring or investing in businesses, product candidates and technologies;
- · the costs of filing, prosecuting, defending and enforcing any patent claims and other intellectual property rights;
- the costs and timing of seeking and obtaining FDA and other regulatory approvals;
- the effect of competing technological and market developments; and
- the economic and other terms and timing of any collaboration, licensing or other arrangements into which we may enter.

Until we can generate a sufficient amount of product revenue to finance our cash requirements, which we may never do, we expect to finance future cash needs primarily through public or private equity offerings, debt financings or strategic collaborations. Although we are not reliant on institutional credit finance and therefore not subject to debt covenant compliance requirements or potential withdrawal of credit by banks, the current economic climate has also impacted the availability of funds and activity in equity markets. We do not know whether additional funding will be available on acceptable terms, or at all. If we are not able to secure additional funding when needed, we may have to delay, reduce the scope of or eliminate one or more of our clinical trials or research and development programs or make changes to our operating plan. In addition, we may have to partner one or more of our product candidate programs at an earlier stage of development, which would lower the economic value of those programs to us.

Off-Balance Sheet Arrangements

As of June 30, 2011, we have no off-balance sheet arrangements.

Critical Accounting Policies

Our discussion and analysis of our financial condition and results of operations is based on our financial statements, which have been prepared in accordance with accounting principles generally accepted in the United States. The preparation of these financial statements requires us to make estimates and judgments that affect the reported amounts of assets, liabilities and expenses and related disclosure of contingent assets and liabilities. We review our estimates on an ongoing basis. We base our estimates on historical experience and on various other assumptions that we believe to be reasonable under the circumstances. Actual results may differ from these estimates under different assumptions or conditions. We believe the judgments and estimates required by the following accounting policies to be critical in the preparation of our consolidated financial statements.

Revenue Recognition

Product sales

We have adopted the following revenue recognition policy related to the sales of Xclair[®] Cream, Numoisyn[®] Liquid and Numoisyn[®] Lozenges. We recognize revenue from these product sales when persuasive evidence of an arrangement exists; delivery has occurred or services have been rendered; the selling price is fixed and determinable; and collectability is reasonably assured.

As we offer a general right of return on these product sales, we must consider the guidance in ASC Topic 605. Under these guidelines, we account for all product sales using the "sell-through" method. Under the sell-through method, revenue is not recognized upon shipment of product to distributors. Instead, we record deferred revenue at gross invoice sales price and deferred cost of sales at the cost at which those goods were held in inventory. We recognize revenue when such inventory is sold through to pharmacies. To estimate products sold through to pharmacies, we rely on third-party information, including information obtained from significant distributors with respect to their inventory levels and sell-through to pharmacies. We estimate product returns based on historical returns experience and other factors that affect the demand for our products as well as the amount of product subject to return.

Stock-based Compensation

The Company grants stock options, restricted stock units and restricted stock to officers, employees, directors and consultants under the Company's 2006 Amended and Restated Equity Incentive Plan, which was amended and restated as of April 14, 2008. We also have outstanding options under various stock-based compensation plans for employees and directors.

ASC 718 requires measurement of compensation cost for all stock-based awards at fair value on date of grant and recognition of compensation over the requisite service period for awards expected to vest. The fair value of restricted stock and restricted stock units is determined based on the number of shares granted and the quoted price of our common stock on the date of grant. The determination of grant-date fair value for stock option awards is estimated using an option-pricing model, which includes variables such as the expected volatility of our share price, the anticipated exercise behavior of our employees, interest rates, and dividend yields. These variables are projected based on our historical data, experience, and other factors.

Such value is recognized as an expense over the requisite service period, net of estimated forfeitures, using the straight-line attribution method. The estimation of stock awards that will ultimately vest requires judgment, and to the extent actual results or updated estimates differ from our current estimates, such amounts will be recorded as a cumulative adjustment in the period estimates are revised. We consider many factors when estimating expected forfeitures, including types of awards, employee class, and historical experience. If our actual forfeiture rate is materially different from our estimate, the stock-based compensation expense could be significantly different from what we have recorded in the current period.

Warrants Liability

With respect to warrants issued in February 2007 as part of a financing and pursuant to ASC 815, since we are unable to control all the events or actions necessary to settle the warrants in registered shares the warrants have been recorded as a current liability at fair value. The fair value of the outstanding warrants is evaluated at each reporting period with any resulting change in the fair value being reflected in the consolidated statements of operations. During the three months ended June 30, 2010 and 2011, the Company recognized gains from the change in the value of warrants of approximately \$0.3 million and \$0.1 million, respectively. During the six months ended June 30, 2010, the Company recognized the change in the value of warrants as a loss of approximately \$0.5 million. During the six months ended June 30, 2011, the Company recognized the change in the value of warrants as a gain of approximately \$0.2 million. Fair value is estimated using an option-pricing model, which includes variables such as the expected volatility of our share price, interest rates, and dividend yields. These variables are projected based on our historical data, experience, and other factors. Changes in any of these variables could result in material adjustments to the expense recognized for changes in the valuation of the warrants liability.

Item 3. Quantitative and Qualitative Disclosures About Market Risk.

We are exposed to market risk related to fluctuations in foreign currency exchange rates.

Foreign Currency Risk

We are exposed to foreign currency rate fluctuations related to the operation of our subsidiary in the United Kingdom. At the end of each reporting period, income and expenses of the subsidiary are re-measured into U.S. dollars using the average currency rate in effect for the period and assets and liabilities are re-measured into U.S. dollars using either historical rates or the exchange rate in effect at the end of the period. Intercompany loans with this subsidiary are denominated in U.S. dollars and unrealized foreign exchange gains and losses arising on these loans have been recorded in the consolidated statement of operations within the separate line item foreign exchange gains/(losses) within other income (expense) up to September 30, 2008.

We currently do not engage in foreign currency hedging. We enter into certain transactions denominated in foreign currencies in respect of underlying operations and, therefore, we are subject to currency exchange risks. We realized losses of \$49,000 and gains of \$2,000 for the three months ended June 30, 2010 and 2011, respectively. During the six months ended June 30, 2010 and 2011, we realized losses of \$38,000 and \$66,000, respectively.

Common Stock Price Risk

In February 2007, we issued common stock and warrants. Pursuant to ASC 815, we recorded the fair value of the warrants as a current liability. The fair value of the outstanding warrants is evaluated at each reporting period with any resulting change in the fair value being reflected in the condensed consolidated statements of operations. We recognized a loss of \$0.3 million and a gain of \$0.1 million on the consolidated statement of operations for the three months ended June 30, 2010 and 2011, respectively, related to the change in the value of warrants. We recognized a loss of \$0.5 million and a gain of \$0.2 million on the consolidated statement of operations for the six months ended June 30, 2010 and 2011, respectively, related to the change in the value of warrants. Fair value of the derivative instruments will be affected by estimates of various factors that may affect the respective instrument, including our stock price, the risk free rate of return and expected volatility in the fair value of our stock price. As the fair value of this derivative may fluctuate significantly from period to period, the resulting change in valuation may have a significant impact on our results of operations.

Item 4. Controls and Procedures

Evaluation of Disclosure Controls and Procedures

Under the supervision and with the participation of our management, including our principal executive officer and principal financial and accounting officer, we conducted an evaluation of the effectiveness, as of June 30, 2011, of our disclosure controls and procedures, as defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934, as amended, or the Exchange Act. The purpose of this evaluation was to determine whether as of the evaluation date our disclosure controls and procedures were effective to provide reasonable assurance that the information we are required to disclose in our filings with the SEC under the Exchange Act (i) is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms and (ii) accumulated and communicated to our management, including our principal executive officer and principal financial and accounting officer, as appropriate to allow timely decisions regarding required disclosure. Based on that evaluation, management has concluded that as of June 30, 2011, our disclosure controls and procedures were not effective at the reasonable assurance level due to the material weaknesses in our internal controls described in the amendment to our Annual Report on Form 10-K for the year ended December 31, 2010 in the section captioned "Item 9A — Controls and Procedures — Management's Annual Report on Internal Control Over Financial Reporting" that remain present.

Changes in Internal Control Over Financial Reporting

There have been no changes in our internal control over financial reporting during the quarter ended June 30, 2011 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting other than those described below.

In response to the material weaknesses in our internal controls noted in our Annual Report on Form 10-K for the year ended December 31, 2010, filed on March 31, 2011, management presented a proposed remediation plan to our audit committee concerning our internal controls over financial reporting, and the audit committee adopted management's remediation plan. We have implemented the plan by hiring qualified finance professionals and retaining outside consultants. Remediation of the material weaknesses will require management time and attention over the coming quarters and will likely result in additional incremental expenses. Any failure on our part to remedy our identified weaknesses or any additional errors or delays in our financial reporting would have a material adverse effect on our business and results of operations and could have a substantial adverse impact on the trading price of our common stock.

Subject to oversight by our board of directors, our chief executive officer will be responsible for implementing management's internal control remediation plan, adopted by our audit committee and approved by our board of directors.

Specifically, the remediation plan consists of strengthening the financial reporting function through the hiring of qualified finance personnel, with experience in the interpretation and application of accounting principles generally accepted in the United States, or GAAP, and designing and placing into operation appropriate controls to prevent or detect on a timely basis any potential material misstatements in the accounting, presentation and disclosure of cumulative preferred dividends. It is anticipated that the remediation plan, once implemented, will materially affect our internal control over financial reporting.

We anticipate that the actions described above will remediate the material weakness described in our Annual Report on Form 10-K for the year ended December 31, 2010.

Inherent Limitation on the Effectiveness of Internal Controls

The effectiveness of any system of internal control over financial reporting, including ours, is subject to inherent limitations, including the exercise of judgment in designing, implementing, operating, and evaluating the controls and procedures, and the inability to eliminate misconduct completely. Accordingly, any system of internal control over financial reporting, including ours, no matter how well designed and operated, can only provide reasonable, not absolute assurances. In addition, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate. We intend to continue to monitor and upgrade our internal controls as necessary or appropriate for our business, but cannot assure you that such improvements will be sufficient to provide us with effective internal control over financial reporting.

PART II. OTHER INFORMATION

Item 1. Legal proceedings.

From time to time, we may be involved in routine litigation incidental to the conduct of our business. On April 27, 2010, we were served with a complaint filed by Celgene Corporation in the United States District Court for the District of Delaware seeking a declaratory judgment that four of our own patents, claiming the use of romidepsin injection in T-cell lymphomas, are invalid and not infringed by Celgene's products, but directly involve the use and administration of Celgene's ISTODAX® (romidepsin for injection) product. On June 17, 2010, we filed our answer and counterclaims to the declaratory judgment complaint. We have filed counterclaims charging Celgene with infringement of each of our four patents and seeking damages for Celgene's infringement as well as injunctive relief. The four patents directly involve the use and administration of Celgene's ISTODAX® (romidepsin for injection) product.

Item 1A. Risk Factors.

In analyzing our company, you should consider carefully the following risk factors, together with all of the other information included in Part I, "Item 1A. Risk Factors" in our Annual Report on Form 10-K for the year ended December 31, 2010. Factors that could cause or contribute to differences in our actual results include those discussed in the following subsection, as well as those discussed above in "Management's Discussion and Analysis of Financial Condition and Results of Operations" and elsewhere throughout this Quarterly Report on Form 10-Q. Each of the following risk factors, either alone or taken together, could adversely affect our business, operating results and financial condition, as well as adversely affect the value of an investment in our common stock.

Risks Associated with Development and Commercialization of Our Drug Candidates

Clinical trial designs that were discussed with the authorities prior to their commencement may subsequently be considered insufficient for approval at the time of application for regulatory approval. Thus, our SPA regarding our SEAMLESS trial does not guarantee marketing approval or approval of our sapacitabine oral capsules for the treatment of acute myeloid leukemia.

On September 13, 2010, we reached agreement with the FDA regarding an SPA on the design of a pivotal Phase 3 trial for our sapacitabine oral capsules as a front-line treatment in elderly patients aged 70 years or older with newly diagnosed acute myeloid leukemia, or AML, who are not candidates for intensive induction chemotherapy, or the SEAMLESS trial. An SPA provides trial sponsors with an agreement from the FDA that the design and analysis of the trial adequately address objectives in support of a submission for a marketing application if the trial is performed according to the SPA. The SPA may only be changed through a written agreement between the sponsor and the FDA or if the FDA becomes aware of a substantial scientific issue essential to product efficacy or safety. On January 11, 2011, we opened enrollment of the SEAMLESS trial.

An SPA, however, neither guarantees approval nor provides any assurance that a marketing application would be approved by the FDA. There are companies that have been granted SPAs but have ultimately failed to obtain final approval to market their drugs. The FDA may revise previous guidance or decide to ignore previous guidance at any time during the course of clinical activities or after the completion of clinical trials. The FDA may raise issues relating to, among other things, safety, study conduct, bias, deviation from the protocol, statistical power, patient completion rates, changes in scientific or medical parameters or internal inconsistencies in the data prior to making its final decision. The FDA may also seek the guidance of an outside advisory committee prior to making its final decision. Even with successful clinical safety and efficacy data, including such data from a clinical trial conducted pursuant to an SPA, we may be required to conduct additional, expensive clinical trials to obtain regulatory approval.

The development program for our lead drug candidate sapacitabine is based, in part, on intellectual property rights we license from others and any termination of this license could seriously harm our business.

Effective July 11, 2011, the Daiichi Sankyo license under which we license certain patent rights for sapacitabine, our lead drug candidate, was amended to irrevocably waive a termination right Daiichi Sankyo possessed under a provision of the agreement that required the Company to obtain regulatory approval to sell sapacitabine in at least one country by September 2011, and releases the Company from all claims and liability of any kind arising under such provision. The amendment further provides that the royalty due from the Company to Daiichi Sankyo on future net sales of sapacitabine be increased by a percentage between 1.25% and 1.50% depending on the level of net sales of sapacitabine realized.

In general, the license may be terminated by us for technical, scientific, efficacy, safety, or commercial reasons on six months notice, or twelve months if after a launch of a sapacitabine-based product, or by either party for material default.

Although we are currently in compliance with all of our material obligations under this license, if we were to breach any such obligations, our counterparty may be entitled to terminate the license. This would restrict or delay or eliminate our ability to develop and commercialize these drug candidates, which could adversely affect our business.

If we fail to enter into and maintain successful strategic alliances for our drug candidates, we may have to reduce or delay our drug candidate development or increase our expenditures.

An important element of our strategy for developing, manufacturing and commercializing our drug candidates is entering into strategic alliances with pharmaceutical companies or other industry participants to advance our programs and enable us to maintain our financial and operational capacity.

We face significant competition in seeking appropriate alliances. We may not be able to negotiate alliances on acceptable terms, if at all. In addition, these alliances may be unsuccessful. If we fail to create and maintain suitable alliances, we may have to limit the size or scope of, or delay, one or more of our drug development or research programs. If we elect to fund drug development or research programs on our own, we will have to increase our expenditures and will need to obtain additional funding, which may be unavailable only on unfavorable terms.

Clinical trials are expensive, time consuming, subject to delay and may be required to continue beyond our available funding.

Clinical trials are expensive, complex, can take many years to conduct and may have uncertain outcomes. We estimate that clinical trials of our most advanced drug candidates may be required to continue beyond our available funding and may take several years more to complete. The designs used in some of our trials have not been used widely by other pharmaceutical companies. Failure can occur at any stage of the testing and we may experience numerous unforeseen events during, or as a result of, the clinical trial process that could delay or prevent commercialization of our current or future drug candidates, including but not limited to:

- delays in securing clinical investigators or trial sites for our clinical trials;
- delays in obtaining institutional review board, or IRB, and other regulatory approvals to commence a clinical trial:
- slower than anticipated rates of patient recruitment and enrollment, or reaching the targeted number of patients because of competition for patients from other trials or other reasons;
- negative or inconclusive results from clinical trials;
- unforeseen safety issues;
- uncertain dosing issues may or may not be related to suboptimal pharmacokinetic and pharmacodynamic behaviors;
- approval and introduction of new therapies or changes in standards of practice or regulatory guidance that render our clinical trial endpoints or the targeting of our proposed indications obsolete;
- inability to monitor patients adequately during or after treatment or problems with investigator or patient compliance with the trial protocols;
- inability to replicate in large controlled studies safety and efficacy data obtained from a limited number of patients in uncontrolled trials;
- · inability or unwillingness of medical investigators to follow our clinical protocols; and
- unavailability of clinical trial supplies.

If we suffer any significant delays, setbacks or negative results in, or termination of, our clinical trials, we may be unable to continue development of our drug candidates or generate revenue and our development costs could increase significantly. Adverse events have been observed in our clinical trials and may force us to stop development of our product candidates or prevent regulatory approval of our product candidates.

Adverse or inconclusive results from our clinical trials may substantially delay, or halt entirely, any further development of our drug candidates. Many companies have failed to demonstrate the safety or effectiveness of drug candidates in later stage clinical trials notwithstanding favorable results in early stage clinical trials. Previously unforeseen and unacceptable side effects could interrupt, delay or halt clinical trials of our drug candidates and could result in the FDA or other regulatory authorities denying approval of our drug candidates. We will need to demonstrate safety and efficacy for specific indications of use, and monitor safety and compliance with clinical trial protocols throughout the development process. To date, long-term safety and efficacy has not been demonstrated in clinical trials for any of our drug candidates. Toxicity and "serious adverse events" as defined in trial protocols have been noted in preclinical and clinical trials involving certain of our drug candidates. For example, neutropenia and gastro-intestinal toxicity were observed in patients receiving sapacitabine and elevations of liver enzymes and decrease in potassium levels have been observed in patients receiving seliciclib.

In addition, we may pursue clinical trials for sapacitabine and seliciclib in more than one indication. There is a risk that severe toxicity observed in a trial for one indication could result in the delay or suspension of all trials involving the same drug candidate. Even if we believe the data collected from clinical trials of our drug candidates are promising with respect to safety and efficacy, such data may not be deemed sufficient by regulatory authorities to warrant product approval. Clinical data can be interpreted in different ways. Regulatory officials could interpret such data in different ways than we do which could delay, limit or prevent regulatory approval. The FDA, other regulatory authorities or we may suspend or terminate clinical trials at any time. Any failure or significant delay in completing clinical trials for our drug candidates, or in receiving regulatory approval for the commercialization of our drug candidates, may severely harm our business and reputation.

If our understanding of the role played by CDKs or AKs in regulating the cell cycle is incorrect, this may hinder pursuit of our clinical and regulatory strategy.

Our development of small molecule inhibitors of CDK and AK is based on our understanding of the mechanisms of action of CDK and AK inhibitors and their interaction with other cellular mechanisms. One of our drug candidates, seliciclib, is a CDK inhibitor, and CYC116 is an AK and VEGFR2 inhibitor. Although a number of pharmaceutical and biotechnology companies are attempting to develop CDK or AK inhibitor drugs for the treatment of cancer, no CDK or AK inhibitor has yet reached the market. If our understanding of the role played by CDK or AK inhibitors in regulating the cell cycle is incorrect, seliciclib and/or CYC116 may fail to produce therapeutically relevant results hindering our ability to pursue our clinical and regulatory strategy.

We are making use of biomarkers, which are not scientifically validated, and our reliance on biomarker data may thus lead us to direct our resources inefficiently.

We are making use of biomarkers in an effort to facilitate our drug development and to optimize our clinical trials. Biomarkers are proteins or other substances whose presence in the blood can serve as an indicator of specific cell processes. We believe that these biological markers serve a useful purpose in helping us to evaluate whether our drug candidates are having their intended effects through their assumed mechanisms, and thus enable us to identify more promising drug candidates at an early stage and to direct our resources efficiently. We also believe that biomarkers may eventually allow us to improve patient selection in connection with clinical trials and monitor patient compliance with trial protocols.

For most purposes, however, biomarkers have not been scientifically validated. If our understanding and use of biomarkers is inaccurate or flawed, or if our reliance on them is otherwise misplaced, then we will not only fail to realize any benefits from using biomarkers, but may also be led to invest time and financial resources inefficiently in attempting to develop inappropriate drug candidates. Moreover, although the FDA has issued for comment a draft guidance document on the potential use of biomarker data in clinical development, such data are not currently accepted by the FDA or other regulatory agencies in the United States, the European Union or elsewhere in applications for regulatory approval of drug candidates and there is no guarantee that such data will ever be accepted by the relevant authorities in this connection. Our biomarker data should not be interpreted as evidence of efficacy.

Due to our reliance on contract research organizations or other third parties to conduct clinical trials, we may be unable to directly control the timing, conduct and expense of our clinical trials.

We do not have the ability to independently conduct clinical trials required to obtain regulatory approvals for our drug candidates. We must rely on third parties, such as contract research organizations, data management companies, contract clinical research associates, medical institutions, clinical investigators and contract laboratories to conduct our clinical trials. In addition, we rely on third parties to assist with our preclinical development of drug candidates. If these third parties do not successfully carry out their contractual duties or regulatory obligations or meet expected deadlines, if the third parties need to be replaced or if the quality or accuracy of the data they obtain is compromised due to the failure to adhere to our clinical protocols or regulatory requirements or for other reasons, our preclinical development activities or clinical trials may be extended, delayed, suspended or terminated, and we may not be able to obtain regulatory approval for or successfully commercialize our drug candidates.

To the extent we are able to enter into collaborative arrangements or strategic alliances, we will be exposed to risks related to those collaborations and alliances.

Although we are not currently party to any collaboration arrangement or strategic alliance that is material to our business, in the future we expect to be dependent upon collaborative arrangements or strategic alliances to complete the development and commercialization of some of our drug candidates particularly after the Phase 2 stage of clinical testing. These arrangements may place the development of our drug candidates outside our control, may require us to relinquish important rights or may otherwise be on terms unfavorable to us.

We may be unable to locate and enter into favorable agreements with third parties, which could delay or impair our ability to develop and commercialize our drug candidates and could increase our costs of development and commercialization. Dependence on collaborative arrangements or strategic alliances will subject us to a number of risks, including the risk that:

- we may not be able to control the amount and timing of resources that our collaborators may devote to the drug candidates;
- our collaborators may experience financial difficulties;
- we may be required to relinquish important rights such as marketing and distribution rights;
- business combinations or significant changes in a collaborator's business strategy may also adversely affect a collaborator's willingness or ability to complete our obligations under any arrangement;
- a collaborator could independently move forward with a competing drug candidate developed either independently or in collaboration with others, including our competitors; and
- collaborative arrangements are often terminated or allowed to expire, which would delay the development and
 may increase the cost of developing our drug candidates.

We have no manufacturing capacity and will rely on third party manufacturers for the late stage development and commercialization of any drugs or devices we may develop or sell.

We do not currently operate manufacturing facilities for clinical or commercial production of our drug candidates under development or our currently marketed ALIGN products. We currently lack the resources or the capacity to manufacture any of our products on a clinical or commercial scale. We depend upon a third party, Sinclair, to manufacture the commercial products sold by our ALIGN subsidiary and we cannot rely upon Sinclair to continue to supply the products. We anticipate future reliance on a limited number of third party manufacturers until we are able, or decide to, expand our operations to include manufacturing capacities. Any performance failure on the part of manufacturers could delay late stage clinical development or regulatory approval of our drug, the commercialization of our drugs or our ability to sell our commercial products, producing additional losses and depriving us of potential product revenues.

If the FDA or other regulatory agencies approve any of our drug candidates for commercial sale, or if we significantly expand our clinical trials, we will need to manufacture them in larger quantities and will be required to secure alternative third-party suppliers to our current suppliers. To date, our drug candidates have been manufactured in small quantities for preclinical testing and clinical trials and we may not be able to successfully increase the manufacturing capacity, whether in collaboration with our current or future third-party manufacturers or on our own, for any of our drug candidates in a timely or economic manner, or at all. Significant scale-up of manufacturing may require additional validation studies, which the FDA and other regulatory bodies must review and approve. If we are unable to successfully increase the manufacturing capacity for a drug candidate whether for late stage clinical trials or for commercial sale or are unable to secure alternative third-party suppliers to our current suppliers, the drug development, regulatory approval or commercial launch of any related drugs may be delayed or blocked or there may be a shortage in supply. Even if any third party manufacturer makes improvements in the manufacturing process for our drug candidates, we may not own, or may have to share, the intellectual property rights to such innovation.

As we evolve from a company primarily involved in discovery and development to one also involved in the commercialization of drugs and devices, we may encounter difficulties in managing our growth and expanding our operations successfully.

In order to execute our business strategy, we will need to expand our development, control and regulatory capabilities and develop financial, manufacturing, marketing and sales capabilities or contract with third parties to provide these capabilities for us. If our operations expand, we expect that we will need to manage additional relationships with various collaborative partners, suppliers and other third parties. Our ability to manage our operations and any growth will require us to make appropriate changes and upgrades, as necessary, to our operational, financial and management controls, reporting systems and procedures wherever we may operate. Any inability to manage growth could delay the execution of our business plan or disrupt our operations.

The failure to attract and retain skilled personnel and key relationships could impair our drug development and commercialization efforts.

We are highly dependent on our senior management and key scientific, technical and sales and marketing personnel. Competition for these types of personnel is intense. The loss of the services of any member of our senior management, scientific, technical or sales or marketing staff may significantly delay or prevent the achievement of drug development and other business objectives and could have a material adverse effect on our business, operating results and financial condition. We also rely on consultants and advisors to assist us in formulating our strategy. All of our consultants and advisors are either self-employed or employed by other organizations, and they may have conflicts of interest or other commitments, such as consulting or advisory contracts with other organizations, that may affect their ability to contribute to us. The success of the commercialization of the ALIGN products depends, in large part, on our continued ability to develop and maintain important relationships with distributors and research and medical institutions. Failure to do that could have a material adverse effect on our ability to commercialize the ALIGN products.

We intend to expand and develop new drug candidates. We will need to hire additional employees in order to continue our clinical trials and market our drug candidates and medical devices. This strategy will require us to recruit additional executive management and scientific and technical personnel. There is currently intense competition for skilled executives and employees with relevant scientific and technical expertise, and this competition is likely to continue. The inability to attract and retain sufficient scientific, technical and managerial personnel could limit or delay our product development efforts, which would adversely affect the development of our drug candidates and commercialization of our potential drugs and growth of our business.

Our drug candidates are subject to extensive regulation, which can be costly and time-consuming, and we may not obtain approvals for the commercialization of any of our drug candidates.

The clinical development, manufacturing, selling and marketing of our drug candidates are subject to extensive regulation by the FDA and other regulatory authorities in the United States, the European Union and elsewhere. These regulations also vary in important, meaningful ways from country to country. We are not permitted to market a potential drug in the United States until we receive approval of an NDA from the FDA. We have not received an NDA approval from the FDA for any of our drug candidates.

Obtaining an NDA approval is expensive and is a complex, lengthy and uncertain process. The FDA approval process for a new drug involves completion of preclinical studies and the submission of the results of these studies to the FDA, together with proposed clinical protocols, manufacturing information, analytical data and other information in an Investigational New Drug, or IND, which must become effective before human clinical trials may begin. Clinical development typically involves three phases of study: Phase 1, 2 and 3. The most significant costs associated with clinical development are the pivotal or suitable for registration late Phase 2 or Phase 3 clinical trials as they tend to be the longest and largest studies conducted during the drug development process. After completion of clinical trials, an NDA may be submitted to the FDA.

In responding to an NDA, the FDA may refuse to file the application, or if accepted for filing, the FDA may grant marketing approval, request additional information or deny the application if it determines that the application does not provide an adequate basis for approval. In addition, failure to comply with the FDA and other applicable foreign and U.S. regulatory requirements may subject us to administrative or judicially imposed sanctions. These include warning letters, civil and criminal penalties, injunctions, product seizure or detention, product recalls, total or partial suspension of production and refusal to approve either pending NDAs, or supplements to approved NDAs.

Despite the substantial time and expense invested in preparation and submission of an NDA or equivalents in other jurisdictions, regulatory approval is never guaranteed. The FDA and other regulatory authorities in the United States, the European Union and elsewhere exercise substantial discretion in the drug approval process. The number, size and design of preclinical studies and clinical trials that will be required for FDA or other regulatory approval will vary depending on the drug candidate, the disease or condition for which the drug candidate is intended to be used and the regulations and guidance documents applicable to any particular drug candidate. The FDA or other regulators can delay, limit or deny approval of a drug candidate for many reasons, including, but not limited to:

- those discussed in the risk factor which immediately follows;
- the fact that the FDA or other regulatory officials may not approve our or our third party manufacturer's processes or facilities; or
- the fact that new regulations may be enacted by the FDA or other regulators may change their approval policies
 or adoption of new regulations requiring new or different evidence of safety and efficacy for the intended use of
 a drug candidate.

With regard to the ALIGN products, and following regulatory approval of any of our drug candidates, we are subject to ongoing regulatory obligations and restrictions, which may result in significant expense and limit our ability to commercialize our potential products.

With regard to our ALIGN products and our drug candidates, if any, approved by the FDA or by another regulatory authority, we are held to extensive regulatory requirements over product manufacturing, labeling, packaging, adverse event reporting, storage, advertising, promotion and record keeping. Regulatory approvals may also be subject to significant limitations on the indicated uses or marketing of the drug candidates. Potentially costly follow-up or post-marketing clinical studies may be required as a condition of approval to further substantiate safety or efficacy, or to investigate specific issues of interest to the regulatory authority. Previously unknown problems with the product or drug candidate, including adverse events of unanticipated severity or frequency, may result in restrictions on the marketing of the drug or device, and could include withdrawal of the drug or device from the market.

In addition, the law or regulatory policies governing pharmaceuticals may change. New statutory requirements may be enacted or additional regulations may be enacted that could prevent or delay regulatory approval of our drug candidates. We cannot predict the likelihood, nature or extent of adverse government regulation that may arise from future legislation or administrative action, either in the United States or elsewhere. If we are not able to maintain regulatory compliance, we might not be permitted to market our drugs and our business could suffer.

Our applications for regulatory approval could be delayed or denied due to problems with studies conducted before we in-licensed the rights to some of our product candidates.

We currently license some of the compounds and drug candidates used in our research programs from third parties. These include sapacitabine which was licensed from Daiichi Sankyo. Our present research involving these compounds relies upon previous research conducted by third parties over whom we had no control and before we in-licensed the drug candidates. In order to receive regulatory approval of a drug candidate, we must present all relevant data and information obtained during our research and development, including research conducted prior to our licensure of the drug candidate. Although we are not currently aware of any such problems, any problems that emerge with preclinical research and testing conducted prior to our in-licensing may affect future results or our ability to document prior research and to conduct clinical trials, which could delay, limit or prevent regulatory approval for our drug candidates.

We face intense competition and our competitors may develop drugs that are less expensive, safer, or more effective than our drug candidates.

A large number of drug candidates are in development for the treatment of leukemia, lung cancer, lymphomas and nasopharyngeal cancer. Several pharmaceutical and biotechnology companies have nucleoside analogs or other products on the market or in clinical trials which may be competitive to sapacitabine in both hematological and oncology indications. These include Celgene, Cephalon, Eisai, Johnson & Johnson, Eli Lilly, Genzyme, GlaxoSmithKline, Hospira, Onconova, Pfizer, Seattle Genetics and Sunesis. There are two other orally available CDK inhibitors in Phase 2 clinical trials. PD-0332991 (Pfizer/Onyx) and PHA-848125 (Nerviano Medical Sciences) target different subsets of kinase enzymes and have a different mechanism of action from seliciclib. We believe that seliciclib is currently the most advanced orally available CDK-specific agent in Phase 2 clinical trials but that there are a number of companies, including AstraZeneca, Astex Therapeutics, Bayer-Schering, Eisai, Merck, Nerviano Medical Sciences, Pfizer, Piramal Life Sciences, and Roche that are developing CDK inhibitors in early stage clinical trials in cancer patients. Although Aventis, a predecessor of Sanofi-Aventis, had previously announced that it has ceased Phase 2 development of alvocidib or flavopiridol, a CDK inhibitor, we believe that the National Cancer Institute's Cancer Therapy Evaluation Program, or CTEP, is continuing to enroll patients in a CTEP sponsored trial in patients with chronic leukemia. A number of companies are pursuing discovery and research activities in each of the other areas that are the subject of our research and drug development programs. We believe that Amgen, Astex Therapeutics, AstraZeneca, Entremed, Merck, jointly with Vertex, Nerviano Medical Sciences, Pfizer, Rigel, Sunesis and Takeda-Millennium have commenced Phase 1, Phase 2, or Phase 3 clinical trials of Aurora kinase inhibitors in patients with advanced cancers. Several companies have reported selection of Aurora kinase inhibitor candidates for development and may have started or are expected to start clinical trials within the next twelve months. We believe that Boehringer Ingelheim, GlaxoSmithKline, Merck, Nerviano Medical Sciences, Takeda-Millennium and Tekmira Pharmaceuticals Corporation have commenced Phase 1 or Phase 2 clinical trials with Plk inhibitor candidates for oncology indications. For our ALIGN products, we believe that Beiersdorf, Daiichi Sankyo, Eisai, Johnson & Johnson, MPM Medical and other companies market products for radiation dermatitis and xerostomia.

Our competitors, either alone or together with collaborators, may have substantially greater financial resources and research and development staff. Our competitors may also have more experience:

- developing drug candidates;
- conducting preclinical and clinical trials;
- obtaining regulatory approvals; and
- commercializing product candidates.

Our competitors may succeed in obtaining patent protection and regulatory approval and may market drugs before we do. If our competitors market drugs that are less expensive, safer, more effective or more convenient to administer than our potential drugs, or that reach the market sooner than our potential drugs, we may not achieve commercial success. Scientific, clinical or technical developments by our competitors may render our drug candidates obsolete or noncompetitive. We anticipate that we will face increased competition in the future as new companies enter the markets and as scientific developments progress. If our drug candidates obtain regulatory approvals, but do not compete effectively in the marketplace, our business will suffer.

The commercial success of the ALIGN products and our drug candidates depends upon their market acceptance among physicians, patients, healthcare providers and payors and the medical community.

It is necessary that our and our distribution partners' products, including Xclair® Cream, Numoisyn® Liquid and Numoisyn® Lozenges achieve and maintain market acceptance. If our drug candidates are approved by the FDA or by another regulatory authority, the resulting drugs, if any, may not gain market acceptance among physicians, healthcare providers and payors, patients and the medical community. The degree of market acceptance of any of our approved drugs or devices will depend on a variety of factors, including:

- timing of market introduction, number and clinical profile of competitive drugs;
- our ability to provide acceptable evidence of safety and efficacy;
- relative convenience and ease of administration;
- cost-effectiveness:
- availability of coverage, reimbursement and adequate payment from health maintenance organizations and other third party payors;
- · prevalence and severity of adverse side effects; and
- other potential advantages over alternative treatment methods.

If our drugs fail to achieve market acceptance, we may not be able to generate significant revenue and our business would suffer.

If we are unable to compete successfully in our market place, it will harm our business.

There are existing products in the marketplace that compete with our products. Companies may develop new products that compete with our products. Certain of these competitors and potential competitors have longer operating histories, substantially greater product development capabilities and financial, scientific, marketing and sales resources. Competitors and potential competitors may also develop products that are safer, more effective or have other potential advantages compared to our products. In addition, research, development and commercialization efforts by others could render our products obsolete or non-competitive. Certain of our competitors and potential competitors have broader product offerings and extensive customer bases allowing them to adopt aggressive pricing policies that would enable them to gain market share. Competitive pressures could result in price reductions, reduced margins and loss of market share. We could encounter potential customers that, due to existing relationships with our competitors, are committed to products offered by those competitors. As a result, those potential customers may not consider purchasing our products.

There is uncertainty related to coverage, reimbursement and payment by healthcare providers and payors for the ALIGN products and newly approved drugs, if any. The inability or failure to obtain or maintain coverage could affect our ability to market the ALIGN products and our future drugs and decrease our ability to generate revenue.

The availability and levels of coverage and reimbursement of newly approved drugs by healthcare providers and payors is subject to significant uncertainty. The commercial success of the ALIGN products and our drug candidates in both the United States and international markets is substantially dependent on whether third party coverage and reimbursement is available. The United States Centers for Medicare and Medicaid Services, health maintenance organizations and other third party payors in the United States, the European Union and other jurisdictions are increasingly attempting to contain healthcare costs by limiting both coverage and the level of reimbursement of new drugs and, as a result, they may not cover or provide adequate payment for our potential drugs. The ALIGN products and our drug candidates may not be considered cost-effective and reimbursement may not be available to consumers or may not be sufficient to allow the ALIGN products or our drug candidates to be marketed on a competitive basis.

In some countries, pricing of prescription drugs is subject to government control. In such countries, pricing negotiations with governmental authorities can take three to 12 months or longer following application to the competent authorities. To obtain reimbursement or pricing approval in such countries may require conducting an additional clinical trial comparing the cost-effectiveness of the drug to other alternatives. In the United States, the Medicare Part D drug benefit implemented in 2006 will limit drug coverage through formularies and other cost and utilization management programs, while Medicare Part B limits drug payments to a certain percentage of average price or through restrictive payment policies of "least costly alternatives" and "inherent reasonableness" Our business could be materially harmed if coverage, reimbursement or pricing is unavailable or set at unsatisfactory levels.

Intellectual property rights and distribution rights for our drug candidate seliciclib and ALIGN products are licensed from others, and any termination of these licenses could harm our business.

We have in-licensed certain patent rights in connection with the development program of our drug candidate seliciclib. Pursuant to the CNRS and Institut Curie license under which we license seliciclib, we are obligated to pay license fees, milestone payments and royalties and provide regular progress reports. We are also obligated to use reasonable efforts to develop and commercialize products based on the licensed patents.

We have in-licensed from Sinclair the distribution rights to the ALIGN products. This license agreement imposes obligations on us and expires in 2015. Although we are currently in compliance with all of our material obligations under this license, if we were to breach any such obligations, Sinclair would be permitted to terminate the license. In addition, if we are unable to extend the term of the license agreement, it would prevent us from distributing the ALIGN products.

Although we are currently in compliance with all of our material obligations under these licenses, if we were to breach any such obligations our counterparties may be entitled to terminate the licenses. This would restrict or delay or eliminate our ability to develop and commercialize the seliciclib or sell the ALIGN products, which could adversely affect our business.

We may be exposed to product liability claims that may damage our reputation and we may not be able to obtain adequate insurance.

Because we conduct clinical trials in humans, we face the risk that the use of our drug candidates will result in adverse effects. We believe that we have obtained reasonably adequate product liability insurance coverage for our trials. We cannot predict, however, the possible harm or side effects that may result from our clinical trials. Such claims may damage our reputation and we may not have sufficient resources to pay for any liabilities resulting from a claim excluded from, or beyond the limit of, our insurance coverage.

As we market commercialized products through our ALIGN subsidiary we are exposed to additional risks of product liability claims. These risks exist even with respect to drugs and devices that are approved for commercial sale by the FDA or other regulatory authorities in the United States, the European Union or elsewhere and manufactured in facilities licensed and regulated by the FDA or other such regulatory authorities. We have secured limited product liability insurance coverage, but may not be able to maintain such insurance on acceptable terms with adequate coverage, or at a reasonable cost. There is also a risk that third parties that we have agreed to indemnify could incur liability. Even if we were ultimately successful in product liability litigation, the litigation would consume substantial amounts of our financial and managerial resources and may exceed insurance coverage creating adverse publicity, all of which would impair our ability to generate sales of the litigated product as well as our other potential drugs.

We may be required to defend lawsuits or pay damages in connection with the alleged or actual violation of healthcare statutes such as fraud and abuse laws, and our corporate compliance programs can never guarantee that we are in compliance with all relevant laws and regulations.

Our commercialization efforts in the United States are subject to various federal and state laws pertaining to promotion and healthcare fraud and abuse, including federal and state anti-kickback, fraud and false claims laws. Anti-kickback laws make it illegal for a manufacturer to offer or pay any remuneration in exchange for, or to induce, the referral of business, including the purchase of a product. The federal government has published many regulations relating to the anti-kickback statutes, including numerous safe harbors or exemptions for certain arrangements. False claims laws prohibit anyone from knowingly and willingly presenting, or causing to be presented for payment to third-party payers including Medicare and Medicaid, claims for reimbursed products or services that are false or fraudulent, claims for items or services not provided as claimed, or claims for medically unnecessary items or services.

Our activities relating to the sale and marketing of our products will be subject to scrutiny under these laws and regulations. It may be difficult to determine whether or not our activities, comply with these complex legal requirements. Violations are punishable by significant criminal and/or civil fines and other penalties, as well as the possibility of exclusion of the product from coverage under governmental healthcare programs, including Medicare and Medicaid. If the government were to investigate or make allegations against us or any of our employees, or sanction or convict us or any of our employees, for violations of any of these legal requirements, this could have a material adverse effect on our business, including our stock price. Our activities could be subject to challenge for many reasons, including the broad scope and complexity of these laws and regulations, the difficulties in interpreting and applying these legal requirements, and the high degree of prosecutorial resources and attention being devoted to the biopharmaceutical industry and health care fraud by law enforcement authorities. During the last few years, numerous biopharmaceutical companies have paid multi-million dollar fines and entered into burdensome settlement agreements for alleged violation of these requirements, and other companies are under active investigation. Although we have developed and implemented corporate and field compliance programs as part of our commercialization efforts, we cannot assure you that we or our employees, directors or agents were, are or will be in compliance with all laws and regulations or that we will not come under investigation, allegation or sanction.

In addition, we may be required to prepare and report product pricing-related information to federal and state governmental authorities, such as the Department of Veterans Affairs and under the Medicaid program. The calculations used to generate the pricing-related information are complex and require the exercise of judgment. If we fail to accurately and timely report product pricing-related information or to comply with any of these or any other laws or regulations, various negative consequences could result, including criminal and/or civil prosecution, substantial criminal and/or civil penalties, exclusion of the approved product from coverage under governmental healthcare programs including Medicare and Medicaid, costly litigation and restatement of our financial statements. In addition, our efforts to comply with this wide range of laws and regulations are, and will continue to be, time-consuming and expensive.

If our supplier upon whom we rely fails to produce on a timely basis the finished goods in the volumes that we require or fails to meet quality standards and maintain necessary licensure from regulatory authorities, we may be unable to meet demand for our products, potentially resulting in lost revenues.

Our licensor and supplier Sinclair contracts with third party manufacturers to supply the finished goods to us to meet our needs. If any of Sinclair's third party manufacturers service providers do not meet our or our licensor's requirements for quality, quantity or timeliness, or do not achieve and maintain compliance with all applicable regulations, demand for our products or our ability to continue supplying such products could substantially decline. As the third party manufacturers are the sole supplier of the products any delays may impact our sales.

In all the countries where we sell or may sell our products, governmental regulations exist to define standards for manufacturing, packaging, labeling and storing. All of our suppliers of raw materials and contract manufacturers must comply with these regulations. Failure to do so could result in supply interruptions. In the United States, the FDA requires that all suppliers of pharmaceutical bulk material and all manufacturers of pharmaceuticals for sale in or from the United States achieve and maintain compliance with the FDA's Current Good Manufacturing Practice or cGMP regulations and guidelines. Failure of our third-party manufacturers to comply with applicable regulations could result in sanctions being imposed on them or us, including fines, injunctions, civil penalties, disgorgement, suspension or withdrawal of approvals, license revocation, seizures or recalls of products, operating restrictions and criminal prosecutions, any of which could significantly and adversely affect supplies of our products. In addition, before any product batch produced by our manufacturers can be shipped, it must conform to release specifications pre-approved by regulators for the content of the pharmaceutical product. If the operations of one or more of our manufacturers were to become unavailable for any reason, any required FDA review and approval of the operations of an alternative supplier could cause a delay in the manufacture of our products.

Our customer base is highly concentrated.

Our principal customers are a small number of wholesale drug distributors. These customers comprise a significant part of the distribution network for pharmaceutical products in the United States. Three large wholesale distributors, AmerisourceBergen Corporation, Cardinal Health, Inc. and McKesson Corporation, control a significant share of the market in the United States. Our ability to distribute any product, including Xclair® Cream, Numoisyn® Liquid and Numoisyn® Lozenges and to recognize revenues on a timely basis is substantially dependent on our ability to maintain commercially reasonable agreements with each of these wholesale distributors and the extent to which these distributors, over whom we have no control, comply with such agreements. Our agreements with wholesaler distributors may contain terms that are not favorable, given our relative lack of market leverage as a company with only three approved products or other factors, which could adversely affect our commercialization of Xclair® Cream, Numoisyn® Liquid and Numoisyn® Lozenges. The loss of any of these customers could materially and adversely affect our ability to distribute our products, resulting in a negative impact on our operations and financial condition.

We may be unable to accurately estimate demand and monitor wholesaler inventory of Xclair® Cream, Numoisyn® Liquid or Numoisyn® Lozenges. Although we attempt to monitor wholesaler inventory of Xclair® Cream, Numoisyn® Liquid or Numoisyn® Lozenges, we also rely on third party information, which is inherently uncertain and may not be accurate, to assist us in monitoring estimated inventory levels and prescription trends. Inaccurate estimates of the demand and inventory levels of the product may cause our revenues to fluctuate significantly from quarter to quarter and may cause our operating results for a particular quarter to be below expectations.

Inventory levels of Xclair® Cream, Numoisyn® Liquid or Numoisyn® Lozenges held by wholesalers can also cause our operating results to fluctuate unexpectedly. For the three months ended June 30, 2010 and 2011, approximately 91% and 90%, respectively, of our product sales in the United States were to three wholesalers, Cardinal Health, Inc., McKesson Corporation and AmerisourceBergen. Inventory levels held by those wholesalers can cause our operating results to fluctuate unexpectedly if our sales to wholesalers do not match customer demand. We have entered into inventory management agreements with these U.S. wholesalers under which they provide us with data regarding inventory levels at these wholesalers. However, these wholesalers may not be completely effective in matching inventory levels to customer demand, as they make estimates to determine customer demand. In addition, inventory is held at retail pharmacies and other non-wholesaler locations, for which we have no inventory management agreements and have no control in respect to their buying patterns. Also, the non-retail sector in the United States, which includes government institutions and large health maintenance organizations, tends to be less consistent in terms of buying patterns, and often causes quarter-over-quarter fluctuations in inventory and ordering patterns. We attempt to monitor inventory of Xclair®, Numoisyn® Liquid or Numoisyn® Lozenges in the United States through the use of internal sales forecasts and the expiration dates of product shipped, among other factors.

The commercialization of our products is substantially dependent on our ability to develop effective sales and marketing capabilities.

Our successful commercialization of Xclair® Cream, Numoisyn® Liquid and Numoisyn® Lozenges in the United States will depend on our ability to establish and maintain effective sales and marketing initiatives in the United States. Although we launched the ALIGN products with a small specialty oncology sales force, we now sell and market our products via unique sales and marketing strategies in order to reduce costs. We contracted, trained and deployed additional telemarketing personnel to call on specialists who prescribe ALIGN products. We also utilize mailings, print advertising, sampling, trade show attendance and other unique marketing programs to reach our customer base. We may increase or decrease the size of our telemarketing sales force in the future, depending on many factors, including the effectiveness of the sales force, the level of market acceptance of Xclair® Cream, Numoisyn® Liquid and Numoisyn® Lozenges and the results of our clinical trials. Prior to our launches of these products, we had never sold or marketed any products.

For our product candidates currently under development, our strategy is to develop compounds through the Phase 2 stage of clinical testing and market or co-promote certain of our drugs on our own. We have limited sales, marketing or distribution capabilities. We will depend primarily on strategic alliances with third parties, which have established distribution systems and sales forces, to commercialize our drugs. To the extent that we are unsuccessful in commercializing any drugs or devices ourselves or through a strategic alliance, product revenues will suffer, we will incur significant additional losses and our share price will be negatively affected.

Defending against claims relating to improper handling, storage or disposal of hazardous chemical, radioactive or biological materials could be time consuming and expensive.

Our research and development involves the controlled use of hazardous materials, including chemicals, radioactive and biological materials such as chemical solvents, phosphorus and bacteria. Our operations produce hazardous waste products. We cannot eliminate the risk of accidental contamination or discharge and any resultant injury from those materials. Various laws and regulations govern the use, manufacture, storage, handling and disposal of hazardous materials. We may be sued for any injury or contamination that results from our use or the use by third parties of these materials. Compliance with environmental laws and regulations may be expensive, and current or future environmental regulations may impair our research, development and production efforts.

Risks Related to Our Business and Financial Condition

The current economic conditions and financial market turmoil could adversely affect our business and results of operations.

Economic conditions remain difficult with the continuing uncertainty in the global credit markets, the financial services industry and the United States capital markets and with the United States economy as a whole experiencing a period of substantial turmoil and uncertainty characterized by unprecedented intervention by the United States federal government and the failure, bankruptcy, or sale of various financial and other institutions. We believe the current economic conditions and financial market turmoil could adversely affect our operations, business and prospects, as well as our ability to obtain funds. If these circumstances persist or continue to worsen, our future operating results could be adversely affected, particularly relative to our current expectations.

We are at an early stage of development as a company and we do not have, and may never have, any products that generate significant revenues.

We are at an early stage of development as a company and have a limited operating history on which to evaluate our business and prospects. While we have earned modest product revenues from the ALIGN business acquired in October 2007, since beginning operations in 1996, we have not generated any product revenues from our product candidates currently in development. We cannot guarantee that any of our product candidates currently in development will ever become marketable products and we do not anticipate material revenues from the ALIGN products in the foreseeable future.

We must demonstrate that our drug candidates satisfy rigorous standards of safety and efficacy for their intended uses before the FDA, and other regulatory authorities in the United States, the European Union and elsewhere. Significant additional research, preclinical testing and clinical testing is required before we can file applications with the FDA or other regulatory authorities for premarket approval of our drug candidates. In addition, to compete effectively, our drugs must be easy to administer, cost-effective and economical to manufacture on a commercial scale. We may not achieve any of these objectives. Sapacitabine, our most advanced drug candidates for the treatment of cancer, is currently in Phase 3 for AML and Phase 2 for MDS. Seliciclib is currently in Phase 2 clinical trials. A combination trial of sapacitabine and seliciclib is currently in a Phase 1 clinical trial. We cannot be certain that the clinical development of these or any other drug candidates in preclinical testing or clinical development will be successful, that we will receive the regulatory approvals required to commercialize them or that any of our other research and drug discovery programs will yield a drug candidate suitable for investigation through clinical trials. Our commercial revenues from our product candidates currently in development, if any, will be derived from sales of drugs that will not become marketable for several years, if at all.

We have a history of operating losses and we may never become profitable. Our stock is a highly speculative investment.

We have incurred operating losses in each year since beginning operations in 1996 due to costs incurred in connection with our research and development activities and selling, general and administrative costs associated with our operations, and we may never achieve profitability. As of December 31, 2010 and June 30, 2011, our accumulated deficit was \$241.8 million and \$250.0 million, respectively. Our net loss for the six months ended June 30, 2010 and 2011 was \$9.0 million and \$8.1 million, respectively. Our net loss applicable to common stockholders from inception through June 30, 2011 was \$291.3 million. Our drug candidates are in the midstages of clinical testing and we must conduct significant additional clinical trials before we can seek the regulatory approvals necessary to begin commercial sales of our drugs. We expect to incur continued losses for several years, as we continue our research and development of our drug candidates, seek regulatory approvals, commercialize any approved drugs and market and promote the ALIGN products: Xclair® Cream, Numoisyn® Liquid and Numoisyn® Lozenges. If our drug candidates are unsuccessful in clinical trials or we are unable to obtain regulatory approvals, or if our drugs are unsuccessful in the market, we will not be profitable. If we fail to become and remain profitable, or if we are unable to fund our continuing losses, particularly in light of the current economic conditions, you could lose all or part of your investment.

Capital markets are currently experiencing a period of disruption and instability, which has had and could continue to have a negative impact on the availability and cost of capital.

The general disruption in the United States capital markets has impacted the broader worldwide financial and credit markets and reduced the availability of debt and equity capital for the market as a whole. These global conditions could persist for a prolonged period of time or worsen in the future. Our ability to access the capital markets may be restricted at a time when we would like, or need, to access those markets, which could have an impact on our flexibility to react to changing economic and business conditions. The resulting lack of available credit, lack of confidence in the financial sector, increased volatility in the financial markets could materially and adversely affect the cost of debt financing and the proceeds of equity financing may be materially adversely impacted by these market conditions.

If we fail to comply with the continued listing requirements of the NASDAQ Global Market our common stock price may be delisted and the price of our common stock and our ability to access the capital markets could be negatively impacted.

Our common stock is currently listed for trading on the NASDAQ Global Market. We must satisfy NASDAQ's continued listing requirements, including among other things, a minimum stockholders' equity of \$10.0 million and a minimum bid price for our common stock of \$1.00 per share, or risk delisting, which would have a material adverse effect on our business. A delisting of our common stock from the NASDAQ Global Market could materially reduce the liquidity of our common stock and result in a corresponding material reduction in the price of our common stock.

In addition, delisting could harm our ability to raise capital through alternative financing sources on terms acceptable to us, or at all, and may result in the potential loss of confidence by investors, suppliers, customers and employees and fewer business development opportunities. During 2009, Cyclacel received notification from the NASDAQ Stock Market that the Company was not in compliance with the minimum \$10 million stockholders' equity requirement for continued listing set forth in NASDAQ Marketplace Rule 5450(b) (1)(A). On January 27, 2010, NASDAQ notified the Company that it regained compliance with the minimum \$50 million market value of listed securities requirement and that it currently complies with all other applicable standards for continued listing on The NASDAQ Global Market. Accordingly, the Company's shares of common and preferred stock will continue to trade on The NASDAQ Global Market.

Raising additional capital in the future may not be available to us on reasonable terms, if at all, when or as we require additional funding. If we issue additional shares of our common stock or other securities that may be convertible into, or exercisable or exchangeable for, our common stock, our existing stockholders would experience further dilution. If we fail to obtain additional funding, we may be unable to complete the development and commercialization of our lead drug candidate, sapacitabine, or continue to fund our research and development programs.

We have funded all of our operations and capital expenditures with proceeds from the issuance of public equity securities, private placements of our securities, interest on investments, licensing revenue, government grants, research and development tax credits and product revenue. In order to conduct the lengthy and expensive research, preclinical testing and clinical trials necessary to complete the development and marketing of our drug candidates, we will require substantial additional funds. Based on our current operating plans of focusing on the advancement of sapacitabine, we expect our existing resources to be sufficient to fund our planned operations for at least the next twelve months. To meet our long-term financing requirements, we may raise funds through public or private equity offerings, debt financings or strategic alliances. Raising additional funds by issuing equity or convertible debt securities may cause our stockholders to experience substantial dilution in their ownership interests and new investors may have rights superior to the rights of our other stockholders. Raising additional funds through debt financing, if available, may involve covenants that restrict our business activities and options. To the extent that we raise additional funds through collaborations and licensing arrangements, we may have to relinquish valuable rights to our drug discovery and other technologies, research programs or drug candidates, or grant licenses on terms that may not be favorable to us. Additional funding may not be available to us on favorable terms, or at all, particularly in light of the current economic conditions. If we are unable to obtain additional funds, we may be forced to delay or terminate our current clinical trials and the development and marketing of our drug candidates including sapacitabine.

To the extent we elect to fund the development of a drug candidate or the commercialization of a drug at our expense, we will need substantial additional funding.

We plan to market drugs on our own, with or without a partner, that can be effectively commercialized and sold in concentrated markets that do not require a large sales force to be competitive. To achieve this goal, we will need to establish our own specialized sales force, marketing organization and supporting distribution capabilities. The development and commercialization of our drug candidates is very expensive, including our Phase 3 clinical trials for sapacitabine. To the extent we elect to fund the full development of a drug candidate or the commercialization of a drug at our expense, we will need to raise substantial additional funding to:

- fund research and development and clinical trials connected with our research;
- fund clinical trials and seek regulatory approvals;
- build or access manufacturing and commercialization capabilities;
- implement additional internal control systems and infrastructure;

- commercialize and secure coverage, payment and reimbursement of our drug candidates, if any such candidates receive regulatory approval;
- maintain, defend and expand the scope of our intellectual property; and
- hire additional management, sales and scientific personnel.

Our future funding requirements will depend on many factors, including:

- the scope, rate of progress and cost of our clinical trials and other research and development activities;
- the costs and timing of seeking and obtaining regulatory approvals;
- the costs of filing, prosecuting, defending and enforcing any patent claims and other intellectual property rights;
- the costs associated with establishing sales and marketing capabilities;
- the costs of acquiring or investing in businesses, products and technologies;
- the effect of competing technological and market developments; and
- · the payment, other terms and timing of any strategic alliance, licensing or other arrangements that we may establish.

If we are not able to secure additional funding when needed, especially in light of the current economic conditions and financial market turmoil, we may have to delay, reduce the scope of or eliminate one or more of our clinical trials or research and development programs or future commercialization efforts.

Any future workforce and expense reductions may have an adverse impact on our internal programs, strategic plans, and our ability to hire and retain key personnel, and may also be distracting to our management.

Further workforce and expense reductions in addition to those carried out in September 2008 and June 2009 could result in significant delays in implementing our strategic plans. In addition, employees, whether or not directly affected by such reduction, may seek future employment with our business partners or competitors. Although our employees are required to sign a confidentiality agreement at the time of hire, the confidential nature of certain proprietary information may not be maintained in the course of any such future employment. In addition, any additional workforce reductions or restructurings would be expected to involve significant expense as a result of contractual terms in certain of our existing agreements, including potential severance obligations as well as any payments that may, under certain circumstances, be required under our agreement with the Scottish Enterprise. Further, we believe that our future success will depend in large part upon our ability to attract and retain highly skilled personnel. We may have difficulty retaining and attracting such personnel as a result of a perceived risk of future workforce and expense reductions. Finally, the implementation of expense reduction programs may result in the diversion of the time and attention of our executive management team and other key employees, which could adversely affect our business.

Budget constraints resulting from our restructuring plan may negatively impact our research and development, forcing us to delay our efforts to develop certain product candidates in favor of developing others, which may prevent us from commercializing our product candidates as quickly as possible.

Research and development is an expensive process. As part of our restructuring plan, we have decided to focus our clinical development priorities on sapacitabine, while still possibly continuing to progress additional programs pending the availability of clinical data and the availability of funds, at which time we will determine the feasibility of pursuing, if at all, further advanced development of seliciclib, CYC116 or additional programs. Because we have had to prioritize our development candidates as a result of budget constraints, we may not be able to fully realize the value of our product candidates in a timely manner, if at all.

We are exposed to risks related to foreign currency exchange rates.

Some of our costs and expenses are denominated in foreign currencies. Most of our foreign expenses are associated with our research and development operations of our United Kingdom-based wholly-owned subsidiary. When the United States dollar weakens against the British pound, the United States dollar value of the foreign currency denominated expense increases, and when the United States dollar strengthens against the British pound, the United States dollar value of the foreign currency denominated expense decreases. Consequently, changes in exchange rates, and in particular a weakening of the United States dollar, may adversely affect our results of operations

Risks Related to our Intellectual Property

We may be subject to damages resulting from claims that our employees or we have wrongfully used or disclosed alleged trade secrets of their former employers.

Many of our employees were previously employed at universities or other biotechnology or pharmaceutical companies, including our competitors or potential competitors. Although no claims against us are currently pending, we may be subject to claims that these employees or we have inadvertently or otherwise used or disclosed trade secrets or other proprietary information of their former employers. Litigation may be necessary to defend against these claims. If we fail in defending such claims, in addition to paying monetary damages, we may lose valuable intellectual property rights or personnel. A loss of key research personnel or their work product could hamper or prevent our ability to commercialize certain potential drugs, which could severely harm our business. Even if we are successful in defending against these claims, litigation could result in substantial costs and be a distraction to management.

If we fail to enforce adequately or defend our intellectual property rights our business may be harmed.

Our commercial success depends in large part on obtaining and maintaining patent and trade secret protection for our drug candidates, the methods used to manufacture those drug candidates and the methods for treating patients using those drug candidates.

Specifically, sapacitabine is covered in granted, composition of matter patents that expire in 2014 in the United States and 2012 outside the United States. Sapacitabine is further protected by additional granted, composition of matter patents claiming certain, stable crystalline forms of sapacitabine and their pharmaceutical compositions and therapeutic uses that expire in 2022 and also patent applications claiming the combination of sapacitabine with decitabine which is being tested as one of the arms of the SEAMLESS Phase 3 trial . In early development, amorphous sapacitabine was used. We have used one of the stable, crystalline forms of sapacitabine in nearly all our Phase 1 and in all of our Phase 2 clinical studies. We have also chosen this form for commercialization. Additional patents claim certain medical uses and formulations of sapacitabine which have emerged in our clinical trials. Seliciclib is protected by granted, composition of matter patents that expire in 2016. Additional patents claim certain medical uses which have emerged from our research programs.

Failure to obtain, maintain or extend the patents could adversely affect our business. We will only be able to protect our drug candidates and our technologies from unauthorized use by third parties to the extent that valid and enforceable patents or trade secrets cover them.

Our ability to obtain patents is uncertain because legal means afford only limited protections and may not adequately protect our rights or permit it to gain or keep any competitive advantage. Some legal principles remain unresolved and the breadth or interpretation of claims allowed in patents in the United States, the European Union or elsewhere can still be difficult to ascertain or predict. In addition, the specific content of patents and patent applications that are necessary to support and interpret patent claims is highly uncertain due to the complex nature of the relevant legal, scientific and factual issues. Changes in either patent laws or in interpretations of patent laws in the United States, the European Union or elsewhere may diminish the value of our intellectual property or narrow the scope of our patent protection. Our existing patents and any future patents we obtain may not be sufficiently broad to prevent others from practicing our technologies or from developing competing products and technologies. In addition, we generally do not control the patent prosecution of subject matter that we license from others and have not controlled the earlier stages of the patent prosecution. Accordingly, we are unable to exercise the same degree of control over this intellectual property as we would over our own.

Even if patents are issued regarding our drug candidates or methods of using them, those patents can be challenged by our competitors who may argue such patents are invalid and/or unenforceable. Patents also will not protect our drug candidates if competitors devise ways of making or using these product candidates without legally infringing our patents. The U.S. Federal Food, Drug and Cosmetic, or FD&C, Act and FDA regulations and policies and equivalents in other jurisdictions provide incentives to manufacturers to challenge patent validity or create modified, noninfringing versions of a drug in order to facilitate the approval of abbreviated new drug applications for generic substitutes. These same types of incentives encourage manufacturers to submit new drug applications that rely on literature and clinical data not prepared for or by the drug sponsor.

Proprietary trade secrets and unpatented know-how are also very important to our business. We rely on trade secrets to protect our technology, especially where we do not believe that patent protection is appropriate or obtainable. However, trade secrets are difficult to protect. Our employees, consultants, contractors, outside scientific collaborators and other advisors may unintentionally or willfully disclose our confidential information to competitors, and confidentiality agreements may not provide an adequate remedy in the event of unauthorized disclosure of confidential information. Enforcing a claim that a third party obtained illegally and is using trade secrets is expensive and time consuming, and the outcome is unpredictable. Moreover, our competitors may independently develop equivalent knowledge, methods and know-how. Failure to obtain or maintain trade secret protection could adversely affect our competitive business position.

Intellectual property rights of third parties may increase our costs or delay or prevent us from being able to commercialize our drug candidates and/or the ALIGN products.

There is a risk that we are infringing or will infringe the proprietary rights of third parties because patents and pending applications belonging to third parties exist in the United States, the European Union and elsewhere in the world in the areas of our research and/or the ALIGN products. Others might have been the first to make the inventions covered by each of our or our licensors' pending patent applications and issued patents and might have been the first to file patent applications for these inventions. We are aware of several published patent applications, and understand that others may exist, that could support claims that, if granted, could cover various aspects of our developmental programs, including in some cases particular uses of our lead drug candidate sapacitabine, seliciclib or other therapeutic candidates, or gene sequences and techniques that we use in the course of our research and development. In addition, we understand that other applications and patents exist relating to potential uses of sapacitabine and seliciclib that are not part of our current clinical programs for these compounds. Numerous third-party United States and foreign issued patents and pending applications exist in the area of kinases, including CDK, AK and Plk for which we have research programs. For example, some pending patent applications contain broad claims that could represent freedom to operate limitations for some of our kinase programs should they be issued unchanged. Although we intend to continue to monitor these applications, we cannot predict what claims will ultimately be allowed and if allowed what their scope would be. In addition, because the patent application process can take several years to complete, there may be currently pending applications, unknown to us, which may later result in issued patents that cover the production, manufacture, commercialization or use of our drug candidates. If we wish to use the technology or compound claimed in issued and unexpired patents owned by others, we will need to obtain a license from the owner, enter into litigation to challenge the validity of the patents or incur the risk of litigation in the event that the owner asserts that we infringe its patents. In one case we have opposed a European patent relating to human aurora kinase and the patent has been finally revoked (no appeal was filed). We are also aware of a corresponding U.S. patent containing method of treatment claims for specific cancers using aurora kinase modulators which, if held valid, could potentially restrict the use of our aurora kinase inhibitors once clinical trials are completed.

There has been substantial litigation and other proceedings regarding patent and other intellectual property rights in the pharmaceutical and biotechnology industries. Defending against third party claims, including litigation in particular, would be costly and time consuming and would divert management's attention from our business, which could lead to delays in our development or commercialization efforts. If third parties are successful in their claims, we might have to pay substantial damages or take other actions that are adverse to our business. As a result of intellectual property infringement claims, or to avoid potential claims, we might:

- be prohibited from selling or licensing any product that we may develop unless the patent holder licenses the patent to us, which it is not required to do;
- be required to pay substantial royalties or grant a cross license to our patents to another patent holder;
- decide to move some of our screening work outside Europe;
- be required to pay substantial damages for past infringement, which we may have to pay if a court determines that our product candidates or technologies infringe a competitor's patent or other proprietary rights; or
- be required to redesign the formulation of a drug candidate so it does not infringe, which may not be possible or could require substantial funds and time.

Risks Related to Securities Regulations and Investment in Our Securities

Failure to achieve and maintain internal controls is accordance with Sections 302 and 404 of the Sarbanes-Oxley Act of 2002 could have a material adverse effect on our business and stock price.

During March 2011, we identified a deficiency in respect of our internal controls over financial reporting, specifically our controls over the accounting for cumulative preferred stock dividends, that constitutes a material weakness as described in the SEC's guidance regarding Management's Report on Internal Control Over Financial Reporting as of December 31, 2010. As a result of this deficiency, the financial statements included in our Form 10-K for the year ended December 31, 2009, filed on March 29, 2010, as amended by Amendment No. 1 to our Annual Report on Form 10-K/A for the year ended December 31, 2009 filed on May 17, 2010 and, as further amended by Amendment No. 2 to our Annual Report on Form 10-K/A for the year ended December 31, 2009 filed on May 19, 2010, included errors related to the presentation and disclosure of undeclared cumulative preferred stock dividends in the consolidated balance sheet and in the statement of stockholders' equity. Unaudited balance sheets for the each of the first three quarters of 2009 and 2010 also contained errors. In addition, in May 2010, we filed an amendment to our Annual Report on Form 10-K for the year ended December 31, 2009, to report a restatement of our financial statements and report a material weakness in our internal control over financial reporting as of December 31, 2009, specifically related to the operational failure of the controls in place to ensure the correct computation of net loss per share and presentation of preferred stock dividends in the consolidated statement of cash flows. In March 2011, our auditors identified a further error in respect of the accounting, presentation and disclosure of cumulative undeclared preferred stock dividends, specifically the inclusion of undeclared cumulative preferred stock dividends as a current liability in its consolidated financial statements, which resulted from the same material weakness as described above. This has resulted in the restatement of our consolidated balance sheets as of March 31, 2009, June 30, 2009, September 30, 2009, December 31, 2009, March 31, 2010, June 30, 2010, and September 30, 2010 and consolidated statement of stockholders' equity for the year ended December 31, 2009 and selected financial data as of and for the year ended December 31, 2009.

If we fail to maintain our internal controls or fail to implement required new or improved controls, as such control standards are modified, supplemented or amended from time to time, we may not be able to conclude on an ongoing basis that we have effective internal controls over financial reporting. Effective internal controls are necessary for us to produce reliable financial reports and are important in the prevention of financial fraud. If we cannot produce reliable financial reports or prevent fraud, our business and operating results could be harmed.

We incur increased costs and management resources as a result of being a public company, and we still may fail to comply with public company obligations.

As a public company, we face and will continue to face increased legal, accounting, administrative and other costs and expenses as a public company that we would not incur as a private company. Compliance with the Sarbanes Oxley Act of 2002, as well as other rules of the SEC, the Public Company Accounting Oversight Board and the NASDAQ Global Market resulted in a significant initial cost to us as well as an ongoing compliance costs. As a public company, we are subject to Section 404 of the Sarbanes Oxley Act relating to internal control over financial reporting. We have completed a formal process to evaluate our internal controls for purposes of Section 404, and we concluded that as of December 31, 2010, our internal control over financial reporting was ineffective. As our business grows and changes, there can be no assurances that we can maintain the effectiveness of our internal controls over financial reporting.

Effective internal controls over financial reporting are necessary for us to provide reliable financial reports and, together with adequate disclosure controls and procedures, are designed to prevent fraud. If we cannot provide reliable financial reports or prevent fraud, our operating results could be harmed. We have completed a formal process to evaluate our internal control over financial reporting. However, guidance from regulatory authorities in the area of internal controls continues to evolve and substantial uncertainty exists regarding our on-going ability to comply by applicable deadlines. Any failure to implement required new or improved controls, or difficulties encountered in their implementation, could harm our operating results or cause us to fail to meet our reporting obligations. Ineffective internal controls could also cause investors to lose confidence in our reported financial information, which could have a negative effect on the trading price of our common stock.

Our common stock may have a volatile public trading price.

An active public market for our common stock has not developed. Our stock can trade in small volumes which may make the price of our stock highly volatile. The last reported price of our stock may not represent the price at which you would be able to buy or sell the stock. The market prices for securities of companies comparable to us have been highly volatile. Often, these stocks have experienced significant price and volume fluctuations for reasons that are both related and unrelated to the operating performance of the individual companies. In addition, the stock market as a whole and biotechnology and other life science stocks in particular have experienced significant recent volatility. Like our common stock, these stocks have experienced significant price and volume fluctuations for reasons unrelated to the operating performance of the individual companies. In addition, due to our existing stock price, we may not continue to qualify for continued listing on the NASDAQ Global Market. To maintain listing, we are required to maintain a minimum closing bid price of \$1.00 per share and, among other requirements, to maintain a minimum stockholders equity value of \$10 million. Factors giving rise to this volatility may include:

- disclosure of actual or potential clinical results with respect to product candidates we are developing;
- regulatory developments in both the United States and abroad;
- developments concerning proprietary rights, including patents and litigation matters;

- public concern about the safety or efficacy of our product candidates or technology, or related technology, or new technologies generally;
- concern about the safety or efficacy of our product candidates or technology, or related technology, or new technologies generally;
- · public announcements by our competitors or others; and
- general market conditions and comments by securities analysts and investors.

Fluctuations in our operating losses could adversely affect the price of our common stock.

Our operating losses may fluctuate significantly on a quarterly basis. Some of the factors that may cause our operating losses to fluctuate on a period-to-period basis include the status of our preclinical and clinical development programs, level of expenses incurred in connection with our preclinical and clinical development programs, implementation or termination of collaboration, licensing, manufacturing or other material agreements with third parties, non-recurring revenue or expenses under any such agreement, and compliance with regulatory requirements. Period-to-period comparisons of our historical and future financial results may not be meaningful, and investors should not rely on them as an indication of future performance. Our fluctuating losses may fail to meet the expectations of securities analysts or investors. Our failure to meet these expectations may cause the price of our common stock to decline.

If securities or industry analysts do not publish research or reports about us, if they change their recommendations regarding our stock adversely or if our operating results do not meet their expectations, our stock price and trading volume could decline.

The trading market for our common stock is influenced by the research and reports that industry or securities analysts publish about us. If one or more of these analysts cease coverage of us or fail to regularly publish reports on us, we could lose visibility in the financial markets, which in turn could cause our stock price or trading volume to decline. Moreover, if one or more of the analysts who cover us downgrade our stock or if our operating results do not meet their expectations, our stock price could decline.

Anti-takeover provisions in our charter documents and provisions of Delaware law may make an acquisition more difficult and could result in the entrenchment of management.

We are incorporated in Delaware. Anti-takeover provisions of Delaware law and our amended and restated certificate of incorporation and amended and restated bylaws may make a change in control or efforts to remove management more difficult. Also, under Delaware law, our Board of Directors may adopt additional anti-takeover measures.

We have the authority to issue up to 5 million shares of preferred stock and to determine the terms of those shares of stock without any further action by our stockholders. If the Board of Directors exercises this power to issue preferred stock, it could be more difficult for a third party to acquire a majority of our outstanding voting stock and vote the stock they acquire to remove management or directors.

Our amended and restated certificate of incorporation and amended and restated bylaws also provides staggered terms for the members of our Board of Directors. Under Section 141 of the Delaware General Corporation Law, our directors may be removed by stockholders only for cause and only by vote of the holders of a majority of voting shares then outstanding. These provisions may prevent stockholders from replacing the entire board in a single proxy contest, making it more difficult for a third party to acquire control of us without the consent of our Board of Directors. These provisions could also delay the removal of management by the Board of Directors with or without cause. In addition, our directors may only be removed for cause and amended and restated bylaws limit the ability our stockholders to call special meetings of stockholders.

Under Section 203 of the Delaware General Corporation Law, a corporation may not engage in a business combination with any holder of 15% or more of its capital stock until the holder has held the stock for three years unless, among other possibilities, the Board of Directors approves the transaction. Our Board of Directors could use this provision to prevent changes in management. The existence of the foregoing provisions could limit the price that investors might be willing to pay in the future for shares of our common stock.

We have the authority to issue up to 5 million shares of preferred stock and to determine the terms of those shares of stock without any further action by our stockholders. If the Board of Directors exercises this power to issue preferred stock, it could be more difficult for a third party to acquire a majority of our outstanding voting stock and vote the stock they acquire to remove management or directors.

Our amended and restated certificate of incorporation and amended and restated bylaws also provides staggered terms for the members of our Board of Directors. Under Section 141 of the Delaware General Corporation Law, our directors may be removed by stockholders only for cause and only by vote of the holders of a majority of voting shares then outstanding. These provisions may prevent stockholders from replacing the entire board in a single proxy contest, making it more difficult for a third party to acquire control of us without the consent of our Board of Directors. These provisions could also delay the removal of management by the Board of Directors with or without cause. In addition, our directors may only be removed for cause and amended and restated bylaws limit the ability our stockholders to call special meetings of stockholders.

Under Section 203 of the Delaware General Corporation Law, a corporation may not engage in a business combination with any holder of 15% or more of its capital stock until the holder has held the stock for three years unless, among other possibilities, the Board of Directors approves the transaction. Our Board of Directors could use this provision to prevent changes in management. The existence of the foregoing provisions could limit the price that investors might be willing to pay in the future for shares of our common stock.

Certain severance-related agreements in our executive employment agreements may make an acquisition more difficult and could result in the entrenchment of management.

In March 2008 (as subsequently amended, most recently as of January 1, 2011), we entered into employment agreements with our President and Chief Executive Officer and our Executive Vice President, Finance, Chief Financial Officer and Chief Operating Officer, which contain severance arrangements in the event that such executive's employment is terminated without "cause" or as a result of a "change of control" (as each such term is defined in each agreement). The financial obligations triggered by these provisions may prevent a business combination or acquisition that would be attractive to stockholders and could limit the price that investors would be willing to pay in the future for our stock.

In the event of an acquisition of our common stock, we cannot assure our common stockholders that we will be able to negotiate terms that would provide for a price equivalent to, or more favorable than, the price at which our shares of common stock may be trading at such time.

We may not effect a consolidation or merger with another entity without the vote or consent of the holders of at least a majority of the shares of our preferred stock (in addition to the approval of our common stockholders), unless the preferred stock that remains outstanding and its rights, privileges and preferences are unaffected or are converted into or exchanged for preferred stock of the surviving entity having rights, preferences and limitations substantially similar, but no less favorable, to our convertible preferred stock.

In addition, in the event a third party seeks to acquire our company or acquire control of our company by way of a merger, but the terms of such offer do not provide for our preferred stock to remain outstanding or be converted into or exchanged for preferred stock of the surviving entity having rights, preferences and limitations substantially similar, but no less favorable, to our preferred stock, the terms of the Certificate of Designation of our preferred stock provide for an adjustment to the conversion ratio of our preferred stock such that, depending on the terms of any such transaction, preferred stockholders may be entitled, by their terms, to receive up to \$10.00 per share in common stock, causing our common stockholders not to receive as favorable a price as the price at which such shares may be trading at the time of any such transaction.

As of June 30, 2011, there were 1,213,142 shares of our preferred stock issued and outstanding. If the transaction were one in which proceeds were received by the Company for distribution to stockholders, and the terms of the Certificate of Designation governing the preferred stock were strictly complied with, approximately \$13,344,563 would be paid to the preferred holders before any distribution to the common stockholders, although the form of transaction could affect how the holders of preferred stock are treated. In such an event, although such a transaction would be subject to the approval of our holders of common stock, we cannot assure our common stockholders that we will be able to negotiate terms that would provide for a price equivalent to, or more favorable than, the price at which our shares of common stock may be trading at such time. Thus, the terms of our preferred stock might hamper a third party's acquisition of our company.

Our certificate of incorporation and bylaws and certain provisions of Delaware law may delay or prevent a change in our management and make it more difficult for a third party to acquire us.

Our amended and restated certificate of incorporation and bylaws contain provisions that could delay or prevent a change in our Board of Directors and management teams. Some of these provisions:

- authorize the issuance of preferred stock that can be created and issued by the Board of Directors without prior stockholder approval, commonly referred to as "blank check" preferred stock, with rights senior to those of our common stock;
- provide for the Board of Directors to be divided into three classes; and
- require that stockholder actions must be effected at a duly called stockholder meeting and prohibit stockholder action by written consent.

In addition, because we are incorporated in Delaware, we are governed by the provisions of Section 203 of the Delaware General Corporation Law, which limits the ability of large stockholders to complete a business combination with, or acquisition of, us. These provisions may prevent a business combination or acquisition that would be attractive to stockholders and could limit the price that investors would be willing to pay in the future for our stock.

These provisions also make it more difficult for our stockholders to replace members of our Board of Directors. Because our Board of Directors is responsible for appointing the members of our management team, these provisions could in turn affect any attempt to replace our current management team. Additionally, these provisions may prevent an acquisition that would be attractive to stockholders and could limit the price that investors would be willing to pay in the future for our common stock.

We may have limited ability to pay cash dividends on our preferred stock, and there is no assurance that future quarterly dividends will be declared.

Delaware law may limit our ability to pay cash dividends on our preferred stock. Under Delaware law, cash dividends on our preferred stock may only be paid from surplus or, if there is no surplus, from the corporation's net profits for the current or preceding fiscal year. Delaware law defines "surplus" as the amount by which the total assets of a corporation, after subtracting its total liabilities, exceed the corporation's capital, as determined by its Board of Directors.

Since we are not profitable, our ability to pay cash dividends will require the availability of adequate surplus. Even if adequate surplus is available to pay cash dividends on our preferred stock, we may not have sufficient cash to pay dividends on the preferred stock or we may choose not to declare the dividends.

On July 8, 2011, the Board of Directors decided not to declare the quarterly cash dividend on the preferred stock with respect to the second quarter of 2011 that would have otherwise been payable on August 1, 2011. In addition, the Board of Directors also did not declare the quarterly cash dividend with respect to each of the four quarters of fiscal year 2009 and the first, second and third quarters of 2010.

We have granted additional rights to our holders of preferred stock with respect to the management of the Company as a result of having not declared quarterly dividends on our preferred stock for a total of six quarterly dividend periods.

As a result of having not declared quarterly dividends on our preferred stock for a total of six quarters, the size of the Company's Board was increased by two members and the holders of the preferred stock, voting separately as a class, voted on May 24, 2011 and elected two directors to fill the vacancies created thereby. The directors elected by the holders of the preferred stock will have the ability to participate in the management of the Company until all accrued but unpaid dividends have been paid in full.

Our common and convertible preferred stock may experience extreme price and volume fluctuations, which could lead to costly litigation for the Company and make an investment in the Company less appealing.

The market price of our common and convertible preferred stock may fluctuate substantially due to a variety of factors, including:

- additions to or departures of our key personnel;
- announcements of technological innovations or new products or services by us or our competitors;
- announcements concerning our competitors or the biotechnology industry in general;
- new regulatory pronouncements and changes in regulatory guidelines;
- general and industry-specific economic conditions;
- changes in financial estimates or recommendations by securities analysts;
- variations in our quarterly results;
- announcements about our collaborators or licensors; and
- · changes in accounting principles.

The market prices of the securities of biotechnology companies, particularly companies like us without product revenues and earnings, have been highly volatile and are likely to remain highly volatile in the future. This volatility has often been unrelated to the performance of particular companies. In the past, companies that experience volatility in the market price of their securities have often faced securities class action litigation. Moreover, market prices for stocks of biotechnology-related and technology companies frequently reach levels that bear no relationship to the performance of these companies. These market prices generally are not sustainable and are highly volatile. Whether or not meritorious, litigation brought against us could result in substantial costs, divert our management's attention and resources and harm our financial condition and results of operations. In addition, due to our stock price from time to time, we may not continue to qualify for continued listing on the NASDAQ Global Market. Please see Risk Factor: *Our common stock may have a volatile public trading price*.

The future sale of our common and preferred stock and future issuances of our common stock upon conversion of our preferred stock could negatively affect our stock price and cause dilution to existing holders of our common stock.

If our common or preferred stockholders sell substantial amounts of our stock in the public market, or the market perceives that such sales may occur, the market price of our common and preferred stock could fall. For example, we were approached by a preferred stockholder that elected to convert 123,400 of its shares of preferred stock, which shares were converted into 239,396 shares of common stock in the first quarter of 2010. In addition, 710,271 shares of preferred stock were converted to 1,416,203 shares of common stock during the second quarter of 2010. If additional holders of preferred stock elect to convert their shares to shares of common stock at renegotiated prices, such conversion as well as the sale of substantial amounts of our common or preferred stock, could cause dilution to existing holders of our common stock, thereby also negatively affecting the price of our common stock.

If we exchange the convertible preferred stock for debentures, the exchange will be taxable but we will not provide any cash to pay any tax liability that any convertible preferred stockholder may incur.

An exchange of convertible preferred stock for debentures, as well as any dividend make-whole or interest make-whole payments paid in our common stock, will be taxable events for United States federal income tax purposes, which may result in tax liability for the holder of convertible preferred stock without any corresponding receipt of cash by the holder. In addition, the debentures may be treated as having original issue discount, a portion of which would generally be required to be included in the holder's gross income even though the cash to which such income is attributable would not be received until maturity or redemption of the debenture. We will not distribute any cash to the holders of the securities to pay these potential tax liabilities.

If we automatically convert the convertible preferred stock, there is a substantial risk of fluctuation in the price of our common stock from the date we elect to automatically convert to the conversion date.

We may automatically convert the convertible preferred stock into common stock if the closing price of our common stock has exceeded \$35.30. There is a risk of fluctuation in the price of our common stock between the time when we may first elect to automatically convert the preferred and the automatic conversion date.

We do not intend to pay cash dividends on our common stock in the foreseeable future.

We do not anticipate paying cash dividends on our common stock in the foreseeable future. Any payment of cash dividends will depend on our financial condition, results of operations, capital requirements, the outcome of the review of our strategic alternatives and other factors and will be at the discretion of our Board of Directors. Accordingly, investors will have to rely on capital appreciation, if any, to earn a return on their investment in our common stock. Furthermore, we may in the future become subject to contractual restrictions on, or prohibitions against, the payment of dividends.

The number of shares of common stock which are registered, including the shares to be issued upon exercise of our outstanding warrants, is significant in relation to our currently outstanding common stock and could cause downward pressure on the market price for our common stock.

The number of shares of common stock registered for resale, including those shares which are to be issued upon exercise of our outstanding warrants, is significant in relation to the number of shares of common stock currently outstanding. If the security holder determines to sell a substantial number of shares into the market at any given time, there may not be sufficient demand in the market to purchase the shares without a decline in the market price for our common stock. Moreover, continuous sales into the market of a number of shares in excess of the typical trading volume for our common stock, or even the availability of such a large number of shares, could depress the trading market for our common stock over an extended period of time.

If persons engage in short sales of our common stock, including sales of shares to be issued upon exercise of our outstanding warrants, the price of our common stock may decline.

Selling short is a technique used by a stockholder to take advantage of an anticipated decline in the price of a security. In addition, holders of options and warrants will sometimes sell short knowing they can, in effect, cover through the exercise of an option or warrant, thus locking in a profit. A significant number of short sales or a large volume of other sales within a relatively short period of time can create downward pressure on the market price of a security. Further sales of common stock issued upon exercise of our outstanding warrants could cause even greater declines in the price of our common stock due to the number of additional shares available in the market upon such exercise, which could encourage short sales that could further undermine the value of our common stock. You could, therefore, experience a decline in the value of your investment as a result of short sales of our common stock.

We are exposed to risk related to the marketable securities we may purchase.

We may invest cash not required to meet short term obligations in short term marketable securities. We may purchase securities in United States government, government-sponsored agencies and highly rated corporate and asset-backed securities subject to an approved investment policy. Historically, investment in these securities has been highly liquid and has experienced only very limited defaults. However, recent volatility in the financial markets has created additional uncertainty regarding the liquidity and safety of these investments. Although we believe our marketable securities investments are safe and highly liquid, we cannot guarantee that our investment portfolio will not be negatively impacted by recent or future market volatility or credit restrictions.

Item 2. Unregistered Sales of Equity Securities and Use of Proceeds.

None.

Item 3. Defaults upon Senior Securities.

None.

Item 4. (Removed and Reserved).

Item 5. Other Information.

None.

Item 6. Exhibits.

- 1.1 Underwriting Agreement, dated as of June 30, 2011, by and among the Company, Leerink Swann LLC and Lazard Capital Markets LLC, as representative of the several underwriters named therein (previously filed as Exhibit 1.1 to the Registrant's Current Report on Form 8-K, originally filed with the SEC on July 1, 2011 (File No. 000-50625), and incorporated herein by reference).
- 4.1 Form of Warrant to purchase shares of Cyclacel Pharmaceuticals, Inc. Common Stock (previously filed as Exhibit 4.1 to the Registrant's Current Report on Form 8-K, originally filed with the SEC on July 1, 2011 (File No. 000-50625), and incorporated herein by reference).
- 10.1*† License Agreement between Sankyo Co., Ltd. and Cyclacel Limited, dated September 10, 2003, and letter amendments dated April 1, 2004 and April 28, 2004.
- 10.2*† Amendment No. 4 to License Agreement between Daiichi Sankyo Company, Limited and Cyclacel Limited, dated July 11, 2011.
- 31.1* Certification of Principal Executive Officer Pursuant to Securities Exchange Act Rule 13a-14(a) As Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 31.2* Certification of Principal Financial Officer Pursuant to Securities Exchange Act Rule 13a-14(a) As Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
- 32.1** Certification of Principal Executive Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
- 32.2** Certification of Principal Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
- The following materials from Cyclacel Pharmaceuticals, Inc.'s Quarterly Report on Form 10-Q for the quarter ended June 30, 2011, formatted in XBRL (Extensible Business Reporting Language): (i) the Condensed Consolidated Statements of Income, (ii) the Condensed Consolidated Balance Sheets, (iii) the Condensed Consolidated Statements of Cash Flows, and (iv) Notes to Condensed Consolidated Financial Statements.

- † Certain portions of the exhibit have been omitted pursuant to a confidential treatment request filed separately with the Securities and Exchange Commission.
- * Filed herein.
- ** Furnished herewith.
- *** XBRL (Extensible Business Reporting Language) information is furnished and not filed or a part of a registration statement or prospectus for purposes of Sections 11 or 12 of the Securities Act of 1933, as amended, is deemed not filed for purposes of Section 18 of the Securities Exchange Act of 1934, as amended, and otherwise is not subject to liability under these sections.

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned.

CYCLACEL PHARMACEUTICALS, INC.

Date: August 12, 2011 By: /s/ Paul McBarron

Paul McBarron Chief Operating Officer, Chief Financial Officer and Executive Vice President, Finance

Exhibit 10.1

CONFIDENTIAL TREATMENT REQUESTED

SANKYO CO., LTD.

- and -

CYCLACEL LIMITED

CS-682 LICENCE AGREEMENT

TABLE OF CONTENTS

	Page
1. DEFINITIONS	1
2. LICENCES	11
3. PAYMENTS	13
4. DEVELOPMENT AND COMMERCIALISATION	18
5. INTELLECTUAL PROPERTY — OWNERSHIP	19
6. INTELLECTUAL PROPERTY — PROSECUTION, MAINTENANCE AND ENFORCEMENT	19
7. WARRANTIES AND LIABILITY	22
8. CONFIDENTIALITY	25
9. TERM AND TERMINATION	27
10. EFFECTS OF TERMINATION	28
11. ASSIGNMENT/SUB-CONTRACTING	29
12. FORCE MAJEURE	30
13. GOVERNING LAW	30
14. JURISDICTION	30
15. WAIVER	31
16. SEVERANCE OF TERMS	31
17. ENTIRE AGREEMENT/VARIATIONS	31
18. NOTICES	32
19. COUNTERPARTS	32
20. THIS AGREEMENT NOT TO CONSTITUTE A PARTNERSHIP	33
21. COSTS	33
22. PUBLICITY	33
SCHEDULE 1 DEVELOPMENT PLAN	35
SCHEDULE 2 LICENSED MATERIALS	36
SCHEDULE 3 LICENSED PATENT RIGHTS	1
SCHEDULE 4 CNDAC PATENT RIGHTS	1
SCHEDULE 5 SUCCESSFUL COMPLETION CRITERIA	1
SCHEDULE 6 RETAINED TECHNICAL DATA	1
SCHEDULE 7 COMMITTED CLINICAL TRIALS	1
SCHEDULE 8 ADDRESSES FOR NOTICES	2
SCHEDULE 9 EXPERT'S DECISION	3

THIS AGREEMENT is made on the Commencement Date

BETWEEN:

- (1) SANKYO CO., LTD. a company incorporated in Japan whose principal place of business is at 5-1 Nihonbashi-honcho 3-chome Chuo-ku Tokyo 103-8426 Japan ("Sankyo"); and
- (2) CYCLACEL LIMITED a company incorporated in England whose principal place of business is at Dundee Technopole, James Lindsay Place, Dundee DD1 5JJ, UK ("Cyclacel").

WHEREAS:

- (A) Sankyo owns certain Patent Rights and Know How (as defined herein) relating to a nucleoside analogue known as CS-682 and is the non-exclusive licensee of certain Patent Rights relating to CNDAC.
- (B) Cyclacel is a biotechnology company specialising in the research and development of products in the field of cancer.
- (C) Cyclacel wishes to have an exclusive licence and non-exclusive sub-licence (as appropriate) to the Patent Rights and Know How described in (A) above for the purpose of developing, marketing and selling a product based on CS-682 within the Territory and Sankyo is willing to grant Cyclacel such a licence on the terms set out herein.

1. DEFINITIONS

- 1.1 In this Agreement the following definitions shall apply unless the context requires otherwise:
 - 1.1.1 "Affiliate" any person, corporation, company, partnership, joint venture, limited liability company and/or other entity which Controls, is Controlled by, or is under common Control with a Party.
 - 1.1.2 "Agreement" this document including any and all schedules, appendices and other addenda to it as may be added and/or amended from time to time in accordance with the provisions of this Agreement.
 - 1.1.3 "Blocking IP" any and all intellectual property owned or controlled by a Third Party (but not, subject to Clause 3.6, a Cyclacel Licensee) which would be infringed by the development, manufacture, import, marketing, distribution, sale or other disposal of Product.
 - 1.1.4 "Business Day" 9.30 am to 5.30 pm local time on a day other than a Saturday, Sunday, corporate holiday of a Party, or public holiday in the UK or Japan.

- 1.1.5 "Candidate" the compound known as CS-682 and other compounds covered by Licensed Patent Rights.
- 1.1.6 "CNDAC" the compound 1-(2-C-cyano-2-deoxy-β-D-arabino-pentofurano-syl) cytosine, or known as CNDAC.
- 1.1.7 "Clinical Trials" means any or all of the Phase I Clinical Trials, Phase II Clinical Trials or Phase III Clinical Trials.
- 1.1.8 "CNDAC Patent Rights" any Patent Rights relating to inventions comprised in the CNDAC Patent Rights listed in Schedule 4.
- 1.1.9 "Combination Product" a product containing Product which also:
 - (a) contains a therapeutically active ingredient that is not Licensed IP or CNDAC Patent Rights; or
 - (b) is administered through a therapeutically active administration technology that is not Licensed IP or CNDAC Patent Rights; or
 - (c) is administered in accordance with a diagnostic detection technology that is not Licensed IP or CNDAC Patent Rights.

each of which shall be an "Other Element".

- 1.1.10 "Commencement Date" the date of execution of this Agreement by the Parties, which is the latest date of signature below.
- 1.1.11 "Competent Authority" any national or local agency, authority, department, inspectorate, minister, ministry official, parliament or public or statutory person (whether autonomous or not) of any government of any country having jurisdiction over either any of the activities contemplated by this Agreement or over the Parties, including the European Commission, The Court of First Instance and the European Court of Justice.
- 1.1.12 "Confidential Information" in the case of obligations on Cyclacel in relation to Confidential Information, shall mean Know How and Material forming part of Licensed IP and, in the case of each of Cyclacel and Sankyo, shall mean the terms and conditions of this Agreement notwithstanding the provisions of Clause 22 and any other commercially sensitive information of a Party marked "confidential" supplied or otherwise made available to them, including but not limited to those Know How or Material disclosed by Sankyo to Cyclacel under the Confidential Disclosure Agreement dated February 26, 2002 between the Parties which relates to the subject matter hereof, or coming into their possession in relation to the performance of this Agreement.

- 1.1.13 "Control" means the ownership either directly or indirectly of more than 50% of the issued share capital or any other comparable equity or ownership interest with respect to a business entity or the legal power to direct or cause the direction of the general management and policies of the Party in question.
- 1.1.14 "CS-682" the compound 1-(2-C-cyano-2-deoxy- β -D-arabino-pentofurano-syl)-N(4)-palmitoylcytosine.
- 1.1.15 "Cyclacel Licensee" an entity or individual other than an Affiliate of Cyclacel appointed by Cyclacel as a licensee pursuant to Clause 2.3.
- 1.1.16 "Development Plan" the plan for the development of Candidate and resultant Product by or on behalf of Cyclacel, an outline of which is attached as Schedule 1, which plan may be updated or amended from time to time pursuant to Clause 4.1.
- 1.1.17 "Disclosing Party" a Party which discloses Confidential Information to another Party.
- 1.1.18 "Documents" reports, research notes, charts, graphs, comments, computations, analyses, recordings, photographs, paper, notebooks, books, files, ledgers, records, tapes, discs, diskettes, CD-ROM, computer information storage means and any other media on which Know How can be permanently stored.
- 1.1.19 "EMEA" European Medicines Evaluation Agency or any successor group thereto.
- 1.1.20 "European Union" the countries of the European Union from time to time.
- 1.1.21 "Exceptional Cause"
 - (a) scientific or other technical cause outside Cyclacel's or Cyclacel Licensee's reasonable control or arising through the activities of Third Parties including any significant, unexpected technical problems experienced with the development of Candidate including but not limited to chemical production of Candidate or polymorphic form of Candidate or Product or significant, unexpected problems which relate to the safety, efficacy or toxicology of Candidate, CNDAC or Product;

- (b) difficulty outside Cyclacel's or Cyclacel Licensee's reasonable control in recruitment of patients into trial(s); or
- (c) any significant, unexpected change in the regulatory requirements in a country concerning the development of Candidate or Product which comes into existence after the Commencement Date, for example a new requirement for studies in specific patient sub-groups.
- 1.1.22 "Excluded Territories" [***]
- 1.1.23 "Expert's Decision" the mechanism set out in Schedule 9.
- 1.1.24 "FDA" the U.S. Food and Drug Administration or any successor agency thereto.
- 1.1.25 "Force Majeure" in relation to either Party, any event or circumstance which is beyond the reasonable control of that Party which event or circumstance that Party could not reasonably be expected to have taken into account at the date of this Agreement and which results in or causes the failure of that Party to perform any or all of its obligations under this Agreement including acts of God, lightning, fire, storm, flood, earthquake, requirement for quarantine due to infectious disease outbreak, accumulation of snow or ice, lack of water arising from weather or environmental problems, asteroid or meteor activity, strike, lockout or other industrial or student disturbance, act of the public enemy, war declared or undeclared, threat of war, terrorist act, blockade, revolution, riot, insurrection, civil commotion, public demonstration, sabotage, act of vandalism, prevention from or hindrance in obtaining in any way materials, energy or other supplies, explosion, fault or failure of plant or machinery (which could not have been prevented by Good Industry Practice) provided that lack of funds shall not be interpreted as a cause beyond the reasonable control of that Party.
- 1.1.26 "Good Industry Practice" in relation to any undertaking and any circumstance, the exercise of that degree of skill diligence, prudence and foresight which would reasonably and ordinarily be expected from a skilled and experienced person engaged in the same type of undertaking under the same or similar circumstances.
- 1.1.27 "Insolvency Event" in relation to either Party or one of its Affiliates, means any one of the following:
 - (a) a notice shall have been issued to convene a meeting for the purpose of passing a resolution to wind up that Party, or such a resolution shall have been passed. For the avoidance of doubt a resolution for the solvent reconstruction or reorganisation of that Party or for the purpose of inclusion of any part of the share capital of that Party in the Official List of the London Stock Exchange or in the list of the American Stock Exchange or quotation of the same on the National Association of Securities Dealers Automated Quotation System or other such recognised securities market shall not constitute an Insolvency Event; or

- (b) a resolution shall have been passed by that Party's directors to seek a winding up, or an administration order or a petition for a winding up or administration order shall have been presented against that Party and appeal proceedings have not been commenced by that Party within seven (7) days from the date of such petition or such an order shall have been made; or
- (c) a receiver, administrative receiver, receiver and manager, interim receiver, custodian, sequestrator or similar officer is appointed in respect of that Party or over a substantial part of its assets or any Third Party takes steps to appoint such an officer in respect of that Party or an encumbrancer takes steps to enforce or enforces its security; or
- (d) a proposal for a voluntary arrangement shall have been made in relation to that Party under Part I Insolvency Act 1986; or
- (e) a step or event shall have been taken or arisen outside the United Kingdom which is similar or analogous to any of the steps or events listed at (a) to (d) above which shall include a Chapter XI filing in the USA; or
- (f) that Party takes any step outside the ordinary course of business (including starting negotiations) with a view to readjustment, rescheduling or deferral of any part of that Party's indebtedness, or proposes or makes any general assignment, composition or arrangement with or for the benefit of all or some of that Party's creditors or makes or suspends or threatens to suspend making payments to all or some of that Party's creditors or the Party submits to any type of voluntary arrangement; or
- (g) where that Party is resident in the United Kingdom it is deemed to be unable to pay its debts within the meaning of Section 123 Insolvency Act 1986.

- 1.1.28 "Know How" technical and other information which is not in the public domain, including information comprising or relating to concepts, discoveries, data, designs, formulae, ideas, inventions, methods, models, assays, research plans, procedures, designs for experiments and tests and results of experimentation and testing (including results of research or development), processes (including manufacturing processes, specifications and techniques), laboratory records, chemical, pharmacological, toxicological, clinical, analytical and quality control data, trial data, case report forms, data analyses, reports, manufacturing data or summaries and information contained in submissions to and information from ethical committees and regulatory authorities. Know How includes Documents containing Know How and shall be deemed to include any rights of action or intellectual property rights protecting such Know How. The fact that an item is known to the public shall not be taken to preclude the possibility that a compilation including the item, and/or a development relating to the item, is not known to the public.
- 1.1.29 "Legal Requirement" any present or future law, regulation, directive, instruction, direction or rule of any Competent Authority or Regulatory Authority including any amendment, extension or replacement thereof which is from time to time in force.
- 1.1.30 "Licensed IP" Licensed Patent Rights, Licensed Know How and Licensed Materials.
- 1.1.31 "Licensed Know How" any and all Know How owned by Sankyo relating to the Candidate.
- 1.1.32 "Licensed Material" any and all Material owned by Sankyo directly related to the Candidate as set out in Schedule 2 as of the Commencement Date unless otherwise specified therein.
- 1.1.33 "Licensed Patent Rights" any Patent Rights owned by Sankyo relating to inventions comprised in the Licensed Material or Licensed Know How, the Patent Rights listed in Schedule 3 and any subsequent Patent Rights owned by Sankyo directly relating to the inventions of CS-682 comprised in the Patent Rights to be added to Schedule 3 in accordance with Clause 6.8. For avoidance of doubt, Licensed Patent Rights exclude the CNDAC Patent Rights.
- 1.1.34 "Material" any chemical or biological substances including any: -
 - (a) organic or inorganic element or compound;

- (b) nucleotide or nucleotide sequence including DNA and RNA sequence;
- (c) gene;
- (d) vector or construct including plasmids, phages or viruses;
- (e) host organism including bacteria, fungi, algae, protozoa and hybridomas;
- (f) eukaryotic or prokaryotic cell line or expression system or any development strain or product of that cell line or expression system;
- (g) protein including any peptide or amino acid sequence, enzyme, antibody or protein conferring target properties and any fragment of a protein or a peptide enzyme or antibody;
- (h) drug or pro-drug;
- (i) assay or reagent; or
- (j) any other genetic or biologic material or microorganism.
- 1.1.35 "Marketing Authorisation" any and all consents or other authorisations or approvals required from a Regulatory Authority to market and sell a Product in any country, but not any form of pricing or reimbursement approval.
- 1.1.36 "Net Sales" with respect to Product in relation to which Marketing Authorisation and, if necessary in such country in the Territory, pricing approval has been granted by the appropriate Regulatory Authority or other Competent Authority for any jurisdiction for at least one indication, the gross total amount originally invoiced by Cyclacel, its Affiliates or Cyclacel Licensees to Third Parties for Products in the Territory less:
 - (a) quantity, trade and/or cash discounts actually granted;
 - (b) amounts repaid or credited and allowances including cash, credit or free goods allowances given by reason of chargebacks, retroactive price reductions (provided that price adjustments may be retrospectively made for up to two (2) consecutive fiscal years) or billing errors and rebates (including government-mandated rebates), actually allowed or paid;

- (c) amounts refunded or credited for Product which was rejected, spoiled, damaged, outdated or returned;
- (d) freight, shipment and insurance costs incurred transporting Product to a Third Party purchaser;
- (e) taxes, tariffs, customs duties and surcharges and other governmental charges incurred in connection with the sale, exportation or importation of Product.
- (f) bad debt losses and costs for the recovery of such bad debts.

The transfer of Product by Cyclacel or one of its Affiliates to another Affiliate or Cyclacel Licensee shall not be considered a sale. In such cases Net Sales shall be determined based on the invoiced sale price by the Affiliate or Cyclacel Licensee to the first Third Party trade purchaser, less the deductions allowed under this Clause.

Upon the sale or other disposal of Product other than in a bona fide arms length transaction exclusively for money or upon any use of Product for the purposes which do not result in a disposal of that Product in consideration of sales revenue customary in the country of sale, such other sale, disposal or use shall be deemed to constitute a sale at the relevant open market price in that country in which the sale, other disposal or use occurs, or, if that price is not ascertainable, a reasonable price assessed on an arm's length basis for the goods or services provided in exchange for the supply. Disposal of Product for, or use of Product in Clinical Trials for the purpose of obtaining a Marketing Authorisation or pre-clinical trials or as free samples in quantities common in the industry for this sort of Product shall not give rise to any deemed sale under this Clause.

Such amounts shall be determined from the books and records of Cyclacel maintained in accordance with GAAP (generally accepted accounting principles), consistently applied.

For the avoidance of doubt it is declared and agreed that if the product can be marketed and sold without pricing approval nothing in this definition requires pricing approval for it to be classified as Product for the purposes of this Agreement.

1.1.37 "Party or Parties" — Cyclacel or Sankyo or Cyclacel and Sankyo, as the case may be.

- 1.1.38 "Patent Rights" patent applications and patents, author certificates, inventor certificates or certifications of invention, applications for certificates of invention, utility certificates, improvement patents and models and certificates of addition and all foreign counterparts of them, including any divisions or divisional applications and patents, refilings, renewals, continuations, continuations-in-part, patents of addition, extensions (including patent term extensions), reissues, substitutions, confirmations, registrations, revalidations, request for continued examination, re-examinations or renewals thereof, pipeline and administrative protections and additions, and any equivalents of the foregoing in any and all countries of or to any of them including foreign counterpart applications, as well as any supplementary protection certificates and equivalent protection rights in respect of any of them.
- 1.1.39 "Phase I Clinical Trial" a human clinical trial normally conducted in healthy volunteers or patients with the aim of establishing the pharmacokinetic, pharmacodynamic and early safety profile of a product (alone or in combination with another agent).
- 1.1.40 "Phase II Clinical Trial" a human clinical trial where a product is tested in a number of patients for the purpose of establishing preliminary data on the efficacy and safety of a product.
- 1.1.41 "Phase III Clinical Trial" a human clinical trial conducted in a sufficient number of patients to establish safety or efficacy for one or more indication(s) tested and required for the filing of a submission to obtain Marketing Authorisation.
- 1.1.42 "Product" a pharmaceutical product comprising the Candidate or otherwise falling within a Valid Claim of the Licensed Patent Rights.
- 1.1.43 "Professors" [***]
- 1.1.44 "Quarter" each period of three months ending on 31 March, 30 June, 30 September or 31 December and "Quarterly" shall be construed accordingly.
- 1.1.45 "Recipient Party" the Party which receives Confidential Information from the other Party.
- 1.1.46 "Regulatory Authority" shall mean any national, supranational, regional, state or local regulatory agency, department, bureau, commission, council or other governmental entity including the FDA and EMEA, in each country involved in the granting of Marketing Authorisation for a Product.

- 1.1.47 "Results" Know How, Patent Rights and Materials arising from any research, development or commercialisation activity undertaken by or on behalf of Cyclacel in connection with this Agreement.
- 1.1.48 "Successful Completion" shall mean that the Phase II Clinical Trial in question has met the end points set out in Schedule 5.
- 1.1.49 "Territory" the World excluding the Excluded Territories.
- 1.1.50 "Third Party" any entity or individual other than Cyclacel or Sankyo or their respective Affiliates.
- 1.1.51 "Valid Claim" a claim of an issued and unexpired patent included within Patent Rights, which has not been permanently revoked, held unenforceable or invalid by a decision of a court or other governmental agency of competent jurisdiction, un-appealable or un-appealed within the time allowed for appeal, and which has not been admitted to be invalid or unenforceable through reissue or disclaimer or otherwise.
- 1.1.52 "Year" twelve (12) months commencing on 1 January and ending on 31 December.

1.2 In this Agreement:

- 1.2.1 Unless the context otherwise requires all references to a particular Clause, Schedule or paragraph shall be a reference to that Clause, Schedule or paragraph, in or to this Agreement as it may be amended from time to time pursuant to this Agreement;
- 1.2.2 The table of contents and headings are inserted for convenience only and shall be ignored in construing this Agreement;
- 1.2.3 Unless the contrary intention appears words importing the masculine gender shall include the feminine and vice versa and words in the singular include the plural and vice versa;
- 1.2.4 Unless the contrary intention appears words denoting persons shall include any individual, partnership, company, corporation, joint venture, trust, association (incorporated or not incorporated), organisation or other entity, in each case whether or not having legal personality;
- 1.2.5 Reference to any statute, directive or regulation includes any modification or re-enactment of that statute or regulation; and
- 1.2.6 References to the word "include" or "including" are to be construed without limitation to the generality of the preceding words.

2. LICENCES

- 2.1 Sankyo hereby grants to Cyclacel
 - 2.1.1 an exclusive licence under the Licensed IP.
 - 2.1.2 a non-exclusive, royalty-free terminable sub-licence under the CNDAC Patent Rights which are licensed with a right to sub-license by the Professors to Sankyo.

in each case to research, develop, make, have made, use, import, have imported, market, have marketed, sell, offer for sale, have sold or otherwise dispose of Candidate (or any constituent part of the same) and Product, for all indications in the Territory.

- 2.2 The Parties [***] which requires [***] with Product in the [***] and thus [***]. Sankyo shall use its reasonable efforts [***].
 - 2.2.1 [***] relation to the Candidate, Sankyo shall so notify Cyclacel in writing [***] shall be [***] under the same commercial terms as set out in this Agreement.
 - 2.2.2 [***] in relation to the Candidate, Sankyo shall so notify [***] discuss in good faith issues relating to [***] including but not limited to terms for the usage of data generated by Cyclacel [***].
 - 2.2.3 Sankyo shall not grant any rights or licences in the Licensed IP or CNDAC Patent Rights to any Third Party [***], or itself carry out any activities utilising the Licensed IP or Sub-Licensed IP [***] which would or would be reasonably expected to have an adverse impact upon the value or scope of the rights granted to Cyclacel hereunder. [***].
- 2.3 The licences set out in Clause 2.1 shall include the right to grant sub-licences. Save as otherwise provided in this Agreement, Cyclacel shall procure that Cyclacel Licensees are bound by the same obligations, to the extent practicable, as those hereunder, including, but not limited to the obligations of confidentiality and non-use for unauthorised purpose. Cyclacel shall be responsible to Sankyo for the acts and omissions, including breach of this Agreement, of any Cyclacel Licensee provided that in the case of a Cyclacel Licensee committing a material breach that would constitute a material breach of this Agreement, Cyclacel shall have sixty (60) days to require such Cyclacel Licensee to remedy the breach or to terminate its sub-licence before Cyclacel shall be considered to have committed a material breach under Clause 9.3.1. Cyclacel shall give Sankyo prompt notification of the identity of each Cyclacel Licensee with whom it concludes a sub-licence.

2.4 As soon as reasonably practicable following the Commencement Date and subject to the provisions of Clause 3.2.2 Sankyo shall disclose, supply copies of, or otherwise make available to Cyclacel the Licensed Materials and Documents containing the Licensed Know How. In particular, and without limiting the foregoing, Sankyo shall supply Cyclacel with quantities of the Candidate and CNDAC as listed in Schedule 2 and all synthesis data and methods relating to the Candidate. Cyclacel acknowledges that (i) save as set out in Clause 7.2 Sankyo does not warrant the quality of any of the Licensed Materials, and (ii) Sankyo has no further obligation to supply the Candidate and CNDAC in addition to the quantities listed in Schedule 2. Unless Cyclacel requests transfer of any or all of those capsules in writing after the Commencement Date, the Parties shall execute a separate entrustment agreement with respect to the maintenance and analysis of capsules on stability in Schedule 2, which sets forth that (i) Cyclacel shall pay the expenses for entrustment to Sankyo and such expenses shall not be subject to Clause 3.3.2, and that (ii) Sankyo shall perform the analysis after the expiration dates for storage periods of such capsules on stability. Cyclacel acknowledges that Sankyo would never be able to further extend expiration dates of such capsules according to Sankyo's standards. Sankyo will transfer ownership of all formulated Candidate as listed in Schedule 2 and the current drug master file of the Candidate to Cyclacel except for certain amount retained by Sankyo necessary for the technology transfer or other purposes, including, but not limited to, to comply with regulatory requirements. Ownership, title and risk of all Licensed Materials provided under this Clause 2.4 shall be transferred from Sankyo to Cyclacel (i) at shipment if shipped by Sankyo or its designee to Cyclacel, (ii) upon transfer of [***] in accordance with Clause 7.4 or (iii) if Licensed Materials otherwise become available to Cyclacel. Sankyo shall provide Cyclacel with all reasonable assistance requested by Cyclacel in interpreting and understanding the Licensed IP and in the transfer, including updating to the FDA, of [***] in accordance with Clause 7.4 at Cyclacel's cost. Such assistance by Sankyo for the CMC technology transfer of synthesis and internal process checks shall be deemed to be completed if either of the following events occurs:

[***]

Sankyo shall maintain its copies of Documents containing the Licensed Know-How for the lesser of eight (8) years or until requested by Cyclacel under Clause 2.7 or otherwise agreed by the Parties. Should Sankyo intend to dispose of the Documents it shall so notify Cyclacel and Cyclacel will be entitled to take ownership of such Documents at Cyclacel's discretion and cost.

2.5 Sankyo shall at Cyclacel's request execute or procure the execution of all such deeds and documents as may be necessary or desirable to record any of the rights granted to Cyclacel under this Agreement with any patent registry.

- 2.6 Sankyo shall notify Cyclacel in the event that Sankyo generates any Patent Rights or Know How relating to the Candidate or relating to CNDAC which also relates to the Candidate and if Cyclacel so requests such Patent Rights or Know How shall then automatically become part of the Licensed IP and shall be licensed to Cyclacel on the same terms as set out herein.
- 2.7 Sankyo shall maintain those items of technical data listed in Schedule 6 and shall make the same available to Cyclacel for use in connection with the development or commercialisation of Product (including, without limitation, for the purposes of cross-referencing or inclusion in regulatory filings relating to Candidate or Product). Sankyo shall exert its reasonable efforts to make such data available to Cyclacel within thirty (30) Business Days of the Commencement Date or the date which such data become available to Sankyo.
- 2.8 Sankyo shall promptly inform Cyclacel of any significant developments of which it becomes aware concerning any part of the Licensed IP or CNDAC Patent Rights which are licensed by Sankyo to Cyclacel.

3. PAYMENTS

- 3.1 In consideration of the rights granted to Cyclacel under this Agreement Cyclacel shall pay to Sankyo payments set out in this Clause 3.
- 3.2 Cyclacel shall pay to Sankyo a licence fee and costs comprised of the following:
 - 3.2.1 [***] payable within thirty (30) days of the Commencement Date;
 - 3.2.2 Sankyo's reasonable costs incurred pursuant to Sankyo's technology transfer obligations under Clause 2.4 within 30 days of receipt of Sankyo's Quarterly invoice and those costs incurred to Sankyo and reimbursed by Cyclacel under Clause 6.1 provided that in the event such amounts exceed [***] any amounts in excess of [***] shall be creditable against the milestone payments set out in Clause 3.3.1 and provided further that such costs in excess of [***] shall be subject to an overall maximum of [***]. The Parties acknowledge that as of the end of June 2003 Sankyo has incurred [***] in transferring Licensed Know How and Licensed Materials to Cyclacel and that Sankyo has invoiced Cyclacel for such amount. Notwithstanding the generality of the foregoing, Cyclacel agrees to pay to Sankyo such [***] within thirty (30) days of the Commencement Date.
- 3.3 Cyclacel shall pay to Sankyo the following non-refundable payments, provided, the milestone payments set out in Clauses 3.3.1 to 3.3.5 shall be payable in relation to the first Product only.
 - 3.3.1 subject to any set-off, if applied pursuant to Clause 3.2, [***] within thirty (30) days of the Successful Completion of the first Phase II Clinical Trial for Product;

- 3.3.2 [***] within thirty (30) days of the commencement of the first Phase III Clinical Trial for Product, provided, however: (a) if Cyclacel commenced the first Phase III Clinical Trial for Product without the Successful Completion of the first Phase II Clinical Trial for Product, the payment under Clause 3.3.1 above shall become also payable to Sankyo in addition to the payment under this Clause 3.3.2 and; (b) in the event that Cyclacel is granted Marketing Authorisation for Product in the USA, any country of the European Union or Japan on the basis of any clinical trial results generated other than pursuant to a formal Phase III Clinical Trial, the milestone payment under this Clause 3.3.2 shall become due and payable by Cyclacel at the same time.
- 3.3.3 [***] within thirty (30) days of the granting of the first Marketing Authorisation and, if required in order to sell Product, pricing approval for Product in the USA;
- 3.3.4 [***] within thirty (30) days of the granting of the first Marketing Authorisation and, if required in order to sell Product, pricing approval for Product in any country of the European Union;
- 3.3.5 [***] within thirty (30) days of the granting of the first Marketing Authorisation and, if required in order to sell Product, pricing approval for Product in Japan;
- 3.3.6 royalties applicable on Net Sales as follows:
 - 3.3.6.1 [***] on annual Net Sales less than [***]
 - 3.3.6.2 [***] on annual Net Sales greater than or equal to [***] and less than [***]
 - 3.3.6.3 [***] on annual Net Sales greater than or equal to [***]
- 3.4 In the case of sales of Product by Cyclacel, its Affiliates or Cyclacel Licensees in countries where such sale does not fall within a Valid Claim of a Licensed Patent Right the royalty payable on Net Sales in such country shall be [***] regardless of the volume of Net Sales in such country.
- 3.5 In the event a Third Party launches a generic product which directly competes with the Product and does not infringe Licensed IP and the volume of sales of Product are less than or equal to [***] of the total volume of sales of Product plus all such competing generic products in such country as evidenced by data compiled by IMS Health of 1499 Post Road, Fairfield, CT 06430, USA or, in the event such data by IMS Health is not available or the Parties agree it is not appropriate to use such data in such country, a similar market research agency the royalty payments set out in Clauses 3.3.6.1 to 3.3.6.3 shall no longer be payable to Sankyo on Net Sales in such country.

- 3.6 [***]
- 3.7 The royalties payable by Cyclacel hereunder shall be payable for the following periods:
 - 3.7.1 In those countries where at the time of first commercial sale in such country the sale of Product falls within a Valid Claim of Licensed Patent Rights subsisting in that country, royalties shall be payable until the latter of (i) the last to expire of such Licensed Patent Rights in that country or (ii) the tenth (10th) anniversary of the date of first commercial sale in that country;
 - 3.7.2 In those countries where at the time of first commercial sale of Product in that country the sale of Product does not fall within a Valid Claim of Licensed Patent Rights subsisting in that country, royalties shall be payable until the tenth (10th) anniversary of the date of first commercial sale in that country.
- 3.8 Disposal of reasonable quantities of Product for, or use of Product, in Clinical Trials performed for the purpose of obtaining a Marketing Authorisation or pre-clinical trials, or as free samples to be in quantities common in the industry for this sort of Product shall not give rise to liability for payment of a royalty under this Clause 3.
- 3.9 In the event that Product is sold as a Combination Product containing a therapeutically active ingredient that is not Licensed IP or CNDAC Patent Rights in a country, the Net Sales of the Product shall be determined by multiplying [***]. In the event Cyclacel wishes to develop a Combination Product which (1) is administered through a therapeutically active administration technology that is not Licensed IP or CNDAC Patent Rights, (2) is administered in accordance with a diagnostic detection technology that is not Licensed IP or CNDAC Patent Rights or (3) otherwise depends for a significant part of its value upon components which are not Licensed IP or CNDAC Patent Rights, then the Parties shall separately discuss how to determine the Net Sales applicable for such Combination Product. In the event the Parties fail to agree an applicable Net Sales price for such combination product within sixty (60) days of commencing such good faith discussions, such price shall be determined by a reputable independent auditor appointed by agreement between the Parties or, failing which, the President (or his or her nominee) for the time being of the Institute of Chartered Accountants of England and Wales (or any successor body thereto).

- 3.10 Cyclacel shall, and shall procure that its Affiliates and Cyclacel Licensees shall, keep true and accurate records and books of account containing all data necessary for the calculation of the amounts payable by it to Sankyo pursuant to this Agreement. Those records and books of account shall be kept for seven (7) years following the end of Year to which they relate and shall, upon reasonable notice having been given by Sankyo or its nominee, be open on Business Days for examination, under the terms of confidentiality contained in this Agreement, by an independent firm of accountants appointed by agreement between the Parties or between Cyclacel and Sankyo's nominee (as appropriate) or, failing such agreement within twenty-eight (28) days, by the President (or his or her nominee) for the time being of the Institute of Chartered Accountants of England and Wales (or any successor body thereto). Such examination shall take place not later than five (5) years following the expiration of the period to which it relates and there shall be no more than one such examination per year. Such independent accountant shall provide a written report of his findings to each of Cyclacel and Sankyo in which he certifies the moneys due to Sankyo in respect of the period under review. If such examination demonstrates that Cyclacel has underpaid royalties due to Sankyo, Cyclacel shall make a balancing payment within thirty (30) days. The cost of the examination, review and certification shall be the responsibility of Cyclacel if Cyclacel is shown to have underestimated the monies payable to Sankyo in any period being reviewed by more than [***] and the responsibility of Sankyo otherwise.
- 3.11 Cyclacel shall make the royalty payments due to Sankyo under this Clause 3 at Quarterly intervals. Within thirty (30) days of the end of each Quarter after first commercial sale in any country, Cyclacel shall prepare a statement which shall show on a Product by Product and a country by country basis for the previous Quarter all monies due to Sankyo under Clause 3 and shall, if applicable, contain Cyclacel's reasonable estimate of any amounts deductible pursuant to Clauses 3.4 and 3.5. That statement shall include details of the particular Product description including to the extent it requires use of Combination Product, sales of the Product and the royalty amount payable for each country in the Territory. Cyclacel shall pay the royalty due under Clause 3.3.6 within thirty (30) days of issuance of such statement. Within thirty (30) days of the end of the final Quarter of each Year Cyclacel shall carry out a reconciliation in relation to the royalty payments made and the royalty rate applied during the previous Year and, if applicable, shall apply the formula set out in Clause 3.6. If such reconciliation demonstrates that Cyclacel has underpaid royalties due to Sankyo, Sankyo shall make a payment to Cyclacel of a balancing refund within thirty (30) days.

- 3.12 If Sankyo give notice to Cyclacel within ten (10) Business Days of the receipt of any statement provided pursuant to Clause 3.11 that it does not accept it, that statement shall be reviewed and certified by an independent accountant appointed by agreement between the Parties or, in default of agreement within twenty eight (28) days, by the President (or his or her nominee) for the time being of the Institute of Chartered Accountants of England and Wales (or any successor body thereto). Cyclacel shall make available all books and records required for the purpose of that review and certification and the statements so certified shall be final and binding between the Parties. The cost of the examination, review and certification shall be the responsibility of Cyclacel if Cyclacel is shown to have underestimated the monies payable to Sankyo in any period being reviewed by more than [***] and the responsibility of Sankyo otherwise. Following any such certification under Clause 3.10 or this Clause 3.12 the Parties shall forthwith make any adjustments necessary in respect of the monies already paid to Sankyo in relation to the period in question.
- 3.13 Where Product is sold in a currency other than US dollars the rate of exchange to be used for converting such other currency into US dollars shall be the Telegraphic Transfer Middle (T.T.M.) rate at which such other currency can be sold for US dollars at the Bank of England in London at the close of business on the last working date for the period for which payment is to be made.
- 3.14 If any sums are unpaid by Cyclacel on the dates specified in this Clause then Sankyo shall be entitled to charge Cyclacel interest on the amount unpaid at the rate of [***] per annum above the prevailing Bank of England base lending rate from time to time until payment in full is made.
- 3.15 Royalties or other payments payable under this Clause shall be payable hereunder without any deduction or set-off save that if any law, regulation or treaty requires a withholding of income taxes on any milestone, royalty payments or other payments due to Sankyo, such taxes will be deducted for such payment and paid to the appropriate taxing authorities. Certificates of such tax payments shall be delivered to Sankyo along with the balance of the milestone, royalty payment or other payments.
- 3.16 All payments made to Sankyo under the Agreement shall be made by telegraphic transfer to the account of Sankyo Co. Ltd. at:

[***]

or any other bank account that may be notified by Sankyo to Cyclacel from time to time.

4. DEVELOPMENT AND COMMERCIALISATION

- 4.1 Cyclacel shall use commercially reasonable endeavours to pursue the development and commercialisation of Product. For the purposes of this Clause 4 such reasonable endeavours shall mean those efforts to develop and commercialise Product which are at least the same as the efforts used by Cyclacel in relation to the development and commercialisation of other Cyclacel pharmaceutical products under similar commercial circumstances that have similar commercial value and development status to the Product. Cyclacel shall carry out such development in accordance with the key elements of the Development Plan which plan may be updated and/or amended from time to time by Cyclacel and Cyclacel shall provide Sankyo with a copy of any such updated or amended plan.
- 4.2 Without limiting the generality of the foregoing Cyclacel shall use its reasonable endeavours [***]. If Cyclacel, its Affiliates or Cyclacel Licensees have not [***] then provided such failure is not due to an Exceptional Cause Sankyo shall be entitled to notify Cyclacel that it intends to terminate this Agreement, Cyclacel shall have sixty (60) days to reply to Sankyo's notification and demonstrate that such failure was due to an Exceptional Cause. In the event that the Parties agree that no such Exceptional Cause has been demonstrated then Sankyo will have the right to terminate this Agreement upon thirty (30) days written notice. [***]. The Parties shall, no later the fifth (5th) anniversary of the Commencement Date discuss, in good faith and review such time periods and, if appropriate, extend such period to [***].
- 4.3 Cyclacel shall within thirty (30) days of each anniversary of the Commencement Date provide Sankyo with a written summary of the progress which Cyclacel, its Affiliates and/or Cyclacel Licensees has made against the Development Plan in the previous twelve (12) month period. In addition, Cyclacel shall provide Sankyo within seven (7) months of each annual written report with a brief update of such progress, which update may be given in writing by facsimile or by e-mail.
- 4.4 In the event that Cyclacel decides to appoint a Third Party, or if Cyclacel receives an offer from a Third Party, in either case to develop and/or commercialise Product solely in Japan or in Japan as part of a multi-territory sub-licence, it shall so notify Sankyo which notification shall not be made prior to the first Successful Completion of a [***] Clinical Trial or the decision by Cyclacel or a Cyclacel Licensee to initiate [***] Clinical Trials and shall be accompanied by (i) details of the terms upon which Cyclacel is prepared to offer for the appointment of a Third Party licensee and (ii) the data and evidence on which Cyclacel or Cyclacel Licensee made such decision. Sankyo shall notify Cyclacel within [***] of receipt of the Cyclacel notification whether Sankyo wishes to develop and/or commercialise Product in Japan. If Sankyo notifies Cyclacel that it does wish to develop and/or commercialise Product in Japan the Parties shall meet and negotiate in good faith the terms of appointment of Sankyo as Cyclacel Licensee in Japan. If Sankyo notifies Cyclacel that it does not wish to develop and/or commercialise Product in Japan or, in the case where Sankyo has notified Cyclacel that it does wish to develop and/or commercialise Product in Japan but the Parties have failed to agree terms within [***] of commencing negotiations, then Cyclacel shall be free to appoint a Third Party to develop and/or commercialise Product in Japan provided that such appointment shall be on terms no more favourable than those terms which Cyclacel was prepared to offer Sankyo. For the avoidance of doubt if Cyclacel receives an offer at any time by a Third Party to commercialise Product on a worldwide basis, Cyclacel shall notify such Third Party that Sankyo has a first refusal right of Marketing in Japan and Cyclacel shall propose to such Third Party to accept such condition. In such case, the conditions imposed under this Clause will be accelerated and Cyclacel will serve notification to Sankyo who will respond in accordance with the provisions herein. [***].

5. INTELLECTUAL PROPERTY — OWNERSHIP

- 5.1 Sankyo owns the Licensed IP and during the period of this Agreement shall not assign, transfer, mortgage, charge or otherwise dispose of or encumber the same without the prior written consent of Cyclacel.
- 5.2 Any and all Results shall vest in and be owned by Cyclacel.

6. INTELLECTUAL PROPERTY — PROSECUTION, MAINTENANCE AND ENFORCEMENT

6.1 Sankyo shall at its own cost and expense file, prosecute and maintain Licensed Patent Rights, including for the avoidance of doubt all annuity and renewal fees and shall be responsible for the conduct of any claims or proceedings relating to the Licensed Patent Rights, including any interference or opposition proceedings. Sankyo shall keep Cyclacel and/or Cyclacel's nominated patent attorney fully and promptly informed of such filing, prosecution and maintenance of the Licensed Patent Rights including, without limitation, providing Cyclacel and/or Cyclacel's nominated patent attorney with copies of all correspondence with any patent office and shall give Cyclacel the right to comment thereon prior to any response being made to any such patent office. Sankyo shall ensure that copies of all such patent office correspondence provided to Cyclacel for comment shall be provided no less that sixty (60) days prior to the expiry of any patent office deadline for response (if applicable) and Cyclacel shall ensure that it provides Sankyo with its comments on such correspondence not less than thirty (30) days prior to the expiry of any such patent office deadline for response. Without limiting the generality of the foregoing obligation to keep Cyclacel fully and promptly informed, Cyclacel may itself or through its patent attorney make a written request to Sankyo or Sankyo's nominated patent attorney not more than once per Quarter for a progress report on the filing, prosecution and maintenance of the Licensed Patent Rights during the previous Quarter and a summary of the anticipated activities for the next Quarter. Notwithstanding this Clause 6.1, Cyclacel shall pay the costs and expenses of such filing, prosecution and maintenance upon Sankyo's Quarterly invoice prior to its payment of the first applicable milestone under Clauses 3.3.3, 3.3.4 or 3.3.5 and thereafter such costs shall be borne by Sankyo, except that such costs borne by Cyclacel after its payment of the Phase III entry milestone under 3.3.2, will be creditable against the first applicable milestone under Clauses 3.3.3, 3.3.4 or 3.3.5.

- 6.2 If Sankyo fails to prosecute or maintain the Licensed Patent Rights or any part thereof it must notify Cyclacel in writing at least thirty (30) days prior to the expiry of any material deadline for the taking of any steps in such action, proceeding or prosecution and thereafter Cyclacel may in its sole discretion prosecute or maintain the Licensed Patent Rights or any part thereof. If Cyclacel decides to prosecute or maintain the Licensed Patent Rights or any part thereof it shall do so at its own cost and expense and Sankyo shall promptly arrange for its patent attorneys to transfer to Cyclacel all relevant papers, files and other documents.
- 6.3 If either Party learns of any infringement or threatened infringement by a Third Party of Licensed IP then such Party shall promptly notify the other Party and shall provide such other Party with available evidence of such infringement.
- 6.4 In the event of an infringement of the Licensed IP Cyclacel shall have the first right to bring any action or proceedings in accordance with the following:
 - 6.4.1 Sankyo shall have the right to join as a co-plaintiff and be separately represented by counsel of its own choice and at its own reasonable cost and expense;
 - 6.4.2 Sankyo shall otherwise at Cyclacel's reasonable request and sole expense, provide Cyclacel with reasonable assistance in relation to such action or proceedings;
 - 6.4.3 if Cyclacel succeeds in any proceedings whether at trial or by way of settlement, Cyclacel shall be entitled to retain such award of costs and damages made in such proceedings or settlement sum after having reimbursed Sankyo for its costs and expenses incurred in assisting with the proceedings, subject to Clause 6.4.1 where Cyclacel and Sankyo shall be equally entitled to retain such award. If Cyclacel should lose such proceedings then Cyclacel shall on demand reimburse Sankyo its costs and expenses incurred in assisting with the proceedings;
 - 6.4.4 if Cyclacel fails to take any such proceedings Sankyo may give Cyclacel notice requiring Cyclacel to take such proceedings jointly with Sankyo within thirty (30) days of the date of the notice and if Cyclacel fails to do so Sankyo shall be entitled to do so at its own cost and expense and Cyclacel shall provide all necessary assistance to Sankyo in relation to such proceedings including lending its name to such proceedings as a co-plaintiff. If Cyclacel does not join with Sankyo in taking part in such proceedings Cyclacel shall not be entitled to be separately represented in such circumstances and Sankyo shall have sole conduct of such proceedings including the right to settle them. If Sankyo succeeds in any such proceedings it shall be entitled to retain the whole of any award of costs and damages made or settlement sum paid. If Cyclacel joins with Sankyo in taking part in such joint proceedings Cyclacel shall be entitled to be separately represented in such circumstances and Sankyo shall jointly conduct such proceedings with Cyclacel including the right to settle them. If Sankyo and Cyclacel succeed in any such proceedings they shall share equally any award of costs and damages made or settlement sum paid.

- 6.5 If during the term of this Agreement either Party receives any notice, claim or proceedings from any Third Party alleging infringement of that Third Party's intellectual property or know how as a result of either Party's activities in relation to this Agreement or use and exploitation of the Licensed IP the Party receiving that notice shall:
 - 6.5.1 forthwith notify the other Party of such notice, claim or proceedings;
 - 6.5.2 make no admission of liability without the prior consent of the other Party;
 - 6.5.3 subject to Clause 6.6.1 below Cyclacel shall conduct the defence of such claims or proceedings, including the right to settle them which shall include taking a licence from such Third Party, and Sankyo shall at Cyclacel's cost and expense co-operate with Cyclacel and its legal counsel and be available at Cyclacel's reasonable request to assist in such proceedings. Cyclacel shall keep Sankyo and its legal counsel fully and promptly informed at all times as to the status of Cyclacel's defence.
- 6.6 Whilst Cyclacel conducts the defence of such claim or proceedings:
 - 6.6.1 Cyclacel shall be responsible for and shall have conduct of such claims or proceedings but shall fully consult with Sankyo in defending or settling such claim or proceedings and Cyclacel shall not consent to any settlement, order or judgment requiring Sankyo to pay costs, damages or provide other relief without the written consent of Sankyo;
 - 6.6.2 if any such proceedings are settled by way of licence, then to the extent any royalties payable to a Third Party under such licence relate to Blocking IP they shall be deducted from the royalties payable by Cyclacel to Sankyo to the same extent and in the same manner as set out in Clause 3.6.

- 6.7 At Cyclacel's request and expense Sankyo shall promptly take all necessary steps to facilitate any application for a supplementary protection certificate extension of term or its equivalent, for any of the Licensed Patent Rights.
- 6.8 Within sixty (60) days after each Year-end, Sankyo shall provide Cyclacel with a report describing the status of the Licensed Patent Rights and Cyclacel shall provide Sankyo with a report describing the status of the any Patent Rights owned by Cyclacel under Clause 5.2. Such reports shall include, at a minimum, the patent country, patent and application numbers, filing date, issue date, expiration date and any other relevant information, and shall be used to update Schedule 3.

Sankyo's report shall be sent to: Cyclacel Limited

Dundee Technopole James Lindsay Place Dundee DD1 5JJ

UK

Attention: Chief Executive Officer Telephone: +44 1382 206062 Facsimile: +44 1382 206067

Cyclacel's report shall be sent to: Sankyo Company, Limited

[***]

7. WARRANTIES AND LIABILITY

7.1 Each Party represents and warrants to the other Party that:

- 7.1.1 it has the corporate power and authority and the legal right to enter into this Agreement and that this Agreement is a legal and valid obligation binding upon such Party and enforceable in accordance with its terms. The execution, delivery and performance of the Agreement by such Party does not conflict with any agreement, instrument or understanding, oral or written, to which it is or by which it is bound, nor violate any law or regulation of any court, governmental body or administrative or other agency having jurisdiction over it;
- 7.1.2 it has not, and during the term of the Agreement will not, without the prior written consent of the other Party grant any rights to any Third Party that would conflict with the rights granted to the other Party hereunder;
- 7.1.3 it has the right to grant the licenses granted or to be granted herein;
- 7.1.4 it is a corporation duly organised, validly existing and in good standing under the laws of the jurisdiction in which it is incorporated; and
- 7.1.5 the execution and delivery of this Agreement and the performance of such Party's obligations do not constitute a default or require any consent under any other contractual obligation of such Party.

- 7.2 Sankyo hereby warrants and undertakes that at the Commencement Date:
 - 7.2.1 it owns absolutely, co-owns or has the right to licence the Licensed IP and the CNDAC Patent Rights sufficient to grant the licences granted herein;
 - 7.2.2 it has obtained from each and every inventor of the Licensed Patent Rights an assignment of all rights such inventors may have in the Licensed Patent Rights and save as otherwise disclosed to Cyclacel in writing it has not partially assigned, licensed, mortgaged, charged or otherwise disposed of or encumbered its right, title or interest in the same;
 - 7.2.3 it has disclosed to Cyclacel all information relating to the Licensed IP and CNDAC Patent Rights and any therapeutic use of the Candidate which has been generated by Sankyo or Third Parties, provided that in the case of Third Parties the disclosure of such information is not prohibited under any confidentiality obligations; and
 - 7.2.4 it has disclosed to Cyclacel the identity of the Third Parties which have generated the information described in Clause 7.2.3; and
 - 7.2.5 it has not granted to any other party any licence to research, develop or commercialise the Candidate or CNDAC or any other compound falling within a Valid Claim of the Licensed Patent Rights or the CNDAC Patent Rights.
 - 7.2.6 it has manufactured the "samples of any capsules in storage", which is identified in Schedule 2, under GMP and it had obtained from [***] a statement that [***] has stored the same under GMP.
- 7.3 Save as is expressly stated in Clause 7.1 or 7.2 NO REPRESENTATION, CONDITION OR WARRANTY WHATSOEVER IS MADE OR GIVEN, EITHER EXPRESSED OR IMPLIED, BY OR ON BEHALF OF CYCLACEL OR SANKYO. THERE ARE NO EXPRESS OR IMPLIED WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, OR THAT THE USE OF THE CANDIDATE WILL NOT INFRINGE ANY PATENT, COPYRIGHT, TRADEMARK, OR OTHER PROPRIETARY RIGHTS. ALL CONDITIONS AND WARRANTIES WHETHER ARISING BY OPERATION OF LAW OR OTHERWISE ARE HEREBY EXPRESSLY EXCLUDED INCLUDING ANY CONDITIONS AND WARRANTIES TO THE EFFECT THAT ANY OF THE LICENSED IP IS VALID OR ENFORCEABLE.

- 7.4 Cyclacel shall have the control of and be responsible for all Clinical Trials conducted in relation to Product after execution of this Agreement including such ongoing Clinical Trials described in Schedule 7 and shall be the sponsor of such trials and in such capacity, shall, notwithstanding its indemnity rights under Clause 7.5, be responsible for the initial payment of any compensation due to any participants in such trials who suffer death or bodily injury pursuant to any legal rights or applicable industry guidelines. If it has not already done so prior to the Commencement Date, Sankyo shall submit to the FDA an Annual Update, including Information Amendment, to [***] detailing an extension to the shelf-life of Product and shall transfer [***] for such ongoing Clinical Trial to Cyclacel. In the event that the FDA does not accept an extension to the shelf life of Product for use in such ongoing Clinical Trials Cyclacel's obligations hereunder in relation to such ongoing Clinical Trials shall be suspended until such time as the FDA has accepted the use of Product in such trials. Cyclacel (i) shall continue to conduct such ongoing Clinical Trial after the Commencement Date in compliance with the Phase I protocol provided by Sankyo, (ii) shall not amend or alter such protocol and (iii) shall not terminate such ongoing Clinical Trial at least until the end of December 2003 for ethical reasons unless otherwise terminated in accordance with the protocol.
- 7.5 Subject to the provisions of Clause 7.6 Sankyo shall be responsible for and shall indemnify Cyclacel and its directors, officers, servants and agents (collectively "the Indemnified Party") against any and all liability, loss, damage, cost and expense (including legal costs) incurred or suffered by the Indemnified Party as a result of a breach of warranty by Sankyo under Clauses 7.1 or 7.2 or as a result of Sankyo's activities in relation to the development of Candidate or Product prior to the Commencement Date of this Agreement. An Indemnified Party that intends to claim indemnification under this Clause 7.5 shall promptly notify Sankyo of any Third Party claim in respect of which the Indemnified Party intends to claim that indemnification. The Indemnified Party shall not compromise or settle the claim prior to any such notice. Sankyo may assume and control the defence of any such Third Party claim, provided however, that an Indemnified Party shall have the right to retain its own counsel at its own cost and expense, if representation of that Indemnified Party by the counsel retained by Sankyo would be inappropriate due to actual or potential differing interests between the Indemnified Party and any other party represented by that counsel in the proceedings. The Indemnified Party shall co-operate with Sankyo and its legal representatives in the investigation of any matter covered by this indemnification.

- 7.6 Cyclacel shall be responsible for and shall indemnify Sankyo and its Affiliates, directors, officers, servants and agents (collectively "the Indemnified Party") against any and all liability, loss, damage, cost and expense (including legal costs) incurred or suffered by the Indemnified Party as a result of any claim brought against Sankyo or its Affiliates by a Third Party which arises
 - 7.6.1 as a result of the activities by Cyclacel or its Affiliates, Cyclacel Licensees, agents or distributors under this Agreement in relation to the development or commercialisation of the Product being a claim that use of any Product has caused death or bodily injury; or
 - 7.6.2 as a result of a breach of warranty by Cyclacel under Clause 7.1.

An Indemnified Party that intends to claim indemnification under this Clause 7.6 shall promptly notify Cyclacel of any Third Party claim in respect of which the Indemnified Party intends to claim the indemnification. The Indemnified Party shall not compromise or settle the claim prior to any such notice. Cyclacel may assume and control the defence of any such Third Party claim, provided however, that an Indemnified Party shall have the right to retain its own counsel at its own cost and expense, if representation of that Indemnified Party by the counsel retained by Cyclacel would be inappropriate due to actual or potential differing interests between the Indemnified Party and any other party represented by that counsel in the proceedings. The Indemnified Party shall co-operate with Cyclacel and its legal representatives in the investigation of any matter covered by this indemnification.

- 7.7 Subject to the indemnities in Clauses 7.5 and 7.6 neither Party shall be liable to the other in contract, tort, negligence, breach of statutory duty or otherwise for any loss, damage, costs or expenses of any nature whatsoever incurred or suffered by the other or its Affiliates:
 - 7.7.1 of a direct nature where the same is a loss of turnover, profits, business or goodwill; or
 - 7.7.2 an indirect or consequential nature including any indirect or consequential economic loss or other indirect or consequential loss of turnover, profits, business or goodwill.

8. CONFIDENTIALITY

- 8.1 Subject to Clause 8.6, each of the Parties undertakes and agrees to:
 - 8.1.1 only use the Confidential Information for the purposes envisaged under this Agreement and not to use the same for any other purpose whatsoever;
 - 8.1.2 ensure that only those of their officers, consultants, employees (including without limitation directors), sublicensees, Affiliates and such third parties who are directly concerned with the carrying out of this Agreement have access to the Confidential Information on a strictly applied "need to know" basis and are informed of the secret and confidential nature of it;

- 8.1.3 keep the Confidential Information secret and confidential and not directly or indirectly to disclose or permit to be disclosed, make available or permit to be made available the same to any Third Party for any reason without the prior written consent of the Disclosing Party (except as set out in the Development Plan and then under equivalent confidentiality provisions);
- 8.1.4 clearly identify the Confidential Information as confidential.
- 8.2 The obligations of confidentiality referred to in Clause 8.1 shall not extend to any Confidential Information which:
 - 8.2.1 is or becomes generally available to the public otherwise than by reason of breach by a Recipient Party of the provisions of that Clause; or
 - 8.2.2 is known to the Recipient Party and is at its free disposal prior to its receipt from the Disclosing Party in circumstances where the Recipient Party holds evidence that it has not been derived from access to the Disclosing Party's Confidential Information; or
 - 8.2.3 is mutually agreed in writing by the Parties to no longer be confidential; or
 - 8.2.4 is subsequently disclosed to the Recipient Party without obligations of confidentiality by a Third Party owing no such obligations to the Disclosing Party in respect of that Confidential Information; or
 - 8.2.5 is required by a Legal Requirement to be disclosed and then only (subject to Clause 8.3) when prompt written notice of this requirement has been given to the Disclosing Party so that it may, if so advised, seek appropriate relief to prevent such disclosure provided always that in such circumstances such disclosure shall be only to the extent so required and shall be subject to prior consultation with the Disclosing Party with a view to agreeing timing and content of such disclosure.
- 8.3 The requirement under Clause 8.2.5 to notify the Disclosing Party when Confidential Information is required to be disclosed pursuant to Legal Requirement shall not apply when such disclosure is required as part of any regulatory submission or approval process.

- 8.4 All Confidential Information disclosed by the Disclosing Party to the Recipient Party shall remain the property of the Disclosing Party subject to the use or disposal of Materials pursuant to the Development Plan. In the event that a court or Competent Authority assumes partial or complete control over the assets of a Recipient Party based on the insolvency or bankruptcy of that Party, the Recipient Party shall:
 - 8.4.1 promptly notify such court or Competent Authority:
 - (a) that Confidential Information received from the Disclosing Party under this Agreement remains the property of the Disclosing Party; and
 - (b) of the confidentiality obligations under this Agreement; and
 - 8.4.2 to the extent permitted by law, take all steps necessary or desirable to maintain the confidentiality and security of the Disclosing Party's Confidential Information and to ensure that the court or Competent Authority maintains that Confidential Information in confidence in accordance with this Agreement.
- 8.5 The Parties agree that the obligations of confidentiality set out in this Clause 8 shall continue to apply following the expiration or termination of this Agreement for so long as there is Confidential Information which is not the subject of Clause 8.2.
- 8.6 Sankyo acknowledges and agrees that Cyclacel shall be free to publish the results arising pursuant to the exercise of its rights and licences under this Agreement. Such publication may also include Licensed Know How comprising data related to the Candidate and the development thereof provided that Cyclacel shall give Sankyo prior written notice of any publication containing such Licensed Know How and Sankyo shall have thirty (30) days to comment upon the same. Cyclacel shall consider any comments from Sankyo in good faith. Cyclacel shall be free to publish the Licensed Know How relating to the Candidate and/or its development following such thirty (30) day period.
- 8.7 Results published under Clause 8.6, will be considered to be covered by Clause 8.2.3 in respect of future publications and Clause 8.6 will no longer apply.

9. TERM AND TERMINATION

- 9.1 This Agreement shall commence on the Commencement Date and shall continue in force until no payments are due hereunder or until termination under this Clause 9 whichever is the earlier.
- 9.2 Cyclacel may terminate this Agreement at any time upon six (6) months' (upon twelve (12) months' if after Product launch) written notice to Sankyo for technical, scientific, efficacy, safety or commercial reasons.

- 9.3 Cyclacel on the one hand and Sankyo on the other hand (the "Terminating Party") shall have the right to terminate this Agreement upon giving thirty (30) days' written notice of termination to the other (the "Defaulting Party") upon the occurrence of any of the following events at any time during this Agreement:
 - 9.3.1 the Defaulting Party committing a material breach of this Agreement which in the case of a breach capable of remedy shall not have been remedied within thirty (30) days, or sixty (60) days in case of breach by Cyclacel Licensee as permitted under Clause 2.3, of the receipt by it of a notice identifying the breach and requiring its remedy. The Parties acknowledge that non-payment of sums due by Cyclacel may amount to a material breach having regard to the level of non-payment and the reasons given by Cyclacel for such non-payment. The Parties agree that any notification of material breach pursuant to this Clause 9.3 shall be sent to the chief executive officer of the Defaulting Party;
 - 9.3.2 if an Insolvency Event occurs in relation to the Defaulting Party.

10. EFFECTS OF TERMINATION

- 10.1 Upon termination of this Agreement by Sankyo pursuant to Clause 4.2 or Clause 9.3 or by Cyclacel pursuant to Clause 9.2:
 - 10.1.1 the licences granted by Sankyo pursuant to Clause 2 under the Licensed IP, and CNDAC Patent Rights shall terminate;
 - 10.1.2 if Sankyo so requests in writing, Cyclacel shall immediately deliver up to Sankyo the Licensed IP, CNDAC Patent Rights and Documents containing or making any reference to the Licensed IP or CNDAC Patent Rights as soon as reasonably practicable. Within sixty (60) days of such termination, the Parties shall negotiate and agree a separate Termination Agreement in regard to the transfer of ownership and return to Sankyo of all regulatory submissions, and documents filed with a Regulatory Authority for the Product in the Territory. The Parties shall cooperate in order to provide the appropriate notice and documentation to the Regulatory Authority for each submission as required by applicable law; and
 - 10.1.3 if Sankyo wishes to obtain an exclusive licence to the Cyclacel Results it shall so notify Cyclacel and Cyclacel shall grant to Sankyo an exclusive, worldwide, sub-licensable licence to the Cyclacel Results the commercial terms of which shall be negotiated in good faith by the Parties, having regard to the (i) length of time this Agreement has been in force, (ii) the development efforts expended by Cyclacel, and (iii) the cause of termination. In the event the Parties fail to agree the terms of such licence within sixty (60) days of the date upon which Cyclacel receives written notice from Sankyo that it wishes to obtain such a licence, the matter shall be referred to Expert's Decision for resolution, whilst such exclusive license shall be temporarily granted to Sankyo during such period.

- 10.2 In no event shall termination of this Agreement:
 - (i) relieve the Parties of any liability arising out of events occurring before the effective date of such termination.
 - (ii) affect the continued operation or enforcement of any provision of this Agreement which by its express terms is to survive termination.
 - (iii) release either Party from the obligation to pay any amounts that become due on or before the effective date of termination.
- 10.3 Upon termination of this Agreement by Sankyo pursuant to Clause 9.3 and at the request of any Cyclacel Licensee Sankyo shall enter into a direct licensing arrangement with such Cyclacel Licensee on terms substantially similar to those contained herein save that any licence granted by Sankyo to such Cyclacel Licensee shall be consistent with the terms of the licence granted by Cyclacel in relation to field, territory, exclusivity/non-exclusivity, whether there is a right to sub-license, and payment provisions and provided that: (i) such sub-licensee is in good standing under and in compliance with all material terms and conditions of the sublicence agreement and this Agreement; (ii) should Sankyo's ongoing direct costs related to such sublicensee including but not limited to IP costs under Clause 6.1, not be fully reimbursed by revenue from such sublicensee on an annual basis then Sankyo can require that such sublicensee reimburse Sankyo's costs, such reimbursement to be credited against future revenue payable to Sankyo by such sublicensee; and (iii) Sankyo's obligations under such sublicense agreement are no greater than those under this Agreement.

11. ASSIGNMENT/SUB-CONTRACTING

11.1 Neither this Agreement nor any interest hereunder shall be assignable by either Cyclacel or by Sankyo without the written consent of the other, such consent not to be unreasonably withheld, provided however that any Party may assign this Agreement to any corporation with which it may merge or consolidate, or to which it may transfer all or substantially all of its assets to which this Agreement relates, subject to obtaining a direct deed of undertaking from such corporation addressed to the other Parties agreeing to be bound by all the terms of this Agreement.

12. FORCE MAJEURE

- 12.1 If a Party (the "Affected Party") is unable to carry out any of its obligations under this Agreement due to Force Majeure this Agreement shall remain in effect but the Affected Party's relevant obligations under this Agreement and the corresponding obligations of the other Party ("Non-Affected Party") under this Agreement, shall be suspended for a period equal to the circumstance of Force Majeure provided that:
 - 12.1.1 the suspension of performance is of no greater scope than is required by the Force Majeure;
 - 12.1.2 the Affected Party immediately gives the Non-Affected Party prompt written notice describing the circumstance of Force Majeure, including the nature of the occurrence and its expected duration, and continues to furnish regular reports during the period of Force Majeure and notifies the Non-Affected Party immediately of the cessation of the Force Majeure:
 - 12.1.3 the Affected Party uses all reasonable efforts to remedy its inability to perform and to mitigate the effects of the circumstance of Force Majeure; and
 - 12.1.4 a soon as practicable after the event which constitutes Force Majeure the Parties discuss how best to continue their operations as far as possible in accordance with this Agreement.

13. GOVERNING LAW

13.1 The validity, construction and interpretation of this Agreement and any determination of the performance which it requires shall be governed by the laws of England.

14. JURISDICTION

14.1 In the event of any material dispute concerning rights or obligations under this Agreement then the Parties shall comply with the following procedure: the Chief Executive Officer of Cyclacel and the President of Sankyo or its nominee shall be notified in writing of the dispute by either Party. The Chief Executive Officer of Cyclacel and the President of Sankyo or their nominees shall meet to resolve the dispute in good faith. If such resolution is not reached within sixty (60) days of such written notice, then the dispute shall be referred to the non-exclusive jurisdiction of the courts of England and Wales.

15. WAIVER

- 15.1 Save as expressly provided in this Agreement neither Party shall be deemed to have waived any of its rights or remedies whatsoever unless the waiver is made in writing, signed by a duly authorised representative of that Party and may be given subject to any conditions thought fit by the grantor. Unless otherwise expressly stated any waiver shall be effective only in the instance and for the purpose for which it is given.
- 15.2 No delay or failure of any Party in exercising or enforcing any of its rights or remedies whatsoever shall operate as a waiver of those rights or remedies or so as to preclude or impair the exercise or enforcement of those rights or remedies. No single or partial exercise or enforcement of any right or remedy by any Party shall preclude or impair any other or further exercise or enforcement of that right or remedy by that Party.

16. SEVERANCE OF TERMS

- 16.1 If the whole or any part of this Agreement is or becomes or is declared illegal, invalid or unenforceable in any jurisdiction for any reason (including both by reason of the provisions of any legislation and also by reason of any decision of any court or Competent Authority which either has jurisdiction over this Agreement or has jurisdiction over any of the Parties):
 - 16.1.1 in the case of the illegality, invalidity or un-enforceability of the whole of this Agreement it shall terminate in relation to the jurisdiction in question; or
 - 16.1.2 in the case of the illegality, invalidity or un-enforceability of part of this Agreement that part shall be severed from this Agreement in the jurisdiction in question and that illegality, invalidity or un-enforceability shall not in any way whatsoever prejudice or affect the remaining parts of this Agreement which shall continue in full force and effect provided that the said remaining parts continue to satisfy the commercial intentions of the Parties and provided that the remaining parts do constitute a substantial part of this Agreement.

17. ENTIRE AGREEMENT/VARIATIONS

- 17.1 This Agreement constitutes the entire agreement and understanding between the Parties and supersedes all prior oral or written understandings, arrangements, representations or agreements between them relating to the subject matter of this Agreement. The Parties acknowledge that no claims shall arise in respect of any understandings, arrangements, representations or agreements so superseded. No director, employee or agent of any Party is authorised to make any representation or warranty to another Party not contained in this Agreement, and each Party acknowledges that it has not relied on any such oral or written representations or warranties. Nothing in this Agreement removes or overrides any right of action by any Party in respect of any fraudulent misrepresentation, fraudulent concealment or other fraudulent action.
- 17.2 No variation, amendments, modification or supplement to this Agreement shall be valid unless agreed in writing in the English language and signed by a duly authorised representative of each Party.

18. NOTICES

- 18.1 Any notice or other communication given pursuant to or made under or in connection with the matters contemplated by this Agreement shall be in writing in the English language and shall be delivered by hand or by courier or shall be sent by post or recorded delivery to the address of the recipient set out in Schedule 8 or as specified by the recipient from time to time in accordance with Clause 18.3. Notices sent by fax or E-Mail shall not be valid of themselves and must be confirmed in hard copy form by hand or by recorded delivery.
- 18.2 Any notice given pursuant to this Clause shall be deemed to have been received:
 - 18.2.1 if delivered by hand or by courier, at the time of delivery; or
 - 18.2.2 if sent by recorded delivery, at the time of delivery.
- 18.3 A Party may notify the other Parties to this Agreement of a change of its name, relevant addressee, address or facsimile number for the purposes of
- 18.4 8 provided that such notification shall only be effective on:
 - 18.4.1 the date specified in the notification as the date on which the change is to take place; or
 - 18.4.2 if no date is specified or the date specified is less than five (5) clear Business Days after the date on which the notice is given, the date falling five (5) clear Business Days after notice of any such change has been given.
- 18.5 For the avoidance of doubt, the Parties agree that the provisions of this Clause shall not apply in relation to the service of Service Documents (as defined in Clause 18.5).
- 18.6 "Service Document" means a writ, summons, order, judgement or other document related to or in connection with any Court proceeding, cause, matter or action arising out of or connected in any way with this Agreement.

19. COUNTERPARTS

19.1 This Agreement may be executed in any number of counterparts and by the Parties on separate counterparts, each of which when so executed shall be an original of this Agreement, and all of which shall together constitute one and the same instrument. Complete sets of counterparts shall be lodged with each Party.

20. THIS AGREEMENT NOT TO CONSTITUTE A PARTNERSHIP

20.1 Nothing in this Agreement and no action taken by the Parties pursuant to this Agreement shall constitute or be deemed to constitute a partnership, association, joint venture or other co-operative entity between the Parties and neither Party shall have any authority to bind the other in any way except as provided in this Agreement.

21. COSTS

21.1 Each Party shall bear its own costs, legal fees and other expenses incurred in the negotiation, preparation, execution and implementation of this Agreement and the documents referred to herein unless otherwise set forth in this Agreement.

22. PUBLICITY

- 22.1 No public announcements or other disclosure to third parties concerning the financial or other terms of this Agreement shall be made, whether directly or indirectly, by either Party to this Agreement, except as may be legally required or as may be required for recording purposes, without first obtaining the approval of the other Party and agreement upon the nature and text of such announcement or disclosure, with the exception that:
 - 22.1.1 a Party may disclose the full terms of this Agreement to its investment bankers, lawyers, accountants and other professional advisors or a Third Party seeking to invest in, lend funds to acquire or merge with or be acquired by such Party without the other Party's prior approval provided that such disclosure is made under terms of confidentiality whether express or implied; and
 - 22.1.2 a Party may disclose the terms of this Agreement to any securities exchange or regulatory authority or government body to which either Party is subject or submits, wherever situated, including (without limitation) the US Securities Exchange Commission, the London Stock Exchange or the Panel on Take-overs and Mergers, if and to the extent required by the force of law provided that it takes advantage of all provisions to keep confidential as many terms of this Agreement as possible.

22.2 In respect of those public announcements and disclosures not permitted by Clause 22.1 the Party desiring to make any such public announcement or other disclosure shall inform the other Party of the proposed announcements or disclosure in reasonably sufficient time prior to public release, and shall provide the other Party with a written copy thereof, in order to allow such Party to comment upon such announcement or disclosure, which comments shall be provided by such other Party within five (5) Business Days. The Parties shall jointly develop press releases and information materials that can be used by either Party for presentations to financial advisers, the UK Stock Exchange, and similar recipients.

IN WITNESS WHEREOF the Parties have executed this Agreement as of the Commencement Date.

SIGNED by)	
for and on behalf of)	Takashi Shoda, President and Representative Director
SANKYO CO. LTD.)	
Date:)	
SIGNED by)	
for and on behalf of)	Spiro Rombotis, Chief Executive Officer
CYCLACEL LIMITED)	
Date:)	

SCHEDULE 1 DEVELOPMENT PLAN

SCHEDULE 2 LICENSED MATERIALS

SCHEDULE 3 LICENSED PATENT RIGHTS

Patent Rights owned by Sankyo

A. Licensed Patent Rights

[*] (Crystal of CS-682)

Publication No. WO 02/064609 A1 (22.08.2002)

Crystal of pyrimidine nucleoside derivative

Country	Appln. No.	Appln. Date	Patent No.	Patent Date	Expiry Date	
	-					
PCT	[*]	06/02/2002				
Australia	[*]					
Brazil	[*]					
Canada	[*]					
China	[*]					
Colombia	[*]					
Czech Republic	[*]					
EPC	[*]					
Austria, Belgium, Switzerland & Liechtenstein, Cyprus, Germany, Denmark, Spain, Finland, France, Great Britain, Greece, Ireland,						
	mbourg, Monaco, I	Netherlands, Portug	al, Sweden, Turke	ey,		
[*]	[*]					
Hungary	[*]					
[*]	[*]					
Israel	[*]					
India	[*]					
Korea	[*]					
Mexico	[*]					
[*]	[*]					
New Zealand	[*]					
[*]	[*]					
[*]	[*]					
Russia	[*]					
[*]	[*]					
[*]	[*]					
U. S. A.	[*]					
[*]	[*]					
[*]	[*]					
Japan	[*]	[*]				

B. Licensed Patent Rights

[*] (CS-682)

Pyrimidine nucleoside derivatives having anti-tumor activity, their preparation and use

Country	Appln. No.	Appln. Date	Patent No.	Patent Date	Expiry Date		
Australia	[*]	30/09/1992	654212	15/02/1995	[*]		
Brazil (pipeline)	[*]	13/05/1997	PI 1100621-8	07/12/1999	[*]		
Canada	[*]	29/09/1992					
China	[*]	30/09/1992	92113067.8	03/08/1996	[*]		
Czech Republic	[*]	30/09/1992	289477	29/11/2001	[*]		
EPC	92308904.9	30/09/1992	0536936	14/08/1996	[*]		
Austria, Belgium, Switzerland & Liechten	stein, Germany, D	enmark, Spain, Frai	nce, Great Britain,	Greece, Ireland, Ita	aly, Luxembourg,		
Monaco, Netherlands, Portugal, Sweden							
Finland	[*]	[*]	105556	15/09/2000	[*]		
Hong Kong	[*]		2042/96	14/08/1996	[*]		
Hungary	[*]	30/09/1992	[*]	[*]	[*]		
Hungary (pipeline)	[*]	[*]	211851	13/08/1996	[*]		
Indonesia	[*]	[*]	ID0002812	22/06/1998	[*]		
Israel	[*]	[*]	103301	14/10/1997	[*]		
Korea	[*]	[*]	255491	15/02/2000	[*]		
Mexico	[*]	[*]					
Norway	[*]	[*]	179675	27/11/1996	[*]		
New Zealand	[*]	[*]	244574	14/06/1994	[*]		
[*]	[*]	[*]					
Russian Fed.	[*]	[*]	2085557	27/07/1997	[*]		
Thailand	[*]	[*]	9368	[*]	[*]		
Taiwan	[*]	[*]	[*]	[*]	29/09/2012		
U. S. A.	[*]	[*]	[*]				
U. S. A.	[*]	[*]	5691319	25/11/1997	25/11/2014		
South Africa	[*]	[*]	[*]	[*]	[*]		
Japan	[*]	[*]	2569251	03/10/1996	[*]		

SCHEDULE 4 CNDAC PATENT RIGHTS

C. CNDAC Patent Rights

[*] 2002. 8.30

(CNDAC: Active form of CS-682) Pyrimidine nucleoside derivatives

Country	Appln. No.	Appln. Date	Patent No.	Patent Date	Expiry Date
[*]	[*]	[*]	[*]	[*]	[*]
[*]	[*]	[*]	[*]	[*]	[*]
		[*]			
[*]	[*]	[*]	[*]	[*]	[*]
[*]	[*]	[*]	[*]	[*]	[*]
[*]	[*]	[*]	[*]	[*]	[*]
[*]	[*]	[*]	[*]	[*]	[*]
[*]	[*]	[*]	[*]		
U. S. A.	[*]	[*]	5616567	01/04/1997	01/04/2014
U. S. A.	[*]	[*]	5654420	05/08/1997	05/08/2014
[*]	[*]	[*]	[*]	[*]	[*]

SCHEDULE 5 SUCCESSFUL COMPLETION CRITERIA

SCHEDULE 6 RETAINED TECHNICAL DATA

SCHEDULE 7 COMMITTED CLINICAL TRIALS

SCHEDULE 8 ADDRESSES FOR NOTICES

Cyclacel Chief Executive Officer Dundee Technopole James Lindsay Place Dundee DD1 5JJ

Tel: +44 1382 206062 Fax: +44 1382 206067

Sankyo Co., Ltd. Director, Licensing Department 5-1 Nihonbashi-honcho 3-chome, Chuo-ku, Tokyo 103-8426 JAPAN

Tel: [***] Fax: [***]

SCHEDULE 9 EXPERT'S DECISION

- In the event the Parties fail to agree upon an appropriate level of consideration pursuant to Clause 10.1.3, the matter shall be determined by an expert ("Expert") which Expert shall be suitably qualified to determine that particular matter and who shall be nominated jointly by the Parties or, failing agreement between the Parties within twenty (20) business days of a written request by either Party to the other seeking to initiate the Expert's decision procedure, either Party may request the International Chamber of Commerce (Paris) to nominate the Expert.
- 2 The Parties shall within fourteen (14) days of the appointment of the Expert file written submissions setting out their respective view on appropriate levels of consideration and appropriate accompanying documents.
- In determining the matter referred pursuant to Clause 10.1.3 the Expert shall take into account (i) the length of time this Agreement has been in force, (ii) the development efforts expended by Cyclacel; and (iii) the cause of termination of the Agreement.
- 4 In all cases the terms of appointment of the Expert by whomsoever appointed shall include:
 - 4.1 a commitment by the Parties to share equally the Expert's fee;
 - 4.2 a requirement on the Expert to act fairly as between the Parties and according to the principles of natural justice;
 - 4.3 a requirement on the Expert to hold professional indemnity insurance both then and for three years following the date of his determination;
 - 4.4 a commitment by the Parties to supply to the Expert all such assistance, documents and information as he or she may require for the purpose of his or her determination.
 - 4.5 a commitment by the Parties that all negotiations connected with the dispute shall be conducted in confidence and without prejudice to the rights of the Parties in any future proceedings.

- 5 The Expert's decision shall be final and binding on the Parties (save in the case of negligence or manifest error).
- The Parties expressly acknowledge and agree that they do not intend the reference to the Expert to constitute an arbitration within the scope of any arbitration legislation, the Expert's decision is not a quasi judicial procedure and the Parties shall have no right of appeal against the Expert's decision provided always that this shall not be construed as waiving any rights the Parties might have against the Expert for breaching his or her terms of appointment or otherwise being negligent.



Licensing Department SANKYO CO., LTD.

3-5-1, Nihonbashi Honcho, Chuo-ku, Tokyo, 103-8476, Japan

Phone: 81-3-5255-7084, Fax: 81-3-5255-7086

Spiro Rombotis Chief Executive Officer Cyclacel Limited Dundee Technopole James Lindsay Place Dundee, DD1 5JJ Scotland, United Kingdom

Re: Amendment to CS-682 License Agreement Between Sankyo Co., Ltd. and Cyclacel Limited

Dear Mr. Rombotis,

As per your request, this letter is to confirm that we have agreed to amend the CS-682 License Agreement as follows:

- 1. Clause 2.1.2 of CS-682 License Agreement is hereby amended by deleting the word "terminable" from that Clause. As amended, the said Clause shall read as follows:
 - 2.1.2 a non-exclusive, royalty-free sub-license under the CNDAC Patent Rights which are licensed with a right to sub-license by the Professors to Sankyo.
- 2. Cyclacel agrees to be responsible for and to indemnify Sankyo and its Affiliates, directors, officers, servants and agents (collectively "the Indemnified Party") against any and all liability, loss, damage, cost and expense (including legal costs) incurred or suffered by the Indemnified Party as a result of claim brought against Sankyo or its Affiliates by a Third Party which arises as a result of the activities by Cyclacel or its Affiliates, Cyclacel Licensees, agents or distributors under the CS-682 License Agreement in relation to the deletion of "terminable" from Clause 2.1.2 thereof.
- 3. Sankyo agrees that in the event that Cyclacel's ability to exercise the said sub-license is jeopardized or lost it will cooperate with Cyclacel, upon Cyclacel's request and at Cyclacel's expenses, in taking any steps which Sankyo considers reasonably required to preserve or restore the said sub-license and the rights granted under the CS-682 License Agreement.
- 4. Notwithstanding with the items above, Cyclacel hereby acknowledges that Sankyo does not amend its warranty as set forth in CS-682 License Agreement with respect to CNDAC Patent Rights.
- 5. This Amendment supersedes all prior oral or written understandings, arrangements, representations or agreements, if any, between the parties relating to the subject matter of this Amendment to the extent that they conflict with this Amendment.
- 6. The validity, construction and interpretation of this Amendment to CS-682 License Agreement and any determination of the performance which it requires shall be governed by the laws of England.

These agreements are made in consideration of the mutual covenants contained herein. Could you please indicate your agreement by countersigning at the space indicated below and returning this agreement to the undersigned.

Sincerely yours and agreed,

/s/ Akira Morita

Akira Morita
Corporate Officer,
Director, Licensing Department

Date: April 28, 2004

AGREED:

/s/ Spiro Rombotis

Spiro Rombotis Chief Executive Officer Date: 28 April 2004



Licensing Department SANKYO CO., LTD.

3-5-1, Nihonbashi Honcho, Chuo-ku, Tokyo, 103-8476, Japan

Phone: 81-3-5255-7084, Fax: 81-3-5255-7086

1st April, 2004

Tel: +44 1382 206062

Fax: +44 1382 206067

Dr. Spiro Rombotis Chief Executive Officer Cylacel Dundee Technopole James Lindsay Place Dundee DD1 5JJ U K

Re: Territory of the CS-682 Licensing Agreement

Dear Dr. Rombotis;

I am happy to inform you that, in accordance with Clause 2.2.1 of the CS-682 License Agreement dated 10th September 2003 (the "Agreement") the license set out in Clause 2.1 of the Agreement is hereby extended to the "Excluded Territories" under the same commercial terms as set out in the Agreement.

[***]

This letter agreement is to amend the Agreement in accordance with Clause 17.2 thereof. Please sign two copies of this letter agreement and send one copy to us for our record.

Sincerely yours,

/s/ Akira Morita
Akira Morita
Corporate Officer,
Director, Licensing Department

Agreed and acknowledged this:

/s/ Spiro Rombotis

Spiro Rombotis Chief Executive Officer Date:

Amendment No. 4

THIS AMENDMENT NO. 4 is made on 11 day of July, 2011 BETWEEN:

- (1) DAIICHI SANKYO COMPANY, LIMITED a company incorporated in Japan whose principal place of business is at 5-1 Nihonbashi-honcho 3-chome Chuo-ku Tokyo 103-8426 Japan ("Daiichi Sankyo"); and
- (2) CYCLACEL LIMITED a company incorporated in England whose principal place of business is at Dundee Technopole, James Lindsay Place, Dundee DD1 5JJ, UK ("Cyclacel").

WHEREAS:

- (A) Daiichi Sankyo (successor of SANKYO CO., LTD) and Cyclacel are Parties to the CS-682 LICENSE AGREEMENT dated September 10, 2003, as amended by letter amendments dated April 1, 2004, April 28, 2004 and January 13, 2005 (the "License Agreement").
- (B) Cyclacel asked Daiichi Sankyo to waive its termination right under the Clause 4.2 of the License Agreement in consideration of the investment that Cyclacel has made in the Product (as defined in the License Agreement).
- (C) Cyclacel and Daiichi Sankyo wish to amend certain terms and conditions of the License Agreement in consideration of Daiichi Sankyo's waiver of the Clause 4.2 termination right.

NOW, therefore both Parties agree as follows:

- 1. Clause 3.3.6 of the License Agreement is hereby entirely deleted and replaced with:
 - 3.3.6 royalties applicable on Net Sales as follows:

```
3.3.6.1 [***] on annual Net Sales less than [***];
```

3.3.6.2 [***] on annual Net Sales greater than or equal to [***] and less than [***];

3.3.6.3 [***] on annual Net Sales greater than or equal to [***]

- 2. Clause 3.4 of the License Agreement is hereby entirely deleted and replaced with:
 - 3.4 In the case of sales of the Product by Cyclacel, its Affiliates or Cyclacel Licensees in countries where such sale does not fall within a Valid Claim of a Licensed Patent Right the royalty payable on Net Sales in such country shall be [***] regardless of the volume of the Net Sales in such country.
- 3. Clause 3.6 of the License Agreement is hereby entirely deleted and replaced with:

[***]

- 4. Subject to the terms and conditions of this Amendment No. 4, Daiichi Sankyo hereby irrevocably waives its termination right under the Clause 4.2 of the License Agreement as defined in (A) and releases Cyclacel and its officers, directors, employees and agents, and their respective successors, heirs and assigns, from all claims and liability of any kind, whether presently known or unknown, arising out of Cyclacel's performance or lack of performance under Clause 4.2 of the License Agreement prior to the effective date of this Amendment No. 4. Clause 4.2 of the License Agreement is hereby entirely deleted as of the effective Date of this Amendment No. 4.
- 5. The remainder of the License Agreement shall remain in full force and effect unless otherwise modified by both Parties in writing.

IN WITNESS WHEREOF the Parties have executed this Amendment No. 4 as of the date first written above.

SIGNED by for and on behalf of Daiichi Sankyo. Date: July 5, 2011

/s/ Noriaki Ishida

Noriaki Ishida, Corporate Officer, Vice President, Business Development & Licensing

SIGNED by for and on behalf of Cyclacel. Date: July 11, 2011

/s/ Paul McBarron

Paul McBarron, Chief Operating Officer, Executive Vice President, Finance

Certification of Principal Executive Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002

- I, Spiro Rombotis, certify that:
- 1. I have reviewed this Quarterly Report on Form 10-Q for the period ended June 30, 2011 of Cyclacel Pharmaceuticals, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting: and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 12, 2011

/s/ Spiro Rombotis

Spiro Rombotis President & Chief Executive Officer (Principal Executive Officer)

Certification of Principal Financial Officer Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002

- I, Paul McBarron, certify that:
- 1. I have reviewed this Quarterly Report on Form 10-Q for the period ended June 30, 2011 of Cyclacel Pharmaceuticals, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 12, 2011

/s/ Paul McBarron

Paul McBarron Chief Operating Officer, Chief Financial Officer and Executive Vice President, Finance (Principal Financial Officer)

Certification of Principal Executive Officer Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

Pursuant to 18 U.S.C. s 1350, as created by Section 906 of the Sarbanes-Oxley Act of 2002, the undersigned officer of Cyclacel Pharmaceuticals, Inc. (the "Company") hereby certifies, to such officer's knowledge, that:

- (i) the Quarterly Report on Form 10-Q of the Company for the period ended June 30, 2011 (the "Report") fully complies with the requirements of Section 13(a) or Section 15(d), as applicable, of the Securities Exchange Act of 1934, as amended; and
- (ii) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 12, 2011 /s/ Spiro Rombotis

Spiro Rombotis

President & Chief Executive Officer

Certification of Principal Financial Officer Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

Pursuant to 18 U.S.C. s 1350, as created by Section 906 of the Sarbanes-Oxley Act of 2002, the undersigned officer of Cyclacel Pharmaceuticals, Inc. (the "Company") hereby certifies, to such officer's knowledge, that:

- (i) the Quarterly Report on Form 10-Q of the Company for the period ended June 30, 2011 (the "Report") fully complies with the requirements of Section 13(a) or Section 15(d), as applicable, of the Securities Exchange Act of 1934, as amended; and
- (ii) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 12, 2011 /s/ Paul McBarron

Paul McBarron Chief Operating Officer, Chief Financial Officer and Executive Vice President, Finance